MIXED SIGNAL PLATFORM CIRCUITS FOR LIFETIME IMPROVEMENT OF ULTRA-LOW POWER SYSTEMS

A DISSERTATION PRESENTED TO THE FACULTY OF THE SCHOOL OF ENGINEERING AND APPLIED SCIENCE UNIVERSITY OF VIRGINIA

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN ELECTRICAL ENGINEERING

 $\mathbf{B}\mathbf{Y}$

AATMESH SHRIVASTAVA

JANUARY 2014

© 2014 AATMESH SHRIVASTAVA

MIXED SIGNAL PLATFORM CIRCUITS FOR LIFETIME IMPROVEMENT OF ULTRA-LOW POWER SYSTEMS ABSTRACT

Ultra-Low Power (ULP) Systems on Chip (SoCs), such as wireless sensor nodes, are being used for an ever-increasing number of applications. They can be used for applications measuring and reporting almost everything, from the flow of crude oil in a remote pipeline, to the degree of corrosion in a steel bridge, to electrocardiogram (EKG), electroencephalogram (EEG), and electromyogram (EMG) signals of a home health patient. Owing to their remote locations and small form factor, these nodes need to consume extremely low power. They should operate from harvested energy, as changing their batteries regularly is not a viable option. Energy autonomy is central to the widespread deployment of these sensor nodes. However, a significant percentage of energy is lost in harvesting, supplying regulated output voltages, and in the "idle mode" of the state-of-the- art sensor nodes. Therefore, energy efficiency is the most important challenge facing the design of these sensor nodes.

The design of the processor, radio, memory, etc. for a sensor node has received much attention. However, a sensor node also needs clock sources and power supplies to be able to operate. These circuits constitute the "infrastructure" around which a digital system can be created. In wireless sensors, these circuits help extract energy from harvesters, provide regulated output voltages and clock sources for the SoC. The design of these infrastructure circuits can impact the energy efficiency and hence the life time of the SoC significantly. In the state-of-the- art energy harvesting wireless sensor SoC, a significant percentage of energy is lost in harvesting and supplying regulated output voltages for the sensor node due to the poor efficiency of the harvester and voltage regulators. Also, wireless sensor nodes typically have a short burst of activity followed by a long idle time. This is done to save energy. The total power consumption of a sensor is often dictated by the power consumed in the idle mode. A clock source is often the only functional circuit in the idle mode, which is used for time keeping for "wake up" and synchronization needs. For such SoCs, clock power determines the life-time.

This work focuses on the flow of energy in ULP SoCs, such as wireless sensors. The proposed circuits enable highly efficient energy extraction from energy harvesters that can harvest solar and thermo-electric energy, efficient voltage regulators to provide power supplies for the SoC, the proposed work enables low voltage operation, and a ULP Clocking scheme to elongate the life-span in idle mode. These circuits increase the amount of energy harvested from the ambient source, reduce the loss in voltage conversion, enable low voltage operation and decrease the power consumption in idle mode to significantly improve the operational life time of a ULP system

APPROVAL SHEET

The dissertation is submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

AUTHOR

The dissertation has been read and approved by the examining committee:

Dr. Benton Calhoun

Advisor

Dr. Mircea Stan

Dr. Yogesh Ramadass

Dr. John Lach

Dr. Kamin Whitehouse

Dr. Travis Blalock

Accepted for the School of Engineering and Applied Science:

James H. Ay

Dean, School of Engineering and Applied Science

May 2014

iii

"गुरु प्यारे का दम दम शुकर गुज़ार"

I dedicate my dissertation to my Guru, Param Guru Huzur Dr. Prem Sharan Satsangi

Sahab.

Acknowledgements

I wish to thank my teacher, research adviser, soon-to-be-colleague, and a good friend, Professor Ben Calhoun. It has been an absolute pleasure to work with him. He provided me with several wonderful opportunities which helped me to improve my personal and professional skills. I have learned a great deal from him on technical topics, design, and engineering and, more recently, on the art of teaching a graduate class. The amount of energy he possesses and the excitement he shows about projects is amazing. I have often found Ben to be more excited about my projects than I have been. He has been the biggest influence on me over the last few years, and helped me to become a better researcher, a better teacher and a better human being. Thank you, Professor Calhoun, for everything. I yearn to become a teacher and researcher like you.

I wish to thank my committee members who were more than generous with their expertise and precious time. A special thanks to Dr. Mircea Stan, my committee chairman, for his inputs and constant encouragements. A special thanks to Dr. Yogesh Ramadass, for taking time out to serve on my committee and for working with me during my internship. It was a pleasure to work with you and I learned a great deal during that internship. Thank you, Professor Blalock, Professor Lach, and Professor Whitehouse for serving on my committee and for providing me with valuable feedback. I would like to acknowledge and thank the engineering school and its members for allowing me to conduct my research and for providing any assistance requested. Special thanks go to Natalie Edwards, Dan Fetko, Ydira Weaver and Terry Tigner for providing me with various official and travel related assistance. I would also like to thank Steven Bartling and Srini Sridhara of Texas Instruments for their support during my summer internship in 2012. Special thanks to founders of PsiKick, Brendan Richardson, David Wentzloff and Ben Calhoun, for their help and support during my internship in 2013. Bengroup has a culture of its own and I was very fortunate to meet bengroupers during my graduate life. I thank everyone in bengroup for tolerating me. Special thanks to Yousef Shaksheer for being a good friend and for helping me out on several occasions for chip testing. I would also like to thank Kyle Craig and Yanqing Zhang for putting up with me during tapeouts and with my unreasonable IO pad demands or area demands. I would also like to thank each member of bengroup. Each one of you has influenced me in one way or another. Interacting with new students Farah and Chris was very enlightening. I wish all of you great success in life.

One cannot go far without friends. I was very fortunate to have met Sudhanshu Khanna, Anurag Nigam and Satya Nalam during my early days of graduate school. I thoroughly enjoyed the daily tea-time, several road trips and playing cricket together. It was during my graduate school that India won the cricket world cup after 28 years and we watched several of these matches together. Anurag and I stood up several nights to watch 5-day test cricket and India was doing great those days. I would also like to thank my good friends Shanshan Chen, Jiawei Huang, and Vijay Srinivasan. It was very enlightening to interact with each one of you and I learned a great deal from each interaction.

I would not be the person I am today without the love and care of my family. I hold a special feeling of gratitude for my loving parents. My sister Parkhi is the rock of my life and always makes sure that I am doing fine. My brother Ankesh is another rock of my life and has always been there for me. I have found good friends in my *Jiju* Amritesh, *Bhabhi* Samvedna and my uncle Prem Swarup. It was the support and love of my family that kept me going through graduate school.

Contents

List of Tables
List of FiguresxiiChapter 11Introduction11.1 Flow of Energy in a Batteryless System41.2 Low Power Clocking81.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
Chapter 11Introduction11.1 Flow of Energy in a Batteryless System41.2 Low Power Clocking81.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
Introduction11.1 Flow of Energy in a Batteryless System41.2 Low Power Clocking81.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.4 Prototype Implementation and Measurement24Functioning29
1.1 Flow of Energy in a Batteryless System.41.2 Low Power Clocking81.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source.162.2.1 Compensated Oscillator Design182 nd Order Compensation20Power Supply Variation.212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.4 Prototype Implementation and Measurement24Functioning29
1.2 Low Power Clocking81.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source.162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
1.3 Energy Harvesting and Power Management101.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
1.4 Dissertation Organization11Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source.162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation.212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning.29
Chapter 214Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
Ultra-Low Power Clock Sources142.1 Prior Art.152.2 ULP On-Chip Clock Source.162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation.212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning.29
2.1 Prior Art.152.2 ULP On-Chip Clock Source162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
2.2 ULP On-Chip Clock Source.162.2.1 Compensated Oscillator Design182 nd Order Compensation Circuit19Process Compensation20Power Supply Variation.212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning.29
2.2.1 Compensated Oscillator Design18 2^{nd} Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
2nd Order Compensation Circuit19Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
Process Compensation20Power Supply Variation212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
Power Supply Variation.212.2.2 Uncompensated Oscillator Design212.2.3 Locking Circuit222.2.4 Prototype Implementation and Measurement24Functioning29
2.2.2 Uncompensated Oscillator Design 21 2.2.3 Locking Circuit 22 2.2.4 Prototype Implementation and Measurement 24 <i>Functioning</i> 29
2.2.3 Locking Circuit 22 2.2.4 Prototype Implementation and Measurement 24 <i>Functioning</i> 29
2.2.4 Prototype Implementation and Measurement 24 <i>Functioning</i> 29
Functioning
Power Consumption
2.3.1 Architecture
2.3.2 Design of Low Power Amplifier
Process Calibration
2.3.3 Amplifier duty-cycling control implementation
<i>Obtaining time of growth of oscillation</i> 37
<i>Obtaining time of decay of oscillation</i>
2.3.4 Complete circuit implementation
2.3.5 Results
2.4 Conclusions

Chapter 3	44
Energy Harvesting and Power Management for ULP Systems	44
3.1 Energy Harvesting and Power Management System	45
3.2 Energy Harvesting for Thermo Electric Generator	48
3.2.1 Functioning of TEG	49
Electrical Characteristics of TEG	52
Maximum Power Point in TEG	53
3.2.2 Design of the TEG Energy Harvester	55
Maximum Power Point Tracking Circuit	57
3.2.3 Control Circuit for Boost Converter	59
Low Side (LS) Control Scheme: Peak Inductor Current Control	60
High Side (HS) Control Scheme: Zero Detection	65
Start-up in the boost-converter	68
3.2.4 Prototype Implementation and Measurement	69
3.3 Voltage Reference Circuit for ULP Systems	78
3.3.1 Bandgap Voltage Reference	79
3.3.2 Background	81
Generation of Bandgap Voltage	82
3.3.3 State-of-the-art Bandgap Circuits for Low Power Applications	86
Banba Bandgap Reference	86
3.3.4 A 80nW, Low Power Bandgap Reference Operational from 0.8V	89
Switched Capacitor Voltage Inverter Circuit	
Proposed Bandgap Circuit : Concept	
Proposed Bandgap Circuit	
3.3.5 Prototype Implementation and Results	
3.4 Power Management Solution for ULP SoCs	
3.4.1 Voltage Regulation for Integrated Circuits	
Voltage Regulator Circuit: Linear Regulator	100
Voltage Regulator Circuit: Switching Regulator	101
3.4.2 Single Inductor Multiple Output DC-DC Converter for ULP Systems	105
3.4.3 Single Inductor EHM for ULP Systems	
Architecture	108
Architecture of the buck-converter	110
<i>I</i> _{PEAK} control for buck converter	

Architecture of the boost-converter	
Prototype Implementation and Results	
3.5 Conclusions	
Chapter 4	
4.1 Modelling for Power Management	
4.1.1 DC-DC converter model	
DC-DC Efficiency with Load Current: PWM Control Scheme	
DC-DC Efficiency with Load Current: PFM Control Scheme	
Verification of the model:	
Efficiency with Output Voltage	
Settling Time	
Supply Rail Switching Energy	
4.1.2 Evaluation of DVS techniques using the proposed model	
Framework for Energy Calculation in DVFS	
Panoptic Dynamic Voltage Scaling	
Framework for Energy Calculation in PDVS	
4.2 Implementation of Power Management for PDVS	
SIMO Digital Controller	
Control of Ripple through PDVS	
4.3 Conclusions	
Chapter 5	
Conclusions	
5.1 Impact of contributions on system life-time	
5.2 Summary of Contributions	
5.2 Conclusions and Open Problems	
Appendix A: List of Acronyms	
Appendix B: Efficiency Dependence of DC-DC on Peak Inductor Current	
Appendix C: List of Publications	
Upcoming Publications : Own	
Upcoming Publications : Group	
Patents:	
Bibliography	

List of Tables

Table 2-1 Comparison summary of the duty-cycled oscillator with previous low power works. 4
Table 3-1 Comparison summary of the boost converter with previous micro-power converters 7'
Table 3-2 Comparison summary with previous low power bandgaps
Table 3-3 Comparison with existing state-of-the-art harvesters 120
Table 4-1 Comparison with prior art for low load converters. This design supports 3 outputs at
lower voltage and load than any prior SIMO with comparable efficiency and ripple but no off-
chip capacitors

List of Figures

Figure 1-1 Deployment of Body Sensor Nodes [2]
Figure 1-2 A batteryless energy harvesting BSN SoC [1]
Figure 1-3 Flow of energy in batteryless system highlighting the requirements for increasing the
system life-time
Figure 1-4 Typical Power Consumption Profile of a ULP SoC
Figure 1-5 Power consumption of the system with duty-cycling9
Figure 1-6 Optimal energy point of a block inside a SoC [7]10
Figure 2-1 Scheme of re-locking low stability oscillator to achieve high effective stability 16
Figure 2-2 Architecture of the clock source
Figure 2-3 Current Controlled ring oscillator
Figure 2-4 a) PTAT current source b) Simulation result of a Constant current source
Figure 2-5 2^{nd} order compensation circuit using LV _T transistor to compensate for increase in
current with temperature
Figure 2-6 Temperature variation of the clock period when employing the 2nd order
compensation circuit
Figure 2-7 a) Binary weighted configuration bits b) Simulation result of current source variation
across process points
Figure 2-8 OSC _{CMP} DCO Architecture
Figure 2-9 OSC _{UCMP} DCO architecture
Figure 2-10 Timing diagram of frequency comparator circuit
Figure 2-11 Fast locking circuit for DCO using binary search
Figure 2-12 Simulation result of the locking transient
Figure 2-13 Measured Clock Period of OSCCMP showing the improvement in stability by 2nd
order compensation circuit
Figure 2-14 Measured Stability of 10-chips from 20 to 40 °C
Figure 2-15 Recalibration of OSC _{UCMP} with OSC _{CMP}
Figure 2-16 Measured waveform of DCO outputs at 100 kHz after recalibration
Figure 2-17 Measured power/stability tradeoff with duty cycling for 1oC/s temperature change26
Figure 2-18 Die photo: Implementation of the prototype clock in 130nm CMOS process

Figure 2-19 Energy per cycle and stability comparison with existing solution	27
Figure 2-20 Crystal Oscillator Circuit	28
Figure 2-21 Negative Resistance (R _N)design method to meet the oscillation criteria for crystal	
oscillator, R _N > ESR	29
Figure 2-22 Proposed control scheme, where oscillator is turned on and off to save power 3	32
Figure 2-23 Power consumption of the amplifier circuit with VDD voltage for fixed negative	
resistance of -70k Ω	\$4
Figure 2-24 Calibration technique for amplifier to address variation effects	35
Figure 2-25 Simulation result of the XTAL at 0.3V VDD and consuming 2nW power	36
Figure 2-26 Circuit to obtain the time of growth (TG) of oscillation	37
Figure 2-27 Circuit to obtain the time of decay (TD) of oscillation	38
Figure 2-28 Complete circuit diagram of the proposed crystal oscillator circuit	39
Figure 2-29 Simulation result of oscillator duty-cycling technique	0
Figure 3-1 Energy harvesting and power management system	15
Figure 3-2 Energy harvesting and power management system in a BSN SoC [1] 4	17
Figure 3-3 Practical Thermoelectric generator connecting large number of <i>pn</i> junction to increase	se
the operating voltage [42]	51
Figure 3-4 Equivalent circuit of a thermoelectric generator	52
Figure 3-5 Energy harvester harvests maximum energy when impedance of the harvester	
matches TEG's input resistance	53
Figure 3-6 Maximum power point characteristics of a TEG energy harvester	54
Figure 3-7 proposed design architecture for the TEG harvester using a boost converter	55
Figure 3-8 Maximum power point tracking circuit for TEG	57
Figure 3-9 Simulation Result of MPP tracking circuit showing5	58
Figure 3-10 Timing diagram of the control signal for the boost converter showing use of two	
phases $\phi 1$ and $\phi 2$, where $\phi 1$ is used for cancelling mismatch to harvest at low voltage	59
Figure 3-11 Efficiency variation with peak inductor current for various input voltages	50
Figure 3-12 LS control circuit to generate the constant peak inductor current	52
Figure 3-13 Timing diagram for LS control circuit	52
Figure 3-14 Simulation result of peak inductor current with VIN and VCAP	55
Figure 3-15 Circuit to generate the HS timing using zero inductor current crossing	66

Figure 3-16 Schematic of zero detection comparator with offset cancellation scheme	. 67
Figure 3-17 Die Photo and design parameter of the boost converter	. 70
Figure 3-18 Maximum power point tracking pulse for MPP CLK	. 70
Figure 3-19 Measured TON of low side LS switch w.r.t. V_{CAP} and V_{IN} indicating peak inductor	or
current control scheme	. 71
Figure 3-20 Measured efficiency with Low Side T _{ON} , showing increased efficiency at Higher	
T_{ON} for Low $V_{IN}.$ It achieves 84% efficiency and 0.4V V_{IN} and 22% efficiency at 10mV V_{IN}	. 72
Figure 3-21 Variation of peak inductor current with process using Monte-Carlo simulation	. 73
Figure 3-22 Measured optimal zero detection	. 74
Figure 3-23 Measured operation of the boost converter with VIN=10mV and VIN=8mV	. 75
Figure 3-24 Measurements showing kick-start from 590mV charging to 1V	. 76
Figure 3-25 Use of voltage reference circuit in basic building blocks such as voltage regulator	S
and ADC	. 78
Figure 3-26 Principle of bandgap voltage reference circuit generating a constant voltage	
reference with temperature	. 81
Figure 3-27 PTAT and CTAT voltage generation circuit	. 82
Figure 3-28 A conventional bandgap reference circuit	. 84
Figure 3-29 Variation of bandgap reference with temperature showing 17ppm/°C variation	. 85
Figure 3-30 Architecture of Banba Bandgap circuit	. 87
Figure 3-31 Circuit diagram of the proposed bandgap reference	. 89
Figure 3-32 Switched capacitor voltage inverter circuit and its operation	. 90
Figure 3-33 Proposed bandgap reference circuit using switched capacitor voltage inverter	. 92
Figure 3-34 Transient behavior of bandgap reference	. 94
Figure 3-35 Bandgap output variation with temperature	. 95
Figure 3-36 Variation with process: Monte-Carlo Simulation	. 96
Figure 3-37 Variation with Vin Voltage	. 96
Figure 3-38 Energy-Delay characteristics of a digital circuit	. 99
Figure 3-39 A linear regulator circuit to supply V _{DD}	100
Figure 3-40 DC-DC converter architectures	103
Figure 3-41 Single inductor multiple output buck-boost converter for ULP systems	105
Figure 3-42 Architecture of the proposed EHM system	107

Figure 3-43 Proposed energy and power management solution using single inductor 108
Figure 3-44 Inductor current waveform for the EHM switching scheme showing the change in
direction of Inductor Current for boost operation
Figure 3-45 Complete architecture of the EHM System 110
Figure 3-46 Architecture of the buck converter used in EHM 111
Figure 3-47 Peak inductor current control scheme in buck converter to achieve high efficiency
Figure 3-48 Variation of I_{PEAK} Vi showing <1% variation with Input Voltage Vi 113
Figure 3-49 Variation of peak inductor current with process and mismatch variation 114
Figure 3-50 High side control scheme for the boost converter to accurately predict the zero
crossing of the inductor current enabling reliable operation for 5V battery/storage using 3.3V
devices
Figure 3-51 Boost converter architecture
Figure 3-52 Cold start circuit using an internal charge-pump to start operation from 380mV 116
Figure 3-53 Voltage regulation output of 1.2V rail showing <15mV ripple
Figure 3-54 Measurement result of multiple output V_{DDs} coming at 1.2V, 3.3V and 1.5V 118
Figure 3-55 Efficiency of buck Converters
Figure 3-56 Efficiency of the boost converter
Figure 3-57 Die photo of EHM system119
Figure 4-1 Structure of the proposed model
Figure 4-2 Loss mechanisms inside a typical switching DC-DC converter
Figure 4-3 Efficiency Variation with load current in a) PWM scheme with $\eta_2=0.9$, $\eta_1=0.68$ and
$I_L{=}1mA$ b) PFM scheme with $\eta 2{=}0.88$ and a=5x10-5
Figure 4-4 Comparison of the load equation with measured work in literature
Figure 4-5 Dynamic efficiency variation with current and voltage
Figure 4-6 Energy consumption for a microcontroller across voltage with and without
consideration of the DC-DC converter efficiency
Figure 4-7 Efficiency profiles for two different converters (a) and the impact of a converter
change on the overall energy drawn from the battery (b)
Figure 4-8 Dedicated DC-DC converter per block
Figure 4-9 Operating condition for dedicated DC-DC

Figure 4-10 Energy Savings with rate of Voltage Scaling 142
Figure 4-11 PDVS: Block Level Voltage Scaling Technique
Figure 4-12 PDVS Operating Condition Evaluation
Figure 4-13 Energy Benefits of PDVS and Dedicated DVFS
Figure 4-14 Energy Benefits of PDVS for different values of the virtual VDD capacitance of the
block
Figure 4-15 Dynamic Voltage Scaling for multi-core system
Figure 4-16 PDVS implementation for multi-core system
Figure 4-17 SIMO DC-DC buck converter for low voltage DVS-enabled. On-chip decoupling
capacitance saves system level volume and cost
Figure 4-18 Conventional HS control
Figure 4-19 Proposed control scheme
Figure 4-20 Control Architecture to reduce the ripple
Figure 4-21 ripple Voltage for different values of decoupling capacitor at different loads 155
Figure 4-22 Converter Operation in CCM and DCM mode
Figure 4-23 Transient behavior of the converter
Figure 4-24 Low Side control Scheme
Figure 4-25 Digital control logic of SIMO controller
Figure 4-26 PDVS load conditions when most of the cores are connected to one supply 161
Figure 4-27 Measured outputs of the SIMO. During different load switches at constant total load
of 10 mA, compensating PDVS changes with rail priority adjustments reduces ripple by 30 mV.
High load efficiency is 86%, and low load efficiency is 62%
Figure 4-28 Measured efficiency plots for different configurations
Figure 4-29 Die microphotograph of the SIMO chip and performance summary 166
Figure 5-1 Energy System level performance impact of the proposed EHM Solution 170
Figure 5-2 Current Consumption break-down of a BSN SoC [86] while performing ECG 171
Figure 5-3 Lifetime improvement by proposed solution. The system start-up energy is reduced
by 3X, start-up time by 20X and operational time is increased by 9X
Figure 5-4 Performance and power consumption of BSN SoC [1] and the proposed solutions 172
Figure 5-5 Operational lifetime improvement in ULP systems by using proposed contributions

Chapter 1

Introduction

Ultra-Low Power (ULP) Systems on Chip (SoCs), such as Body Sensor Nodes (BSNs) or Wireless Sensor Nodes (WSNs), and more recently internet of things (IoT) promise to change the way we experience life by providing rich information about our activities, health, and the environment. These miniaturized nodes are responsible for sensing data periodically, processing it, and communicating information wirelessly. They can be used for hundreds of sensing applications – measuring and reporting everything from the flow of crude oil in a remote pipeline, to the degree of corrosion in a steel bridge, to EKG, EEG and EMG signals of a home health patient. These devices require small size and must consume extremely low power to be able to operate from harvested energy for their longer life time. Significant progress has been made over the last few years. Sensor nodes, which can operate from harvested energy without batteries, have been demonstrated [1]. However, there is a need to improve the operating lifetime of these systems for their extensive deployment in the environment. This is an ongoing research topic at various levels, from system to software to hardware. This dissertation focuses on improving the lifetime of a ULP SoC at the hardware level. It proposes improved energy harvesting and power management solutions, clock sources, a modeling framework to enable accurate power management, etc. These circuits and the model constitute the basic infrastructure for the efficient use of energy in ULP Systems.



Figure 1-1 Deployment of Body Sensor Nodes [2]

Figure 1-1 shows the deployment of body sensor nodes in a body area sensor network implementation. While the figure here shows a specific application in body area networks, a similar principle exists in interconnected wireless sensor nodes and internet of things. These networks have several sensor nodes deployed at several locations to accomplish a range of sensing functionality. In Figure 1-1, several BSNs are placed on a human subject to monitor his ECG, blood pressure, EMG, etc. The sensor nodes interact wirelessly with the body area aggregator, which acts as a base station in a wireless system. The body area aggregator can be a cell phone or other wireless transceiver system capable of processing. The interconnected system can be used to monitor the health of the human subject, provide information on his bio-metrics and communicate in cases when intervention is required. The sensor nodes are essential systems for the proper operation of these networks, as they provide the raw data. For large scale and pervasive use of

these systems, sensor nodes must be of small factor, consume low power, and have long lifetimes. Since these systems are expected to be ubiquitous and are often remotely located, a batteryless sensor node is desirable, as it requires no intervention and provides energy autonomy. The dissertation focuses on providing power autonomy for sensor nodes by improving the energy harvesting, reducing the power consumption and providing operation from low energy levels.



Figure 1-2 A batteryless energy harvesting BSN SoC [1]

Figure 1-2 shows a BSN SoC [1] that is a batteryless system that harvests energy from body heat. It operates using the harvested energy to enable a perpetual energy autonomous system. It has an Analog Front End (AFE) and Analog to Digital Converter (ADC) to perform sensing, which can be ECG, EKG or other environmental sensing. The sensed data is sent for digital processing, which uses a micro-controller, memory etc. The processed data is then transmitted through an RF transceiver to a nearby aggregator (for example, the Body Area Aggregator, as shown in Figure 1-1). The SoC in Figure 1-2 harvests energy from a Thermo-Electric Generator (TEG) utilizing body heat. The circuitry that extracts energy from the harvester consists of a boost converter that collects the energy from the TEG at lower voltage and stores it at a higher voltage on a capacitor. Higher voltage is needed for circuit operation. Finally, a voltage regulation block provides

different power supply rails (V_{DDS}) to run different blocks in the SoC, such as analog, RF, etc. The lifetime of the system is impacted by the amount of energy harvested, the power consumption of the system and minimum energy level at which the system can operate. This dissertation contributes to improve on all these components to improve the life-time of these systems.

1.1 Flow of Energy in a Batteryless System



Figure 1-3 Flow of energy in batteryless system highlighting the requirements for increasing the system life-time

Figure 1-3 shows the flow of energy in an energy harvesting battery-less ULP system. First, energy is harvested from an ambient source, such as solar cells, TEG, or other ambient source, using an energy harvester circuit. The harvested energy is stored on a capacitor in the form of voltage V. This stored energy is then taken up by the power management circuit to provide various power

supply voltages (V_{DDS}). These V_{DDS} are used to provide power supply voltages to various communication and processing blocks on the system. While the first part of the system constitutes the process for obtaining energy from the environment, the second part of the system is the use of this energy to accomplish the sensing task. Improvements in low power sub-threshold design [3][4], as well as improvements in the area of low power radios for communication [5][6], are reducing the power consumption of processing and communication which helps in improving the lifetime of the system. However, there is a need for improvement at each step of energy flow in the system to significantly improve its life-time. This can be accomplished through the following design requirements.

- The system should be able to harvest from very low voltage and power levels from the ambient sources to extract every possible bit of energy available there.
- The energy harvester should be efficient and minimize the losses involved in transferring the energy from an ambient source to the storage capacitor.
- The voltage on the storage capacitor is the indicator of the amount of energy available for the system. The system can turn on only if the voltage on the capacitor reaches a set threshold level, which would indicate a minimum energy level for the system to be able to operate. Energy below this level is not useful for the system. The life-time of the system can be improved significantly if the operating voltage on the capacitor is brought down. This would bring down the minimum energy level needed for the system to turn on. In other words, the system can turn on from a lower level of harvested energy. Therefore, the operating voltage of the system should be brought down to improve its lifetime.
- The stored energy on the capacitor is transferred to generate various power supplies needed for the system to operate different blocks in the system at different voltage level (which is

done to reduce the processing and communication power). Therefore, power losses in this step should be minimized and the voltage conversion should be very efficient.

• The ULP systems spend most of their time in idle mode, with short bursts of high activity. The total power consumption of the system is often dictated by the idle mode power consumption. To increase the life-time of the system, the power consumption in idle mode needs to be reduced.

Therefore, the lifetime of the system is increased by increasing the amount of harvested energy, reducing the losses in voltage conversion, reducing the operating voltage, and reducing the standby power consumption. This is the central theme of this dissertation, and the following contributions are made to achieve higher life-time for ULP systems.

- A stable on-chip clock source that can lock to a given clock frequency, a locking scheme to lock the clock source and ultra-low power on-chip clock sources to reduce the idle mode or standby current.
- An ULP crystal oscillator circuit as an alternative clocking solution for lower power consumption and higher stability.
- A low power, high efficiency energy harvester extraction circuit that can harvest from 10mV ambient sources to increase the amount of harvested energy.
- A low power, low voltage bandgap reference for energy harvesting and power management to reduce the idle mode power consumption and enable low voltage operation.
- A single inductor multiple output buck-boost converter that enables high efficiency circuits as well as low voltage operation.

- A single inductor multiple output energy harvesting and power management solution to harvest from low power solar cells, to enable high efficiency, low cost, integrated harvesting and power management system.
- A model that accurately establishes the benefits of power management techniques for ULP SoCs in the presence of voltage regulators to enable design time choices for regulators and power management techniques.
- A single inductor multiple output Voltage regulator with on-chip decoupling capacitors to implement panoptic dynamic voltage scaling technique.
- While not included in this dissertation, the author also made the following contributions for the overall improvement in ultra-low power system design.
 - o A clock and data recovery (CDR) circuit for the RF interface of BSN
 - A low power interconnect circuit operational from 0.3V to reduce the interconnect power consumption
 - An ultra-low power analog to digital converter for ECG, EKG applications.
 - Design and system level improvements and options for inter-chip communication between ULP ICs
 - o Modeling techniques to assess circuit and system reliability
 - Circuit technique to improve the start-up time of the crystal oscillators by an order of magnitude.

These contributions will help us enable longer life-time, low power systems which will help in making ubiquitous sensing and sensor networks a reality which in turn will help in improving our way of living by improvements in health care, safety, entertainment, activity tracking and various other aspects of our lives. Next section gives a brief description of the contributions from the system perspective.

1.2 Low Power Clocking



Figure 1-4 Typical Power Consumption Profile of a ULP SoC

A typical operation of a ULP SoC like BSN constitutes a short burst of activity followed by a long idle time. The short burst of activity consumes higher power, whereas in idle mode, the system consumes very little power. Spending larger time in idle mode saves energy relative to the continuous operation. Figure 1-4 shows the power consumption profile of a BSN. The system wakes up first and starts the CPU. Analog blocks are then turned on to perform the sensing application and finally, RF turns on for communication. The power consumption of the system rises in steps as each block turns on. The system goes back into sleep mode after completing the RF communication. These SoCs use a real-time clock (RTC) for keeping time. The RTC maintains precise timing for waking up and synchronizing the chip. Figure 1-4 also shows that the total power consumption of the chip can vary depending on the application. For example, the power can be

dominated by the active mode if the SoC has higher activity, and is dominated by the idle mode (RTC power) if the SoC has lower activity. There is a need to reduce and optimize the power consumption in both regions of activity in order to extend the system life time. In this work the idle mode power is reduced by the proposed low power clocking solution while active mode power consumption is reduced by efficient energy harvesting and power management (EHM) circuits.



Figure 1-5 Power consumption of the system with duty-cycling

Figure 1-5 shows that the power consumption of a SoC, which stays in the idle mode for the majority of time, is dominated by the RTC. The RTC is typically implemented using an off-chip crystal resonator (XTAL). Their power consumption can be anywhere from a few 100nW to hundreds of μ W. This work presents ULP clock sources to reduce the power consumption in the RTC. A solution designed with an on-chip, silicon-based oscillator reduces the power consumption of the clock to 150nW. A locking scheme is proposed to lock the on-chip clock source to a known frequency. The proposed solution has been fabricated and demonstrated in silicon. A second

version of the design brings the power down to 20nW. The proposed on-chip solution has a lower cost when compared to off-chip RTCs utilizing XTAL. Further, we also propose an alternate design of crystal oscillator reducing its power consumption to 1nW. This ULP XTAL can reduce the power consumption in RTC by a few orders of magnitude. It can be used for applications where higher stability is preferred at the expense of cost. Chapter 2 covers the implementation details of the proposed circuit.

1.3 Energy Harvesting and Power Management



Figure 1-6 Optimal energy point of a block inside a SoC [7]

An optimal energy operating point exists for each block in a SoC [7]. Figure 1-6 shows the variation of energy consumption with VDD. Various blocks of a BSN, such as AFE, RF, or CPU, need their own power supply voltage (VDD) to be able to operate at the optimal energy and performance point. Therefore, a BSN SoC needs multiple supply voltages apart from the energy harvesting interface. Conventional approaches to obtaining and managing energy from ambient

sources either employ more than one inductor, which increases the cost, or use low drop out (LDO) regulators which have lower efficiency. The lower efficiency of LDOs can result in the loss of energy, which is not desirable. There is a need to provide an efficient and low-cost energy harvesting and power management solution for BSNs. This work proposes a highly efficient energy harvesting and power management solution with various design options for BSNs. The proposed solution can harvest energy from TEG and solar cells. It also proposes single inductor power management circuits with on-chip and off-chip decoupling capacitor options.

In addition, the dissertation also presents a model that accurately establishes the benefits of power management techniques for ULP SoCs. The benefits of power management techniques like Dynamic Voltage and Frequency Scaling (DVFS) cannot be established accurately without assessing their impact on voltage regulators like the DC-DC converter. For example, DVFS uses a high voltage to support higher performance and a lower voltage to save power. However, changing the output voltage of a DC-DC converter incorporates significant power overhead, and the efficiency can vary widely across voltage and current loads. These overheads may offset the benefits from DVFS. There is a need to measure the benefits of power management techniques like DVFS, clock gating, etc., in conjunction with their impact on the DC-DC converter. The proposed model enables the study of various power management techniques by taking into account their impact on DC-DC converters of different topologies.

1.4 Dissertation Organization

The dissertation is organized as follows. Chapter 2 presents a clocking solution for ULP SoC. An ultra-low power clock source that can be locked to a desired system clock frequency is presented.

The proposed clock source is very stable and achieves low power by applying duty-cycling techniques. A process calibration scheme and other compensation techniques make it perform well across varying environments. Further, an ULP crystal oscillator circuit that consumes less than a 1nW of power is also presented in Chapter 2. The proposed circuit can operate at 0.3V to enable functionality at very low power levels. The crystal oscillator solution provides a low power and more stable alternative to the on-chip clock source. The contributions in the low power clocking solution help in lowering the idle mode power consumption to increase the system life-time.

Chapter 3 presents the energy harvesting and power management solution for ULP systems. Here we present a low power, high efficiency energy harvester that can harvest from a 10mV ambient source to increase the amount of harvested energy. We also present a low power, low voltage bandgap reference for energy harvesting and power management to reduce the idle mode power consumption and enable low voltage operation. Further, we also present a single inductor, multiple output buck-boost converter that has high efficiency and enables low voltage operation. Finally, we present a single inductor, multiple output, energy harvesting and power management solution to harvest from low power solar cells that enable high efficiency, low cost, integrated harvesting and power management system.

Chapter 4 presents a model which accurately establishes the benefits of power management techniques for ULP SoCs in the presence of voltage regulators to enable design time choices for regulators and power management techniques. It also presents a Single Inductor Multiple output Voltage regulator with on-chip decoupling capacitors to implement panoptic dynamic voltage scaling technique.

Finally, Chapter 5 concludes the dissertation. It quantifies the benefits in life-time improvements of the ULP system based on the contributions made in the dissertation. This chapter also points out areas for future research work and improvements for the design techniques described in this dissertation.

Chapter 2

Ultra-Low Power Clock Sources

In an Integrated Circuit (IC), a clock source is used for various functions, which can include synchronous implementation of an arithmetic and logic unit, inter-chip and intra-chip communication, and time keeping. The clock source constitutes a key component in an IC. It is extremely important for ULP SoCs like a BSN. A typical BSN operation constitutes a short burst of activity followed by a long idle time. The total power consumption of a BSN is often dictated by the power consumed in the idle mode. A clock source is often the only functional circuit in the idle mode. It is used for time keeping, for "wake up", and for synchronization needs of the BSN. The clock source, therefore, can determine the power consumption of a low duty-cyle BSN.

A clock source can be implemented either using off-chip components, such as a crystal resonator, or using on-chip devices. An on-chip implementation of a clock source is typically lower in cost but poor in stability when compared to the off-chip implementation. In this chapter, we will present a stable on-chip clock source and an ultra-low power crystal oscillator circuit with power consumption in the range of 1nW. The proposed circuits are suited for BSN and show lower power and more stability compared to the state of the art. The proposed solutions provide options in the BSN platform for the implementation of the clock. While an on-chip clock source provides a stable, low power, and cheap solution for clocks, an off-chip crystal oscillator can be used for the cases where more stability and even lower power consumption is desired.

2.1 Prior Art

BSNs typically require a stable clock source for precise data sampling, an RF modulation clock, and keeping time to reduce the cost of re-synchronizing to other radios. The conventional approach of using a crystal oscillator (XTAL) adds 3-4 off-chip passives, has startup times in the ms to second range, and can consume an appreciable fraction of the system power. For example, the energy harvesting BSN SoC in [1] consumes 19μ W while measuring ECG, extracting heart rate, and sending RF packets every few seconds, and over 2µW of that total is in the 200kHz XTAL. The alternative clocking scheme, to replace XTAL with low power and low cost on-chip reference oscillators for low power systems, is an ongoing effort [8]-[11]. Widely varied approaches exist. In [11], authors present a CMOS relaxation oscillator. High temperature stability is achieved using poly and diffusion resistors together to realize the resistor in an RC relaxation oscillator. These resistors have complementary temperature dependence, and they cancel the effect of temperature variation to achieve a temperature stability of 60ppm/°C. On-chip oscillators, using the gate leakage current, have been proposed in [8]-[10]. Gate leakage current has very small temperature dependence, which makes these oscillators stable. However, these oscillators can operate only at very low frequency (0.1-10 Hz) due to the low magnitude of gate-leakage current. Also, the oscillation frequency is not very well controlled across the process. The most stable oscillator [10] in this category has a temperature stability of 32ppm/°C operating at 0.4Hz.

Parallel to the exploration of on-chip clocks for BSNs, research has been going on for off-chip clock sources using crystal oscillator. Recent work on the design of real-time clocks using crystal oscillator shows power consumption of 5.58nW [13], making it possible to use them for BSN applications. Power consumption of crystal oscillator can be reduced by reducing the amplitude of

oscillation by operating them at lower voltages. Low power electronic watches use this technique to operate crystal oscillator circuits in sub-threshold or weak inversion regions of operation for transistors [14]. The lowest reported power for such circuit is at 27nW [15].

In this chapter we propose two solutions for providing clocks for BSNs. First, we present a new scheme to replace XTALs with a ULP on-chip clock source (no off-chip passives) for the ~100kHz range that is roughly 7X lower power than typical XTALs (operating at 100kHz) and has a temperature stability of 5ppm/°C. Further, for an off-chip solution, we propose a technique of duty cycling the crystal oscillator in conjunction while operating them in a sub-threshold region. We achieve a simulated power consumption of close to 1nW for a real time clock. These solutions provide flexible platform circuits for BSNs clocking schemes, from which a designer can choose, depending on the needs of the application.

2.2 ULP On-Chip Clock Source¹





¹ A lot of content comes from [AS5]

To maintain a stable on-chip clock source, we need a temperature compensated oscillator (Osc_{CMP}). However, these circuits tend to consume power comparable to XTALs, in the few µW range for ~100kHz. We propose a scheme to use an ULP uncompensated oscillator (Osc_{UCMP}), with the Osc_{CMP} to provide both stability and ULP consumption. Figure 2-1 shows the concept. As temperature changes at a given rate, both oscillators will aggregate time error relative to the ideal reference, with OscucMP accumulating error much faster, due to its lower stability. If we periodically lock Osc_{UCMP} to Osc_{CMP}, then its effective stability stays within a bounded error relative to Osc_{CMP}, and we can make this error arbitrarily small by changing the duty cycle of relocking. We can also re-lock to the original reference if it is available (e.g. XTAL or signal over RF). In between lock points, the higher power oscillator(s) can shut down, setting the system power to that of the Oscucmp. To make this work well, we need a fast locking circuit, low power oscillators with rapid on/off, and digital calibration storage. Figure 2-2 shows the architecture of our clock source. The locking circuit locks clock source B to A. Osc_{CMP} or Osc_{REF} can be selected at A and Osc_{UCMP}, or other clocks to be locked, can be selected at B. The locking circuit digitally calibrates B to provide an output clock with the same period as A.



Figure 2-2 Architecture of the clock source
2.2.1 Compensated Oscillator Design



Figure 2-3 Current Controlled ring oscillator

Our Osc_{CMP} uses a current controlled ring oscillator, and Figure 2-3 shows the circuit diagram. It is designed to be stable across temperature, and it uses configuration bits to compensate for process variations. The frequency is set by current Io, capacitance C_L , which are MIM (metal-insulator-metal) capacitors with very small temperature variation, and by the inverter's switching threshold. In this architecture inverter threshold is set to $V_{DD}/2$. To set the constant current Io, we use a PTAT and a CTAT current source. The current of a PTAT current source (Figure 2-4 (a)) increases almost linearly with temperature. We use a long channel MOS transistor operated in the strong inversion region to implement the CTAT, whose current decreases linearly with temperature. We add the current from the PTAT and the CTAT using current mirror to get a current independent of temperature. Figure 2-4 (b) shows that Io varies by 1% over a 100°C range (~100ppm/°C).



Figure 2-4 a) PTAT current source b) Simulation result of a Constant current source.

2nd Order Compensation Circuit



Figure 2-5 2^{nd} order compensation circuit using LV_T transistor to compensate for increase in current with temperature

The oscillator based on the current source of Figure 2-4 (b) will have an oscillation period that will still have a small dependence on temperature in the second order because the current sources are not purely linear . The period of oscillation will reduce as the temperature goes up from 20°C-100°C. We employ 2^{nd} order compensation to further improve stability. This circuit comprises an off, low threshold (LV_T) MOS, a switch, and an inverter. Figure 2-5 shows the circuit diagram of the 2^{nd} order compensation. It essentially forms a leakage pull-up path that adds charge to C_L , slightly increasing the delay. This slight increase in delay increases with temperature.



Figure 2-6 Temperature variation of the clock period when employing the 2nd order compensation circuit

Figure 2-6 shows how this nullifies the 2nd order increase in current at high temperature, giving a

nearly flat period for a 200kHz reference over the range of 20°C to 90°C.

Process Compensation

We use configuration bits to compensate for the effects of process variation on the constant current source and delay element. In the current source, variation may offset the PTAT and CTAT current, so that one dominates in the target frequency range, making temperature stability impossible. We vary resistance R in the PTAT shown in Figure 2-4 (a) to balance its current with the CTAT against global variation using 5 binary bits. Similarly, we use 6-bit binary weighted off-transistors in the 2^{nd} order compensation to align the leakage current to compensate for the process drift in leakage.



Figure 2-7 a) Binary weighted configuration bits b) Simulation result of current source variation across process points

We use the constant current source as an input to a binary weighted (8-bit) current mirror used in Figure 2-7 (a). This lets us generate the desired current for a given desired frequency across process at 20ns resolution. The PTAT current source addresses the variation of delay of the delay-line with temperature. The simulation results in Figure 2-7 (b) show that the similar stable current can be achieved across process. We include 10-bit coarse and 5-bit fine control for 1ns and 20ps resolution, respectively, which are set by the locking circuit (Section 2.2.3). This results in a Digitally Controlled Oscillator (DCO) that can lock to a desired frequency. Figure 2-8 shows the architecture of the DCO.



Figure 2-8 OSC_{CMP} DCO Architecture

Power Supply Variation

The Osc_{CMP} is sensitive to variations in the power supply. The PTAT circuit has better power supply tolerance but the CTAT is more sensitive, as it uses V_{DD} directly at its gate. Osc_{CMP} shows a dependence of ~0.1%/mV. This is still better than [4], where power supply dependence is 0.42%/mV, but it could still be a problem for stability if V_{DD} is poorly regulated. Osc_{CMP} will aggregate error over time. This can be addressed either by using a low noise regulator, such as proposed in [6], or by relocking Osc_{CMP} often with the external references.

2.2.2 Uncompensated Oscillator Design

The Osc_{UCMP} uses leakage as the current source to the delay elements, along with the same digital inverter-based coarse and fine delay compensation as the Osc_{CMP} . It uses binary weighted off LV_T transistors for the realization of the current source. Figure 2-9 shows the Osc_{UCMP} architecture. The use of off LV_T transistors gives a lower area DCO at ~100kHz. Long channel transistors are needed to realize it at ~100 kHz frequencies with on transistors. This will increase the area significantly.



Figure 2-9 OSC_{UCMP} DCO architecture

Since no compensation is needed for Osc_{UCMP} , we chose the lower area solution. Osc_{UCMP} consumes 80nW of power at 100kHz in simulation. It has a poor temperature stability of 1.67%/°C.

2.2.3 Locking Circuit

We use a 5-bit counter as a frequency comparator to compare Osc_{CMP} to a reference or to compare Osc_{UCMP} to Osc_{CMP} . The reference clock (REF_CLK) is divided by 2 and fed to the frequency comparator. The divided clock is called REF. As REF goes high, DCO gets enabled and will start



Figure 2-10 Timing diagram of frequency comparator circuit

oscillating. The counter starts counting the rising edges of the DCO when the reference input is high. If it counts more than 1, the output of the frequency comparator goes high; otherwise, it is low. Figure 2-10 shows the timing diagram.

Our locking circuit comprises this comparator, SAR logic, and either DCO in a feedback configuration. Figure 2-11 shows the architecture of the locking circuit. The frequency comparator gives 1 when the DCO's output frequency is higher than the reference and 0 when it is lower. The SAR logic approximates the current and delay inside the DCO, based on the logical output of the comparator and sets the 23 DCO control bits, which are stored in a register. The comparison is performed when REF is high, and the current source is configured when REF is low, to let the current source settle correctly.



Figure 2-11 Fast locking circuit for DCO using binary search

Figure 2-12 shows the locking transient at 200kHz. The lock takes 46 reference cycles and has a resolution error of ~20ps. Once the calibration is performed, the DCO runs on its own. The temperature compensation scheme controls the drift. In this way, the lower power oscillator is configured to the frequency of the stable clock. We can make the stability of the lower power oscillator similar to the stable clock by performing this lock often.



Figure 2-12 Simulation result of the locking transient

2.2.4 Prototype Implementation and Measurement



Figure 2-13 Measured Clock Period of OSCCMP showing the improvement in stability by 2nd order compensation circuit

We implemented the clock source (0.5mmx0.5mm) in 130nm CMOS. Osc_{CMP} and Osc_{UCMP} consume 1µW and 100nW at 100kHz, 1.1V VDD, respectively. The process tuning bits give us a full measured locking range from 15kHz to 350kHz. In this range, we can lock successfully to the

reference within the accuracy of jitter on the input clock. Figure 2-13 shows the measured stability of the Osc_{CMP} after calibration as 5ppm/°C in a BSN compatible range of 20-40°C (14ppm/°C from 20°C to 70°C). Without the 2nd order compensation, this stability degrades to 60ppm/°C. Figure 2-14 shows the temperature stability across 10 chips from a temperature range of 20-40°C.



Figure 2-14 Measured Stability of 10-chips from 20 to 40 °C



Figure 2-15 Recalibration of OSC_{UCMP} with OSC_{CMP}

We lock Osc_{CMP} to a 100kHz reference, lock Osc_{UCMP} to Osc_{CMP} , then power down Osc_{CMP} . Figure 2-15 shows the measurement result of recalibration of Osc_{UCMP} with Osc_{CMP} . In this test, Osc_{UCMP} and OSC_{CMP} were calibrated to 100kHz by an external reference at 20°C. After that, we shut down Osc_{CMP} and raised the temperature of measurement to 50°C. Then we wake up Osc_{CMP} and recalibrated Osc_{UCMP} with it. Figure 2-15 shows the process, and Figure 2-16 shows the output

waveform after calibration. Using this scheme, we can achieve stability (5ppm/oC) at ultra-low power.



Figure 2-16 Measured waveform of DCO outputs at 100 kHz after recalibration



Figure 2-17 Measured power/stability tradeoff with duty cycling for 1oC/s temperature change

Figure 2-17 shows the result of duty cycling for the case when temperature varies by 1°C/s or less. We achieve a stability of 5ppm/°C at 150nW. During locking, we can turn on Osc_{CMP} at the rising edge of Osc_{UCMP} and pass Osc_{CMP} as the clock, to avoid glitching, or clock gate the system clock to prevent problems in downstream circuits. Figure 2-18 shows the die photo with the proposed circuit implementation in 130nm CMOS process.



Figure 2-18 Die photo: Implementation of the prototype clock in 130nm CMOS process



Figure 2-19 Energy per cycle and stability comparison with existing solution

Figure 2-18 shows the die photo. Figure 2-19 shows that our work is 10X more stable and consumes 3X less energy than the prior art. Some ULP timers for BSNs are approaching our stability [3-4], but they are 10^6 slower frequency. We exhibit similar stability to an XTAL with

~7X less power and no off-chip components, providing a low cost ULP solution for wireless sensors and BSNs.

2.3 ULP Crystal Oscillator

In this section, we present an ULP crystal oscillator circuit. A crystal is an electromechanical resonator which resonates at its natural frequency when excited with electrical energy. Figure 2-20 shows a conventional crystal oscillator circuit. The equivalent circuit of a crystal consists of a series RLC circuit, with a parasitic parallel capacitor Cp. The frequency of oscillation is mainly determined by Lm and Cm. The effective series resistor (ESR) is the energy dissipating component of the crystal. The inverting amplifier, Amp, provides the negative resistance that overcomes the loss from ESR and pumps energy into the crystal, making it oscillate at its natural frequency.



Figure 2-20 Crystal Oscillator Circuit

The output frequency of the crystal oscillator is very precise and its stability is usually specified at parts per million or per billion (ppm/ppb). The extremely precise output frequency of crystal oscillators makes them a natural choice for implementing clocks. A crystal oscillator can be a more convenient solution to implement clocks in BSNs, when compared to the on-chip solution

presented in section 2.2, because it would not need calibration for process and voltage given the application. A crystal oscillator's frequency of oscillation is independent of voltage and process. However, it needs off-chip components and can incur higher costs.

Functioning

Figure 2-20 shows the architecture of the crystal oscillator. It consists of a series RLC circuit, with a parallel parasitic capacitance Cp. The quality factor (Q) of the crystal oscillator is in the 50,000 range, which provides a very precise frequency. The circuit can be made to oscillate in series or parallel mode. Parallel mode is the preferred mode of oscillation. In parallel mode, the crystal is connected with an inverting amplifier with two load capacitors connected in parallel (C_L), as shown in Figure 2-20. In parallel mode, the crystal oscillator appears as an inductor and oscillates with the load capacitors (C_L). In order to oscillate, the circuit needs to meet the Barkhausen criteria of oscillation. The inverting amplifier, Amp in Figure 2-20, is designed to meet the oscillation criteria. Figure 2-21 shows the design method to meet the oscillation criteria.



Figure 2-21 Negative Resistance (R_N)design method to meet the oscillation criteria for crystal oscillator, R_N > ESR

In this circuit, crystal is removed from the oscillator and an *ac* current source is connected across the amplifier terminal, as shown in Figure 2-21 (a). The inverting amplifier presents a negative resistance to the applied *ac* current. The value of the negative resistance is a function of frequency. Figure 2-21 (b) shows the negative resistance of the amplifier with frequency. The negative resistance of the amplifier is an active component and is needed to overcome the damping effect in crystal's ESR in the RLC circuit of Figure 2-20. The ESR of the crystal is an energy dissipating component, and the amplifier compensates for this dissipation through the negative resistance by supplying/replenishing the dissipated energy in the crystal. The Barkhausen criteria for oscillation are met when negative resistance of the amplifier is greater than the ESR of the crystal. As a rule of thumb, the negative resistance of the amplifier should be greater than the ESR by at least three times in magnitude. The ESR of the crystal oscillator at 32.768kHz is typically in the range of $30k\Omega$. Figure 2-21 (b) shows that the designed amplifier shows a negative resistance of $-90k\Omega$. The start-up time of the crystal oscillator is typically very large (~1s range) because of the high Q of the crystal circuit. The negative resistance of the amplifier also controls the start-up time and, the higher the negative resistance, the faster will be the start-up time of the crystal [19].

Power Consumption

The power consumption of the crystal oscillator is determined by the crystal and the design of the amplifier. The energy dissipating component in the crystal oscillator is the crystal's ESR. ESR dissipates energy in the form of heat loss, as Joule's heating, and is given by I^2R , where R is the value of resistance of the crystal's ESR and I is the RMS (Root Mean Square) current flowing into the crystal. This loss is directly proportional to the amplitude of oscillation. To reduce the loss and hence to reduce the power consumption of the crystal oscillator, the amplitude of oscillation

is often reduced. This can be done by operating the amplifier in the sub-threshold region [14][15]. Recent work applies circuit techniques where the amplitude of oscillation can be reduced by using delay locked loop (DLL) [13] or by simply quenching the oscillation amplitude [15]. These techniques have reduced the power consumption of a 32kHz crystal oscillator to 5.58nW [13], making it possible to use the crystal oscillator for BSNs, where a lower power clock is desired.

In this section we propose a 32kHz crystal oscillator circuit that consumes close to 1nW of power. We first operate the crystal oscillator in sub-threshold at $0.3V V_{DD}$ to reduce the overall power consumption to 2nW. We further reduce the power by applying a duty-cycling technique to turn-off the amplifier often. This technique brings down the over-all power consumption of the crystal oscillator to 1nW, improving the state of the art by almost five times. This proposed solution is more suitable for BSN applications.

2.3.1 Architecture

In this work, we designed the crystal oscillator operating at 0.3V V_{DD}, in the sub-threshold voltage. The transistor sizes of the amplifier determines the negative resistance and power dissipation; increasing the transistor size increases the negative resistance (R_N), but it also increases power. Decreasing the size makes $|R_N| < ESR$, failing to meet the oscillation criteria. The low voltage operation at 0.3V VDD provides the proper R_N for the oscillator. The amplifier consumes 5-10nA of quiescent current, which sets the power consumption of the crystal oscillator at 2-10nW.

In order for the oscillator to start, R_N > ESR. The start-up time, the time the oscillator takes to reach the full amplitude, is also controlled by R_N . The higher the value of R_N , the faster will be the startup [19]. However, a higher value of R_N means higher power consumption. The amplitude of oscillation saturates after start-up. After the oscillation saturates, the R_N of the amplifier decreases because of the non-linearity in the circuit due to saturation, and effectively $R_N = ESR$ at saturation. The saturation of oscillation creates higher harmonics, resulting in unnecessary power dissipation. While higher power is needed during start-up [19], it is not needed when the oscillation saturates. We propose further improvements in design to save this power.



Figure 2-22 Proposed control scheme, where oscillator is turned on and off to save power

The energy of a crystal oscillator is stored in its equivalent inductor and capacitor. After the saturation of oscillation, the stored energy in the crystal's equivalent inductor and capacitor is saturated. After the saturation, if the amplifier is disabled, the oscillation will start decaying, and if we enable it again, it will start growing again. The power consumption becomes negligible when the oscillator is disabled. However, oscillation doesn't die right away and decays slowly, with the

time constant given by ESR and Lm of the crystal. The output of the crystal oscillator is still useful and can be used to provide the clock when it is decaying. Therefore, the power consumption of a crystal oscillator can be further reduced by switching the amplifier. In our design, we switch the amplifier periodically, while keeping the amplitude of oscillation high enough for the receiver circuit to detect the oscillation. Figure 2-22 (a) shows the concept of the control scheme. When the amplifier is disabled, the oscillation at XI will decay with a time constant (TD), which is determined by the ESR and Lm. When the amplifier is enabled, it grows with a time constant (TG), which is determined by R_N –ESR and Lm [19]. For optimal power savings, the amplifier should be disabled for a time proportional to TD, and enabled for a time proportional to TG, as shown in Figure 2-22 (b). A counter running on oscillator output frequency is enabled when the amplitude crosses a set threshold. It counts until C1 and stops when the amplitude crosses a higher threshold. This gives us a digital o/p proportional to TG. Similarly, C2 proportional to TD can be obtained. A clock with the period (C1+C2) is obtained, with C1 as High and C2 as low, as shown in Figure 2-22 (b). The proposed technique enables a calibrated switching of the amplifier of the crystal oscillator and helps in cutting down the power further to 1nW.

2.3.2 Design of Low Power Amplifier

The architecture of the proposed crystal oscillator involved the design of a low power amplifier and control circuit to switch the amplifier periodically. To reduce the power consumption of the crystal oscillator, power at both stages of the design has to be reduced. This section talks about the design of the low power amplifier circuit. Various inverting amplifier architectures can be used to implement the amplifier. A simple push-pull inverter (a digital inverter) with a big bias resistor, as shown in Figure 2-20 (a), is one of the design options and is commonly used because it is single stage and hence consumes the least power [15]. However, the inverter circuit needs to be designed properly to meet the Barkhausen oscillation criterion. At lower driver strength (smaller sizes for nMOS and pMOS), the negative resistance of the amplifier is lower and cannot meet the oscillation criterion. Increasing the size increases the negative resistance. However, after a certain size, the negative resistance starts decreasing, because of the self-loading in the inverter through the gatedrain capactiance, C_{GD} (miller effect). Also, increasing the size of the inverter increases the power consumption. Therefore, the inverter needs to be sized properly for the power consumption, as well as for negative resistance. Apart from proper sizing, the power consumption can also be reduced by operating the amplifier circuit at lower V_{DD} voltage. At lower V_{DD} voltage, the amplifier can be sized to meet the oscillation criterion. The sizes for lower V_{DD} are typically bigger than the sizes for higer V_{DD} . However, the overall power consumption decreases.



Figure 2-23 Power consumption of the amplifier circuit with V_{DD} voltage for fixed negative resistance of -70k Ω

FFigure 2-23 shows the power consumption of the amplifier circuit designed to provide a fixed negative resistance of $-70k\Omega$ at 32kHz for different V_{DD} voltage. The power consumption increases with the voltage almost linearly. This is largely because the bias current of the amplifier almost

remains the same for a given R_N . As a result, power increases because of the increase in the V_{DD} . Therefore, the oscillator needs to be operated at a lower V_{DD} to reduce the power. We designed the inverter to operate at 0.3V V_{DD} and provide -70k Ω negative resistance at 32kHz as shown in Figure 2-23. We used LV_T transistors with longer length to implement the inverter. LV_T transistors give better performance or gain at 0.3V V_{DD} , owing to the lower threshold voltage, while a longer length of the transistor helps in reducing the bias the current.

Process Calibration



Figure 2-24 Calibration technique for amplifier to address variation effects

The amplifier was designed to operate in the sub-threshold region with a V_{DD} of 0.3V. Owing to the sub-threshold region of operation, the amplifier is sensitive to process variation. As a result, the negative resistance and power consumption of the amplifier can vary a lot. At some process corners, the amplifier can consume higher power and give very high negative resistance, and at other process corners, its negative resistance can be low, and it may fail to meet the Barkhausen oscillation condition. We propose a calibration method to address this variation. Figure 2-24 shows the calibration circuit. We set the drive strength of the amplifier transistors MP and MN using this circuit. The amplifier is enabled when ENP=0 and ENN=1. For calibration of MN to a given drive strength, ENN and ENP are set to one. This enables the calibration circuit where MN gets connected to an external resistor through the switch MNC. XI is connected to Ref, which is



Figure 2-25 Simulation result of the XTAL at 0.3V VDD and consuming 2nW power

selected to be at $V_{DD}/2$. The size of transistor MN is changed using the SAR logic and the comparator in the feedback loop. This happens in the following way. XI and Ref are set to $V_{DD}/2$ and the pull-down path is enabled, while the pull-up path is disabled. The external resistor Rc is connected to V_{DD} . If the size of MN is very big, then it will pull down the XO node below Ref which will cause the comparator output to go low. This low signal tells the SAR logic to reduce the size of the transistor MN. The size of MN is successively approximated, and the method is similar to the one explained in Section 2.2.3 for the on-chip clock source, and it takes 5 clock cycles. This algorithm is like a binary search for the right drive strength of the transistor MN set by the external resistor. This way MN can be sized to the right drive strength. Similarly, MP is sized by setting ENN and ENP to zero and connecting the external resistor to ground. We use a

size of external resistor such that the amplifier can be sized to supply 5-20nA of bias current, which provides enough drive strength to meet the Barkhausen criteria for oscillation. Figure 2-25 shows the simulation result of the oscillator.

2.3.3 Amplifier duty-cycling control implementation

To further reduce the power consumption of the XTAL, the duty cycling technique explained in Section 2.3.1 is implemented. Figure 2-22 shows the control scheme. In this control scheme, the amplifier explained in Section 2.3.2 is periodically turned on and off. When the amplifier is turned-off, the power consumption of the oscillator goes to 0.3nW, and it consumes 2nW when enabled. The oscillation grows when the amplifier is enabled and decays when it is disabled. The amplitude of oscillation during this time is kept high enough for detection. The on time for the amplifier should be proportional to the rate of growth of oscillation (TG). Similarly, the off time of the amplifier should be proportional to the decay of the oscillation (TD). The implementation of the control scheme involves design techniques to obtain the time constants TG and TD.

Obtaining time of growth of oscillation



Figure 2-26 Circuit to obtain the time of growth (TG) of oscillation

Figure 2-26 shows the circuit to obtain the TG of the oscillation. It consists of comparators and SR flip-flop. Threshold voltages VREFH=220mV and VREFL=200mV are applied at the negative terminal of the comparator, while XI is applied at the positive terminal. Once oscillation's amplitude goes above VREFL, the output of Comp1 goes high and corresponding SR flip-flop is set. This sets COUNTG to high. A counter is enabled using this signal to count. The amplitude of oscillation keeps on increasing. Once the oscillation crosses VREFH, Comp2 goes high and sets COUNTG to zero. This stops the counter and sets the value of the counter, which is proportional to the growth of oscillation. The value is digital and is stored, while the circuit is disabled to save power. The proposed circuit consumes active power only when the count value is needed.

Obtaining time of decay of oscillation



Figure 2-27 Circuit to obtain the time of decay (TD) of oscillation

Figure 2-27 shows the circuit implementation for obtaining TD for the oscillator. The circuit is very similar to the circuit used for obtaining TG. It also enables a counter, which counts when XI is between VREFH and VREFL. While TG is obtained when the amplifier is enabled, TD is obtained when it is disabled. Both TG and TD are stored digitally and their corresponding circuits

are disabled to save power. After obtaining TG and TD, the oscillator control turns on the amplifier for time=TD and turns it off for time=TG. This way, the power consumption is reduced and dutycycled. The total power consumption goes below 1nW. The next section talks about the complete control architecture of the circuit.

2.3.4 Complete circuit implementation

Figure 2-28 shows the complete circuit diagram of the proposed crystal oscillator circuit. First, the calibration of the amplifier is performed, which can be done once after manufacturing. The calibration circuit sets the drive strength of the amplifier and compensates for the process variation.



Figure 2-28 Complete circuit diagram of the proposed crystal oscillator circuit After the calibration, the time constant generation circuit obtains the time of growth (TG) and time of decay (TD) of the oscillator. These time constants are used to configure the clock, (DCCLK) to

switch the amplifier on and off. The duty cycle of DCCLK is determined by TG and TD, with high time=TG and low time=TD. Once the DCCLK is configured, the time constant generation circuit is disabled. Similarly, the calibration circuit is disabled after calibration and all the digital bits are stored. This eliminates the power overhead of the calibration circuit or time constant generation circuit. The power consumption is given by the amplifier with duty cycling. A clock buffer is used to level convert the clock to higher voltage if needed.



2.3.5 Results

Figure 2-29 Simulation result of oscillator duty-cycling technique

40

We implemented the crystal oscillator circuit (0.25mmx0.25mm) in 130nm CMOS. Figure 2-29 shows the simulation result of the proposed circuit. Initially, time constants TG and TD are obtained and DCCLK is held low. DCCLK is then configured with a time period of TG+TD with high time of TG and low time of TD. Once DCCLK is configured, oscillation switches on and off and achieves a power consumption of less than 1nW. Figure 2-29 shows that the oscillator's amplitude increases when DCCLK is high and decreases when it is low, achieving the duty-cycling effect in the design.

	[13]	[15]	[17]	[18]	This work
Operating Frequency	32kHz	32 kHz	32kHz	2.1MHz	32kHz
Area	0.3mm ²	N/A	25mm ²	0.41mm ²	0.0625mm ²
Power Consumption	5.58nW	22nW	220nW	700nW	1nW
Operating VDD	0.92-1.8V	0.71V	3V	3V	0.3V
Number of Power/Gnd	2/2	1/1	1/1	1/1	1/1
Amplitude of Oscillation	100mV	65mV	N/A	N/A	200mV
Technology	0.18µm	2µm	2µm	2µm	0.13µm

Table 2-1 Comparison summary of the duty-cycled oscillator with previous low power works

We implemented the crystal oscillator circuit (0.25mmx0.25mm) in 130nm CMOS. Figure 2-29 shows the simulation result of the proposed circuit. Initially, time constants TG and TD are obtained and DCCLK is held low. DCCLK is then configured with a time period of TG+TD with high time of TG and low time of TD. Once DCCLK is configured, oscillation switches on and off and achieves a power consumption of less than 1nW. Figure 2-29 shows that the oscillator's amplitude increases when DCCLK is high and decreases when it is low, achieving the duty-cycling effect in the design.

Table 2-1 shows the comparison summary of the proposed crystal oscillator circuit with previous work. Our crystal oscillator circuit consumes 1nW of power and has an area of 0.0625mm² .It has over 5X lower power and 8X lower area compared to [13], which is the lowest reported power crystal oscillator circuit. We also operated the circuit at 0.3V, which is the lowest reported power supply voltage for a crystal oscillator amplifier circuit. The circuit in [13] also uses a DLL based technique and employs two power supplies and two grounds to achieve lower power. Our circuit uses a single power supply. Previously reported work, such as [15] or [16], achieves a power consumption of 22nW and 18nW by reducing the amplitude of oscillation. We reduce the amplitude of oscillation by operating the circuit at 0.3V. The proposed circuit provides a low power, lower area crystal oscillator circuit. It applies lower voltage design in conjunction with a duty-cycling technique to achieve lower power suitable for BSN applications.

2.4 Conclusions

We presented an ultra-low power clock source platform for BSN application. It provides an onchip and an off-chip oscillator circuit utilizing a crystal oscillator. Proposed circuits utilize low voltage design and duty-cycling technique to achieve lower power. The ULP on-chip clock source uses a 1µW temperature compensated on-chip digitally controlled oscillator (Osc_{CMP}) and a 100nW uncompensated oscillator (Osc_{UCMP}), with respective temperature stabilities of 5ppm/°C and 1.67%/°C. We also presented a fast locking circuit that re-locks Osc_{UCMP} to Osc_{CMP} often enough to achieve high temperature stability. Measurements of a 130nm CMOS chip show that this combination gives a stability of 5ppm/°C from 20°C to 40°C (14ppm/°C from 20°C to 70°C) at 150nW, if temperature changes by 1°C or less every second. The proposed circuit exhibits similar stability to an XTAL with ~7X less power and no off-chip components. We also proposed a crystal oscillator circuit for a more precise and lower power solution in an off-chip implementation. The crystal oscillator circuit operates at 0.3V, with a power consumption of 2nW without duty-cycling and 1nW with duty-cycling, over 5X lower power and 8X lower area compared to the state-of-the-art. The proposed circuit has the least reported power and power supply for the crystal oscillator operating at 32.768kHz.

Chapter 3

Energy Harvesting and Power Management for ULP Systems

The previous chapter showed the clocking solution for ULP SoCs. The clock source is usually always on and adds to the standby power consumption of the system. The clocking scheme proposed in Chapter 2 provides low power solutions suitable for BSN applications. Another component of standby power consumption in a ULP SoC is the energy harvesting and power management circuit (EHM). While the voltage sources (V_{DDS}) are always needed for the SoC, the energy harvesting circuit also needs to be on to tap into dynamically varying harvesting sources. Further, in order to increase the overall life time of the system, the EHM circuits need to be more efficient and operate at lower output voltage.

ULP SoCs, like BSN, can now operate from harvested energy obtained from ambient sources such as sunlight, ambient light, vibration, RF or the body heat of a human being. These sources provide low energy due to the constrained operating condition. It is vital that the energy be harvested efficiently and managed with minimum loss, in order to enable battery less operation of ULP systems. It is also vital that the energy harvesting and management circuit do not increase the cost or bill of materials (BOM) of the BSN, which can hinder their wide spread deployment. Conventional approaches to obtaining and managing energy from ambient sources either employ more than one inductor, which increases the cost and bill of material (BOM), or use low drop out (LDO) regulators, which have lower efficiency. The lower efficiency of LDOs can result in the loss of energy, which is not desirable. There is a need to provide an efficient and low cost energy harvesting and power management solution. This chapter proposes a highly efficient energy harvesting and power management solution with various design options. The proposed solution can harvest energy from TEG, and solar cells. It proposes a single inductor power management circuit to provide regulated output voltages for BSN, which helps in reducing the cost of the power-management circuit. It also presents a bandgap reference voltage circuit which operates at lower voltage and consumes lower power than state of the art reference voltage circuits. The lower voltage operation of the bandgap reference circuit enables a lower start-up or power-on voltage for the body sensor node, which helps in increasing the life-time of BSN node.

3.1 Energy Harvesting and Power Management System



Figure 3-1 Energy harvesting and power management system

Figure 3-1 shows the typical block-level architecture of an energy harvesting and power management (EHM) system for an ultra-low power system such as a BSN. The functioning of such

a system can be broadly divided into two categories; harvesting energy and power management. The system can harvest energy from ambient sources, such as a thermo-electric generator (TEG), solar cell or a photovoltaic cell (PV), RF sources, or vibration, using a piezoelectric harvester. Usually, the harvested energy is stored on a capacitor. The power management circuit then generates multiple meaningful voltages (V_{DD}s) for the SoC. These output voltages are regulated using the reference voltage. The purpose of the reference voltage is to supply a well-controlled absolute reference voltage. The reference voltage is also used to supply the clamp circuit for the energy harvester. The purpose of the clamp circuit is to prevent the output voltage on the storage capacitor from going higher than a set maximum for its reliable operation.

The design and architecture of the EHM system impact the overall operation and lifetime of the BSN. For a longer lifetime of the node, various design and architectural requirements are to be met. For example, the system should be able to harvest energy efficiently at very low levels of output power from the ambient sources. Another requirement is that the system should start-up at lower initial voltage on the storage capacitor. This is limited by the output voltage at which the reference voltage starts-up. A lower start-up voltage on the storage capacitor increases the life time of the node because more energy on the capacitor can be used. The regulated output voltages coming out of the power management circuit should also be efficiently regulated and their loss is to be minimized. The system includes several bias circuitries, including the reference voltage, which consumes quiescent power even in the absence of any load on the SoC. This power is used for the control circuitry in the EHM system. It is very important to reduce the static power consumption for the longer life time. Finally, the cost of implementation of such a system should also be very minimal for its widespread deployment.



Figure 3-2 Energy harvesting and power management system in a BSN SoC [1]

Figure 3-2 shows the power management block of a state-of-the-art BSN chip [1]. The implementation of each block on the system is not the existing state-of-the-art for that given block. However, the implementation presents all the blocks needed for EHM and is the state of the art solution for the complete implementation of EHM for a BSN SoC. The solution provides an energy harvester, as well as power management, to provide multiple V_{DD} rails for the SoC. It also includes a reference voltage circuit, Bandgap reference for voltage regulation. It uses an inductor based boost-converter circuit to harvest energy from a thermo-electric generator (TEG). The energy harvester can harvest from an input voltage of 20mV or 1-2°C temperature difference on TEG. The Bandgap reference circuit provides a 0.5V reference voltage for power management circuits and for the voltage clamp on the boost converter. It starts at the storage voltage of 0.9V and consumes a power of 200nW. Therefore, the EHM circuit on [1] cannot start before the voltage on the storage capacitor goes to 0.9V. The EHM circuit's standby power consumption is also greater than 200nW. Further, the existing power management solution for the chip includes multiple linear regulators and a switched capacitor regulator to supply different V_{DD} rails. A linear regulator such as an LDO suffers from poor efficiency. The existing power management solution for the

chip includes multiple linear regulators and a switched capacitor regulator to supply different rails. The theoretical maximum efficiency of an LDO cannot be more than V_{DD}/V_{BOOST} . Therefore, more than 63%, 26% and 12% of energy on the V_{DD} rails, 0.5V, 1.0V, and 1.2V gets wasted in obtaining the regulated V_{DD} voltage. There is a need to provide energy efficient solutions for the power management, which can increase the lifetime of the BSN chip. This work proposes a highly efficient energy harvesting and power management solution with various design options for BSNs. It proposes novel circuits for each component of the EHM and improves the state of the art for each one of them.

3.2 Energy Harvesting for Thermo Electric Generator

There are various ambient sources for harvesting energy. However, the following four sources are commonly explored for ULP systems. These are light, vibration, thermal, and RF. Energy harvesting from outdoor sun-light is a well explored research area. Usually, these harvesters are much bigger in size. They are used to harvest large amounts of power (in kWs) at high voltages. These converters achieve high efficiency with very good maximum power point tracking [20]-[22]. Recently, interest in micro-power energy harvesters using solar cells has emerged for their use in ULP systems, such as WSNs or BSNs energy harvesters [23]-[25]. These are used for low power systems. Usually, they are small in size and harvest very low amounts of energy (in μ Ws), due to the smaller size of the solar cell and low ambient light because of the indoor lighting condition. Another harvester that is commonly suited for BSN application is the piezoelectric harvester, harvesting mechanical energy from vibration. These harvesters can harvest 10-100's of μ W of available power [26]. Electrical energy is generated when mechanical stress is applied on a piezoelectric material, which underlines the physics behind energy harvesting. Recently,

piezoelectric harvesters suitable for BSNs have been demonstrated. These harvesters can harvest output powers of a few μ W to 100's of μ W, at efficiencies above 80% [27]-[29]. Harvesting energy from available RF power is another means of obtaining energy for BSNs. However, RF energy is not true energy harvesting, as RF signals are powered from other sources. Nevertheless, RF can also supply enough energy to power BSNs [30]-[32]. Finally, thermoelectric generators (TEG) can also be used to harvest energy from thermal gradients existing in our surroundings or systems. TEGs have found widespread use in industrial space, where large temperature gradients, such as heat exhausts, are common [33]-[35]. These are usually big in size and harvest energy at the rate of a few 100's of Watt. Smaller TEGs have also been designed to be used for ULP SoCs. One of the application of TEG is harvesting energy for SoCs that are worn by the human body. 1-10°C of temperature difference can exist between a human body and its ambience. We can harvest 100's of μW of available power [36][37]. Recent TEG energy harvesters can harvest energy from very low temperature difference, usually 1-2°C, resulting in an input voltage of less than 50mV [38]-[41]. In this chapter, we present an energy harvester using a TEG that can harvest from 10mV input voltage. We also present a control scheme for improving the efficiency of the harvester. We will first explain the functioning of a thermo-electric generator and then dive into the design details of the extraction circuit.

3.2.1 Functioning of TEG

A thermoelectric generator converts the thermal energy arising from the difference in temperature between two conductors into electrical energy and vice-versa. The physics behind the thermoelectric phenomenon is the Seebeck effect. The Seebeck effect is the production of an electromotive force (emf) and, consequently, voltage or current between two dissimilar conductors, when their junctions are maintained at different temperatures. The conductors can be metals or semiconductors and they need not be solid. Apart from generation of emf, thermo-electrics are also used for measuring temperature, as well as for heating or cooling. When electricity is passed through the junction of two different conducting materials, heat can either be generated or taken away (cooling). This effect is called the Peltier effect. Similarly, if a temperature difference exists between two metals, the amount of electromotive force generated is proportional to the difference in temperature. By measuring the produced voltage or current, the temperature can be measured.

The thermoelectric materials used for generating electricity need to be good conductors of electricity, as a scattering effect can generate heat on both sides of the barrier. Also, the material needs to be a poor conductor of heat; otherwise, the temperature difference that must be maintained between the hot side and cold side produces large heat backflow. Materials which optimize these electrical and thermal properties fit the bill. The highest performance has been shown by heavily doped semiconductors, like Bismuth-telluride or Silicon-Germanium [42]. The semi-conductor material also needs to form a base on which both n-type and p-type semiconductor can be generated. Segments of p-type– and n-type–doped semiconductor materials, such as suitably doped bismuth telluride, are connected together to form an electric circuit. The shunts are made of an excellent electrical conductor, such as copper. A voltage drives a current through the circuit, passing from one segment to another, through the connecting shunts. For determining efficiency, this configuration is equivalent to the electrons passing directly from one thermoelectric material to the other. Conventional thermoelectric cooling/heating modules are constructed of thermoelectric segments repeated many times, and organized into arrays such as the one shown in Figure 3-3. When current flows within the module, one side is cooled and the other is heated. If the current is

reversed, the hot and cold sides reverse also. The geometry for power generators (Figure 3-3) is conceptually the same. In this case, the top side is connected to a heat source and the bottom to a heat sink. Thermoelectric power generators often are similar in physical form to cooling modules, except that fewer taller and thicker elements are used.



Figure 3-3 Practical Thermoelectric generator connecting large number of *pn* junction to increase the operating voltage [42]

The Seebeck effect in the semiconductor material creates a flow of excess electrons from the hot junction in the n-type material to the cold junction. In the p-type material, holes migrate toward the cold side, creating a net current flow in the same direction as that of the n-type material. The measure of the performance of the material used in TEG is the Seebeck coefficient, defined as the change in voltage per degree of temperature change.

$$S = \frac{dV}{dT} \tag{3.1}$$

Since heat flows from the top to the bottom, all of the thermoelectric legs are thermally connected in parallel. In the power-generation mode, heat flowing from the top to the bottom drives an electric current through an external load. The voltage obtained at the output of the thermal harvester is proportional to the temperature difference across the thermoelectric element. For body-worn devices, the temperature difference is very small and an output voltage of few mV - 50mV is all that can be produced using a 10cm² thermal harvester. The energy harvester should be able to harvest energy from such low input voltages. Several circuit design challenges had to be overcome to be able to harvest from such low voltages. In this chapter, we present a TEG harvester that can harvest from an output voltage of 10mV and above.

Electrical Characteristics of TEG



Figure 3-4 Equivalent circuit of a thermoelectric generator

The thermoelectric generator can be modeled as a voltage source in series with an input resistance [43][44]. Figure 3-4 shows the equivalent circuit of the TEG. The open circuit voltage V_{TEG} is directly proportional to the temperature difference between the hot and cold sides, and is given by,

$$\boldsymbol{V}_{\boldsymbol{T}\boldsymbol{E}\boldsymbol{G}} = \boldsymbol{S}\Delta\boldsymbol{T} \tag{3.2}$$

where *S* is Seebeck coefficient given by equation (3.1), and ΔT is the temperature difference between the hot and cold sides of TEG. Commercial TEGs use semiconductor material bismuth

telluride due to its good electrical conductivity and poor thermal conductivity [42]. The Seebeck coefficient of an n-type bismuth telluride material is -287μ V/K at 54°C. The lower voltage coming out of one TEG cell requires that multiple cells be used in series to increase the output voltage, as shown in Figure 3-3. Connecting more cells in series increases the input resistance of the TEG and can reduce the efficiency. Also, a smaller size TEG is needed for body-worn devices. Therefore, the output voltage of the TEG can be very small, at times a few mV only. This chapter presents a design to harvest energy from such low output voltages in TEG material.

Maximum Power Point in TEG



Figure 3-5 Energy harvester harvests maximum energy when impedance of the harvester matches TEG's input resistance

Figure 3-5 shows the condition when a harvester is connected to the TEG. The harvester will present a load with output impedance Z_{LOAD} to the TEG. The power drawn from TEG is dependent on the value of the Z_{LOAD} . When Z_{LOAD} is very high, the current drawn by the harvester will be very small and so the output power will be small. Similarly, if the Z_{LOAD} is low, output voltage will be small and, consequently, the output power will be small. From basic electrical engineering, we know that the maximum power will be delivered to the load when,
$$\boldsymbol{Z}_{\boldsymbol{LOAD}} = \boldsymbol{Rin} \tag{3.3}$$

Figure 3-6 shows the output characteristics of an energy harvester connected to TEG. It shows the output power and output voltage as functions of load current. The output power peaks and a maximum power point exists for the load. The maximum power point happens when the load matches the input resistance Rin by satisfying equation (3.3). At this operating condition, the output voltage Vin in Figure 3-5 is given by,



$$Vin = V_{TEG}/2 \tag{3.4}$$

Figure 3-6 Maximum power point characteristics of a TEG energy harvester

The ambient condition in which a thermoelectric generator operates is dynamic. For example, the temperature difference between the hot side and the cold side in a TEG can change. This can lead to a change in the open circuit voltage of the TEG. As a result, the maximum power point (MPP) of a TEG is a dynamic quantity that changes with time. The energy harvester should always track

and operate at the maximum power point to maximize the harvested energy. In the proposed solution, we present a maximum power point circuit that continuously tracks and always operates the energy harvester at the MPP point.

3.2.2 Design of the TEG Energy Harvester



Figure 3-7 proposed design architecture for the TEG harvester using a boost converter.

Figure 3-7 shows the architecture of the proposed energy harvester for TEG. It uses a boost converter to harvest energy. The boost converters are switching mode power converters and they operate on the following basic principle [45]. Referring to Figure 3-7, first, the signal LS goes high and transistor M_{LS} is turned on. This connects the inductor between TEG and ground. As a result, the current in the inductor starts rising and starts storing energy. It obtains its energy from TEG. After a well-defined time, LS is brought down to ground, and M_{HS} is turned on. Now the stored current in the inductor discharges on the capacitor and raises the voltage of V_{CAP} . This switching pattern repeats over and over to charge the output voltage to a higher voltage. The above

description of the operation of the boost converter is very basic, and a more detailed description of the proposed converter in the context of the challenges for low voltage TEGs will follow in succeeding sections. However few important design topics can be introduced here for the reader.

In the ideal case, the energy that is taken from the TEG should be stored on V_{CAP} without loss. However, this is not possible in a practical system, and some energy losses are involved. The efficiency of the converter, defined as the ratio of power delivered to the load to the power obtained from the source, is the measure of the performance. Some of the losses that are involved are explained as follows. The switches M_{LS} and M_{HS} have finite resistance, and they carry current during the switching cycle. As current flows through a resistor, conduction loss takes place in the form of Joule's heating in the switches. Also, the switches M_{LS} , M_{HS} and other circuits in the converter, switch in a periodic fashion to harvest energy. This switching also involves energy loss and is termed simply as switching loss. Finally, various bias current circuits are needed to implement the control of the converter. This results in power loss that is always present and is called static loss. In order to maximize the efficiency of the converter, all these losses are to be minimized. Finally, maximum power can be harvested from the TEG when the efficiency of the boost converter is at its maximum, and the boost converter's operating point is always the maximum power point of the TEG.

The proposed architecture of Figure 3-7 has three components. It consists of an MPP tracking circuit to operate the converter at the maximum power point. The second component of the design is the boost converter. The boost converter is designed to have maximum possible efficiency for the given operating condition, while the MPP tracking circuit tracks the MPP point of the TEG

and operates at that point. The next section talks about the architecture of the MPP tracking scheme. The section after that, talks about the design of the boost converter control circuit.



Maximum Power Point Tracking Circuit

Figure 3-8 Maximum power point tracking circuit for TEG

Figure 3-8 shows the maximum power point tracking circuit for the energy harvester connecting to the boost converter. It consists of an MPP sampling circuit, a comparator and a clock generator. The clock generator generates a clock with periods of 150ms, which is fast enough to track the slow changing ambient condition, as well as the leakage of the V_{MPP} capacitor. This clock is used to generate a pulse that goes low for 10ms shown in Figure 3-9. 10ms is large enough time to charge a 10nF cap using a 100k Ω resistor. When this pulse goes low, the boost converter is disabled and, as a result, the load current goes to zero. Since no load is connected to the output of TEG, the TEG voltage goes to its open circuit voltage V_{TEG}. This pulse is also used to close the switches S₁ and S₂, as shown in Figure 3-8. The two resistors get connected to the TEG voltage and the V_{MPP} node goes to the voltage V_{TEG}/2 through the resistor divider. Therefore, the capacitor C_M gets charged to V_{TEG}/2. The output voltage of the V_{MPP} node is maintained by the cap C_M. The

MPP sampling circuit samples the half of the TEG's open circuit voltage and stores it on the cap. After the pulse goes high again, the boost converter starts drawing the current from the TEG. As it draws current from the TEG, the voltage level of *Vin* will start decreasing. The boost converter increases the current drawn from the TEG source by increasing the frequency of switching. As long as the output of the comparator is high, the boost converter switches at higher frequency and the current drawn from the TEG source is high.



Figure 3-9 Simulation Result of MPP tracking circuit showing

Figure 3-9 shows the simulation result of the MPP tracking circuit. As the current is drawn, the output voltage of TEG goes low, and eventually reaches V_{MPP} value, which is set to $V_{TEG}/2$. As it crosses V_{MPP} value, the comparator output goes low and the boost converter is disabled. As the boost converter is disabled, the *Vin* voltage will start going higher and it will go above V_{MPP} . At this point, comparator output goes higher and the converter is enabled again. This way the output voltage of TEG is maintained at $V_{TEG}/2$, which is the maximum power point of the circuit given by the equation (3.4). The output voltage of the TEG is maintained at its maximum power point by the control circuit with a voltage ripple around it. The amount of ripple is a function of the

capacitor connected at TEG. The ripple can be made negligibly small by connecting a large capacitor. We use 5μ F cap at the output. This way the circuit is maintained at its maximum power point. The output voltage is sampled frequently using the clock, which helps keep track of the TEG's open circuit voltage to account for the dynamic changes in the operating condition.



3.2.3 Control Circuit for Boost Converter

Figure 3-10 Timing diagram of the control signal for the boost converter showing use of two phases $\phi 1$ and $\phi 2$, where $\phi 1$ is used for cancelling mismatch to harvest at low voltage

The boost converter design involves charging of the inductor current and storing it on the capacitor on V_{CAP} . The signals LS and HS, as shown in Figure 3-10, are used to accomplish this switching. The signal LS goes high to turn on M_{LS} transistor in Figure 3-7. This starts charging the inductor current. After a well-defined time, which sets the inductor peak current, the inductor current is discharged on the capacitor, raising V_{CAP} . This is done when HS goes low, as shown in Figure 3-10. Usually, LS and HS signals are controlled to control the performance in a boost converter. However, the proposed boost converter is designed to harvest energy from very low input voltage, such as 10mV. The effect of mismatch or non-idealities in the circuit needs to be addressed if energy is harvested at low voltage levels. The mismatch between the threshold voltage

of two transistors can itself be as high as ~50mV. It is not possible to harvest energy at low voltages if the mismatch between the devices in circuits such as the comparator is not compensated. Therefore, the energy harvesting control scheme is broken down into two phases. In phase one, designated by control signal ϕ 1, the mismatch throughout the design is compensated for. Phase one is followed by phase two ϕ 2, which includes control and generation of signals LS and HS. Figure 3-10 shows the timing diagram. We will now explain the circuits needed to control the LS and HS signals.





Figure 3-11 Efficiency variation with peak inductor current for various input voltages

The low side control signal (LS) controls the amount of energy harvested from TEG. During the time when LS is on, the inductor current increases and reaches peak value, I_{PEAK} . The value of the peak current depends on the on-time of the LS switch M_{LS} . The longer the on-time, the higher the

peak current would be. For micro-power design, all the inductor current is discharged into the storage capacitor on V_{CAP} . So, the energy transferred from TEG in each cycle is given by,

$$E = 0.5 * LI_{PEAK}^2 \tag{3.5}$$

Losses are involved when a transfer of energy happens. These are switching and conduction loss. The conduction loss is dependent on I_{PEAK} . The efficiency of the converter greatly depends on the value of I_{PEAK} . Figure 3-11 shows the variation of the efficiency with I_{PEAK} for different input voltages. At lower value of I_{PEAK} , the switching loss in the converter dominates and lowers the efficiency. This happens because the energy transferred, given by equation (3.5), is small compared to the switching loss. At higher values of I_{PEAK} , the conduction loss increases greatly to reduce the efficiency. There are values of I_{PEAK} where the efficiency can be maximized. Therefore, it needs to be well-controlled (See Appendix B for analytical details).

It is a common practice, in switch mode power supply design, to control the peak inductor current for maximizing efficiency (See Appendix B). In the conventional approach, the resistance of the MOS transistor is used to measure the peak inductor current. The drop across the LS transistor M_{LS} , when the inductor is on, is the representation of peak inductor current. However, this method is sensitive to mismatch and can result in $\pm 20-40\%$ error in the peak inductor current [46]. Also, this method consumes higher power, which can reduce the efficiency by more than 3% [46]. Usually, LS is turned on for a fixed time [38][39] to control the peak inductor current. This method is not expensive in terms of power. However, the efficiency can be reduced because the inductor current is not controlled. Also, the process variation can result in a variation of the peak current further impacting the efficiency of the converter. We propose a low power control scheme to control the peak inductor current. The proposed control scheme is one of the first implementations of peak inductor control for micro-power harvesters. The proposed scheme has very small dependence ($\pm 2.5\%$) because of the mismatch between the devices. We also propose means to control the process variation or voltage variation.



Figure 3-12 LS control circuit to generate the constant peak inductor current Figure 3-12 shows the circuit that is used for generating constant peak inductor current. The circuit generates the timing for LS signal. The value of the peak inductor current can also be programmed by this circuit. It uses phase 1 and phase 2 of the switching cycle to generate the timing for LS. The functioning of this circuit is explained with the help of the following timing diagram.



Figure 3-13 Timing diagram for LS control circuit

The circuit operates in the following manner. During phase $\phi 1$, node *a* is connected to V_{IN}, which the output voltage of TEG, V_{CLS} is held to ground and comparator C1 is disabled. The output of

LS is set to ground. The transistor M_{P1} is made weak, which sets the output voltage of node *b* and is given by $V_{IN}+V_{TM1}$, where V_{TM1} is the threshold voltage of the transistor M_1 . Weak M_{P1} drives M_1 very close to its threshold voltage. In phase $\phi 2$, node *a* is connected to ground, while the connections between *b* and *c* are removed. The transistor M_1 is designed to be in saturation and sets the current I_{LSct1} which is given by,

$$I_{LSctl} = k * (V_{IN} + V_{TM1} - V_{TM1})^{2}$$
$$I_{LSctl} = k * (V_{IN})^{2}$$
(3.6)

The current generated for LS control is made proportional to the square of the input voltage. It is used for the generation of LS timing. This current is mirrored to charge the capacitor C_{LS} . As $\phi 2$ goes high, the capacitor starts charging. The timing of LS is given by the charging of the capacitor. Once $\phi 2$ goes high, LS goes high and the capacitor starts charging. Once the capacitor C_{LS} crosses V_{IN} , output of the comparator C1 goes low, which resets the flip-flop and brings LS to ground. Figure 3-13 shows the timing diagram for the generation of LS. The timing of LS is given by,

$$I_{LSctl} = C_{LS} \frac{dv}{dt}$$

as the capacitor charges from 0 to V_{IN} .

$$T_{ON_LS} = C_{LS} \frac{V_{IN}}{I_{LSCTI}}$$

Using equation (3.6), we get

$$T_{ON_LS} = \frac{C_{LS}}{kV_{IN}}$$
(3.7)

This is the time for which the inductor is kept on. It increases as V_{IN} decreases. Now, let us find out how the peak inductor current is obtained. Assuming negligible drop across M_{LS} , we can write the basic equation for inductor,

$$L\frac{di}{dt} = V_{IN}$$

The inductor current charges from 0 to IPEAK during the time LS is on.

So,
$$I_{PEAK} = \frac{V_{IN}T_{ON_LS}}{L}$$

Using equation (3.7), we get

$$I_{PEAK} = \frac{C_{LS}}{kL} \tag{3.8}$$

Equation (3.8) gives the formula for peak inductor current. The proposed circuit reduces it into a very compact form. The expression shows that the I_{PEAK} is independent of V_{IN} and V_{CAP} voltages. It depends on the value of the capacitance CLS and the inductor value L. By controlling the value of C_{LS}, peak inductor current can be set to a constant value, which gives the maximum efficiency, as shown in Figure 3-11. The constant k in the expression is a function of process and temperature. As a result, the peak inductor becomes a function of variation in process. The capacitor C_{LS} and the inductor L are the design components. They can be changed to compensate for the process variation. Often, it is not practical to change the value of inductance, either because of the cost involved or because of the size, as it is an off-chip component. However, the capacitor is on-chip and can easily be controlled. Therefore, we propose to address the variation using the capacitor C_{LS}. We use 5 bit binary control on the capacitor to address the process variation. Figure 3-14 shows the simulation result for the peak inductor current at different values of V_{CAP} and V_{IN}. Simulation shows that the circuit shows very small dependence on V_{CAP} or V_{IN}. The peak current value decreases at lower values of input voltage V_{IN}. The capacitor C_{LS} can be trimmed to increase the peak inductor current. The inductor current can also saturate at lower V_{IN} voltages. The switch resistance has to be small to avoid this. Also, the peak inductor current should be reduced to avoid saturation if it doesn't reduce the efficiency (see Figure 3-11).



Figure 3-14 Simulation result of peak inductor current with V_{IN} and V_{CAP}

High Side (HS) Control Scheme: Zero Detection

The previous section shows the method to generate a set peak inductor current to maximize the efficiency. Boost conversion also requires the transfer of this energy stored in the inductor to the capacitor on V_{CAP} shown in Figure 3-7. This is achieved by controlling the high side switch of the converter. The HS switch needs to be well-controlled to achieve better efficiency. The HS conduction happens in the following way. Once the inductor has charged to I_{PEAK} , the HS signal goes low and the transistor M_{HS} turns on. The inductor current starts discharging on to the capacitor. At this point, the node V_X goes above V_{CAP} to satisfy the conduction. As the capacitor charges the inductor current reduces and the V_X voltage drops. The inductor current eventually goes to zero and V_X becomes equal to V_{CAP} . The switch M_{HS} needs to be turned off at this point or the inductor current reverses direction and starts taking charge away from V_{CAP} , which will reduce the efficiency. Similarly, if the switch is turned off before the inductor current goes to zero, the remaining current discharges through a high impedance diode, which also hurts the efficiency. Therefore, an inductor current's zero crossing needs to be detected for accurate timing. This is usually termed as zero detection. Figure 3-15 shows the circuit for generating the HS timing using the zero detection. Comparator C2 is used and it compares the V_X node with V_{CAP} . Once V_X crosses V_{CAP} and goes below it, the comparator output goes low and turns off the switch M_{HS} . The previous works on micro-power converters have considered comparator-based zero detection very high power [38][39]. They do not implement it and instead use a correction technique for zero crossing. In this work, we present the comparator based zero detection scheme as shown in Figure 3-15. There are several challenges involved in designing this for a micro-power converter. We address these challenges by proposing new circuit architectures for the comparator block.



Figure 3-15 Circuit to generate the HS timing using zero inductor current crossing The HS signal is generated using the zero detection comparator C2, by monitoring the node V_X as shown in Figure 3-15. The design of the comparator plays a very important role in zero detection. Several issues need to be addressed. First, the performance of the comparator needs to be very good. If the delay through the comparator is high, the exact timing of HS control cannot be met. Second, the power consumption of the comparator needs to be small, as it will add to the loss in the boost converter. Finally, the mismatch between the devices inside the comparator can result in high offset which can alter the zero detection. This is particularly critical for harvesting from very low input voltage, as offset itself can be much higher than the input voltage, V_{IN}. Therefore, the offset due to mismatch needs to be cancelled. This work proposes a comparator that addresses all the above issues.



Figure 3-16 Schematic of zero detection comparator with offset cancellation scheme

Figure 3-16 shows the circuit diagram of the zero detection comparator. It uses a common gate amplifier, which is biased at 20μ A of quiescent current. The use of a common gate amplifier provides very good performance. However, if this comparator is always on, then the static power of the boost converter will become high and it will result in very low efficiency at low input voltages of V_{IN}. The comparator is duty-cycled to address this. It is turned on only when the switching happens. For example, during MPP tracking the switching happens at low frequency.

Every time a switching cycle is activated through the MPP comparator of Figure 3-7, the zero detection comparator is turned on, and it is turned off after the completion of zero detection. Therefore, the comparator is on only during the switching cycle and its power consumption is the component of the switching loss. The power overhead because of the zero-detect comparator is greatly reduced. It reduces the efficiency of the converter by ~2% at V_{IN} of 10mV and by ~0.3% at V_{IN} of 100mV.

While the performance and power issues are addressed as explained above, the circuit also addresses the offset issues. The offset in the design is cancelled in phase $\phi 1$ of the switching cycle. It works as follows. In $\phi 1$, nodes d and e are set at V_{CAP} while f is set at V_{SS}, and switch T1 is turned on while T2 is off. The feedback from node g sets V_{OFFSET} to remove offset in the comparator. If there is no offset in the design, V_{OFFSET}=V_{REF}. The measured comparator offset after compensation was <1mV. The other comparators in the converter use a similar offset compensation circuit. After offset compensation, zero detection is performed. Once LS goes low, the comparator is enabled in phase ϕ^2 . Node d is connected to VCAP while e gets connected to V_X and switch T1 is turned off while T2 is on. The comparator is configured to normal operation. As inductor current decreases, the node V_X starts going down. Once V_X crosses V_{CAP}, the common gate amplifier changes state and comparator output goes low, which turns off M_{HS}. The higher performance and offset cancellation method ensures correct zero detection. The measured V_X waveform (Figure 3-22 in Section 3.2.3) shows no overshoot or undershoot at t1 indicating ideal zero detection, which confirms that the HS timing is correctly turning off MHS when I_L=0. *Start-up in the boost-converter*

The boost converter circuit needs a start-up voltage. It is not possible to start charging up from low input voltages such as 10mV from TEG. Several start-up techniques are proposed in literature. One of the ways to start-up is to implement a mechanical switch which initially transfers charge to raise the voltage on storage node high enough to run the circuits reliably [39]. Another method to start-up can be to use a ring oscillator on V_{TEG} . The oscillator can be used to implement switching of the inductor and can charge the V_{CAP} voltage [40]. However, this technique requires a higher voltage at V_{IN} . These circuits start at voltages higher than 330mV [40]. The boost converter can also be started up using transformers [41]. They can start-up at voltages of 40mV. However, transformers are often very big in size. Our start-up circuit consists of the RF kick-start circuit method proposed in [1]. It consists of a 30-stage RF rectifier, The RF rectifier circuit receives the RF signal from outside and starts charging the V_{CAP} voltage. The boost converter circuit starts-up at 500mV.

3.2.4 Prototype Implementation and Measurement

The circuit of the boost converter was implemented in 130nm CMOS process. Figure 3-17 shows the die photo of the circuit. We targeted the resistance of V_{IN} to V_{SS} line to 300m Ω , which included bond-wire, inductor's parasitic DC resistance, board's trace resistance and resistance of the MOS transistor M_{LS}. The inductor used for the design was a 10µH coil-craft inductor. The total area for the implementation was $0.12\mu m^2$. We used three IO pads for V_{SS} and two IO pads for V_{CAP} . Figure 3-18 shows the maximum power point tracking scheme. It shows the pulse period where the boost converter is disabled and switching stops. The maximum power point voltage is sampled at this point and stored on the capacitor, which is used for the implementation of control.



Parameter	Value			
Technology	130nm bulk CMOS			
Total area	600 µm x 200 µm			
Inductor value, L	10 µH			
Designed V _{IN} -to- V _{SS} M _{LS} resistance	300 mΩ (Target)			
Inductor DCR	70 mΩ			
Bondwire	50 mΩ (1mm,1mil,gold)			
NMOS Res	120 mΩ			
Layout parasitic	60 mΩ			

Figure 3-17 Die Photo and design parameter of the boost converter



Figure 3-18 Maximum power point tracking pulse for MPP CLK



Figure 3-19 Measured TON of low side LS switch w.r.t. V_{CAP} and V_{IN} indicating peak inductor current control scheme

Figure 3-19 shows the measured LS switch pulse width with V_{CAP} and V_{IN} . The LS pulse width is directly proportional to the peak inductor current. The Figure shows small dependence of peak inductor current change with V_{CAP} for a given V_{IN} . With decreasing V_{IN} , the LS time increases. As I_{PEAK} is directly proportional to V_{IN} , decreasing V_{IN} requires that T_{ON} should increase to maintain a constant peak inductor current. Figure 3-19 shows increasing T_{ON} with decreasing V_{IN} .

Figure 3-20 shows the efficiency measurement of the boost converter at various input voltages V_{IN} . The efficiency is measured by changing the time period of the LS switch. It shows that a peak efficiency point exists for each value of V_{IN} . At very low value of T_{ON} , the peak inductor current is low and switching loss dominates, resulting in lower efficiency, with at higher value of T_{ON} peak inductor current is high and it causes higher conduction loss, again reducing the efficiency. The T_{ON} time for the peak efficiency point for each V_{IN} increases with decreasing V_{IN} . measurements agree with the requirement to control I_{PEAK} to maximize the efficiency, as shown in Figure 3-11.



Figure 3-20 Measured efficiency with Low Side T_{ON} , showing increased efficiency at Higher T_{ON} for Low V_{IN} . It achieves 84% efficiency and 0.4V V_{IN} and 22% efficiency at 10mV V_{IN}

Figure 3-20 shows the measurement of efficiency for low input voltage and supports maximum of 10mA load current. The converter achieves a peak efficiency of 84% at an input voltage of 0.4V. It is able to harvest at input voltages as low as 10mV, with an efficiency of 22%. This is the least reported voltage from which energy can be harvested in a boost converter. It also improves over the state-of-the-art boost converter harvesting at 20mV [38]. The efficiency of the converter in [38]

is 48% at 20mV, while our work achieves an efficiency of 53% at 20mV. Figure 3-21 shows the Monte-Carlo simulation result of the peak inductor current. The peak inductor current varies with process. This is compensated by trimming the timing capacitor C_{LS} in the design.



Figure 3-21 Variation of peak inductor current with process using Monte-Carlo simulation

Figure 3-22 shows the result of the zero detection scheme. The optimal zero detection helps in achieving higher frequency. The behavior of the node V_X during switching indicates the performance of zero detection. If the switch M_{HS} is opened before or after inductor current goes to zero, the output at the node V_X will over-shoot or under-shoot. For example, suppose the inductor was still carrying current when M_{HS} was opened. As a result, the low impedance of the switch is replaced by the high impedance of the diode. Therefore, the drop between V_X and V_{CAP} must increase. Therefore, the V_X node overshoots. Similarly, if the switch is turned on for a longer time and the current crosses zero and changes the direction and starts removing charge from the V_{CAP} voltage, then the V_X node will undershoot. Absence of overshoot or undershoot upon opening the switch is the indication of optimal zero detection. Figure 3-22 shows the measured zero detection

for various combinations of V_{IN} and V_{CAP} . Node V_X in all the waveform does not overshoot or undershoot, indicating ideal zero detection.



V_{IN}=0.3V, V_{CAP}=0.75V



V_{IN}=0.015V, V_{CAP}=0.7V







Figure 3-23 Measured operation of the boost converter with VIN=10mV and VIN=8mV.

Figure 3-23 shows the measured boost conversion from an input voltage of 10mV and 8mV. In this set-up, the V_{CAP} was broken into two rails, one supplying the circuit and another as the output of boost converter. The V_{CAP} supplying circuit was initially pre-charged to higher voltage. Figure 3-23 shows that the output can charge from V_{IN} of as low as 10mV. Figure 3-24 shows the waveform for kick-start. In this set-up, VCAP is initially charged to 590mV and then left to charge from boost converter. The boost converter charges the rail from 590mV to 1V.



Figure 3-24 Measurements showing kick-start from 590mV charging to 1V

Table 3-1 compares this work with previously reported micro-power boost converters for low input voltages. Our work reports harvesting from least input voltage at 10mV, with an efficiency of 22%. Our static power consumption is 300nW and the circuits are designed in 130nm. The start-up voltage is similar to [38][40] at 500mV. The cold start-up mechanism for our system is RF assisted. Other work starts using mechanical kick-start [39] or transformers [41] to start at low voltages such as 35-40mV. This work proposes a peak inductor current control scheme and offset calibration implementation to improve performance, and to achieve an efficiency of 84% at 0.4V input voltage over 4% better than [40] for higher voltage category. In a low voltage category it improves the efficiency by 6% over [38] for input voltage of 20mV.

A thermoelectric boost converter combines an I_{PEAK} control scheme with offset compensation and duty cycled comparators, to enable energy harvesting from TEG inputs as low as 5mV to 10mV, 50% to 75% lower than prior work. Maintaining constant I_{PEAK} allows the converter to sustain high efficiency across a broad V_{IN} range, achieving 52% and 84% efficiency at 20mV and 400mV, respectively, which improves on prior designs. These features allow the converter to extend the operating window for thermal harvesting with low thermal gradients, which is ideal for body-worn sensors.

	This work	[38]	[39]	[40]	[41]
Harvesting Mode	TEG	TEG	TEG	TEG/PV	TEG
Minimum V _{IN}	10mV	20mV	25mv	-	40mV
Maximum V _{IN}	none	250mV	none	none	-
V _{OUT}	1V	1V	1.8V	-	0.3V
Start-up Voltage	0.5V	0.6V	35mV	330mV	40mV w/ x-
			w/ mech.		former
Quiescent Current	300nW	~1µW	-	330nA	-
Constant I _{PEAK} Control	\checkmark	×	×	×	×
Offset Compensation	\checkmark	×	×	×	×
MPP Tracking	\checkmark	×	~	\checkmark	\checkmark
Peak Efficiency @ V _{IN}	84% @ 0.4V	75% @	58%	80% @	61%@
(end to end)	81% @0.3V	0.1V	@0.1V	0.5V	0.3V
	79% @ 0.1V				
Efficiency at low V _{IN}	52% @ 20mV	46% @	-	-	30% @
	22% @10mV	20mV			50mV
Technology	130nm	130nm	350nm	-	130nm

Table 3-1 Comparison summary of the boost converter with previous micro-power converters

3.3 Voltage Reference Circuit for ULP Systems

An ideal voltage reference is independent of variations of power supply or temperature. The term *reference* is used to indicate that the voltage coming out of a voltage reference circuit is more precise than ordinarily found in other sources. Many applications in a system require voltage reference. Figure 3-25 shows the block diagram of a voltage regulator circuit and a successive approximation register (SAR) analog to digital converter (ADC). The voltage reference block forms an essential component in these circuits. In a voltage regulator circuit, output V_{DD} is controlled through the voltage reference. A precise reference voltage gives a precise V_{DD} . In an ADC circuit, the analog voltage from the external world is converted into digital bits. The external signal is compared with the reference. A voltage reference circuit is an essential component in a SoC, and greatly determines the performance of various blocks inside it. In this section, we present a voltage reference circuit suitable for ULP applications.



Figure 3-25 Use of voltage reference circuit in basic building blocks such as voltage regulators and ADC

The voltage reference circuit is needed for our EHM system, for both the energy harvester circuit and the power management circuit, for regulation and control as shown in Figure 3-1. The design of voltage reference is very critical for our EHM system and it needs to meet several design requirements. The voltage reference should have very small variations with process, temperature, and voltage which, in this case, is V_{CAP} of Figure 3-1. These are typical requirements for voltage references in any system. However, for ULP systems running on harvested energy [1], additional requirements have to be met. First, the voltage reference circuit needs to be low power. Second, the area of the circuit should be small to keep the size of the BSN node small. Finally, the voltage reference circuit should become operational at a lower voltage of V_{CAP}. This is very critical for ULP systems, for the following reason: The V_{CAP} value at which voltage reference becomes active is typically the voltage at which the node can turn on. At this voltage, all the V_{DDS} for the system become operational. Therefore, the voltage reference circuit puts a threshold voltage for V_{CAP} at which the BSN node can turn on. If this voltage is low, the BSN will have a longer life-time. The voltage reference circuit used in [1] has a start-up voltage of 0.9-1V, which is typical of circuits used for voltage references in low power areas, and consumes 200nW. In this work, we present a voltage reference circuit which has a start-up voltage of 0.8V and which consumes 80nW power.

3.3.1 Bandgap Voltage Reference

The design of the voltage reference circuit has been a subject of research in the circuit design community since the early days of integrated circuits. This is because of the impact voltage reference has on the overall performance of the system. In the early days of circuit design, temperature compensated zener diodes were used for generating the voltage references. The break down voltage of zeners was close to 6V in those days, they were very noisy, they consumed large

amount of power and they were discrete, off-chip components. The voltage reference using zeners did not meet many requirements of a typical reference. The voltage reference circuit needs to be process and temperature independent. It should support large voltage ranges and be independent of supply variation. The bandgap voltage reference circuit [47] (pp. 153-159) meets the key parameters for all of the above requirements. The concept and first implementation of bandgap reference was proposed by Wildar [48], providing a reference at 5V. A key advantage of this circuit in those days was its ability to scale. Soon after the invention of the first bandgap reference circuit [48], voltage reference ICs producing 10V [49] and 2.5V [50] were demonstrated, using the same concept. The bandgap voltage reference is produced by adding a voltage that is proportional-to-absolute-temperature (PTAT) to another voltage that is complementary-to-absolute-temperature (CTAT), which produces a temperature-compensated voltage.

Over the years of CMOS scaling and evolution in circuits, the bandgap reference has undergone changes in its architecture. However, the concept remains the same since the first bandgap reference was published more than 30 years ago. Various voltage reference circuits have been proposed as alternatives to bandgap reference. Some of the recent publications in this area are listed in the Bibliography [51][52][53]. These circuits use different characteristics of the MOS transistor to obtain a temperature-compensated voltage. However, the bandgap reference circuit outperforms these alternatives in one or more critical parameters for voltage references. As a result, the alternative circuits are used only in limited applications with relaxed specifications. On the other hand, the bandgap voltage reference is widely used. In this section, we present a bandgap voltage reference circuit with a new architecture for our EHM system, shown in Figure 3-1. The proposed circuit is designed to meet the power, area, and voltage requirements of a ULP system.

This section is divided into the following sub-sections. In Section 3.3.2, we present the basic concept of bandgap reference. In Section 3.3.3, we present the current architectures of bandgap reference and their limitations, particularly in the context of ULP systems such as BSNs. In Section 3.3.4, we present the proposed design. In Section 3.3.5, we present the results and limitations of the proposed design.

3.3.2 Background



Figure 3-26 Principle of bandgap voltage reference circuit generating a constant voltage reference with temperature

In this section, we present a qualitative background of bandgap voltage reference circuit. A more theoretical background is available in [47] (pp. 153-159). Figure 3-26 shows the principle behind the bandgap reference circuit. The voltage V_{BE} is complementary-to-absolute-temperature (CTAT). It decreases with increasing temperature with a slope given by -2.2mV/°C. The CTAT voltage is added to the voltage V_t , which is proportional-to-absolute-temperature (PTAT), and its slope is given by 0.085mV/°C. The PTAT voltage is multiplied with a constant K and added to the CTAT

voltage to obtain the bandgap reference, V_{REF} . The value of K is chosen such that the temperature dependence of CTAT and PTAT cancel each other, and V_{REF} becomes a temperature independent voltage reference (10ppm/°C).

Generation of Bandgap Voltage



Figure 3-27 PTAT and CTAT voltage generation circuit

Figure 3-27 shows the CTAT and PTAT voltage generation circuit used for bandgap. The bipolar junction transistors (BJT) are used for the generation of PTAT and CTAT voltages. The CTAT voltage is simply generated by connecting the BJT transistor Q into a diode configuration, as shown in Figure 3-27. The CTAT voltage is given by the base-emitter voltage, V_{BE} of the transistor. As temperature increases, the voltage V_{BE} decreases because of the increase in the number of carriers. Since the number of carrier increases, the conductivity of the transistor increases, which decreases the V_{BE} voltage. The expression of V_{BE} with temperature is given by [47] (pp. 155),

$$V_{BE} = V_{G0} \left(1 - \frac{T}{T_0} \right) + V_{BE0} \left(\frac{T}{T_0} \right) + \frac{\gamma kT}{q} ln \left(\frac{T}{T_0} \right) + \frac{kT}{q} ln \left(\frac{J_c}{J_{c0}} \right)$$
(3.9)

where V_{G0} is the bandgap voltage of silicon, *T* is the temperature, T_0 is the room temperature, V_{BE0} is the V_{BE} voltage at room temperature, J_C and J_{C0} are the current densities, and γ, k, q and α are various physical constants. The value of V_{BE} decreases with temperature. The PTAT voltage is obtained by using two BJT transistors, Q1 and Q2 as shown in Figure 3-27. The transistor Q2 is bigger than Q1 by multiplicity factor M. If the same current is made to flow through Q1 and Q2, the voltage drop across these transistors, V_{BE1} and V_{BE2} , will be different, because of the different current densities (J_C in equation (3.9)). The difference between V_{BE1} and V_{BE2} is called ΔV_{BE} and can be obtained by using equation (3.9),

$$\Delta V_{BE} = V_{BE1} - V_{BE2} = \frac{kT}{q} ln \left(\frac{J_{C1}}{J_{C2}}\right) = V_t ln \left(\frac{J_{C1}}{J_{C2}}\right)$$
(3.10)

The ΔV_{BE} voltage is PTAT and increases with temperature. Thus, equation (3.9) and (3.10) give the expression of CTAT and PTAT voltages from the circuit shown in Figure 3-27. The temperature independent voltage V_{REF} is given by addition of CTAT and PTAT voltages,

$$\boldsymbol{V}_{\boldsymbol{R}\boldsymbol{E}\boldsymbol{F}} = \boldsymbol{V}_{\boldsymbol{B}\boldsymbol{E}\boldsymbol{1}} + \boldsymbol{K}\boldsymbol{\Delta}\boldsymbol{V}_{\boldsymbol{B}\boldsymbol{E}} \tag{3.11}$$

where *K* is the multiplication factor for PTAT to cancel the temperature coefficient in V_{REF} . Solving (3.11) for V_{REF} to obtain zero temperature coefficient, gives *K* as,

$$K = \frac{V_{G0} - V_{BE0} - (\gamma - \alpha)V_t}{V_{t0}}$$
(3.12)

which gives V_{REF} as,

$$V_{REF} = V_{G0} + (\gamma - \alpha)V_{t0} \tag{3.13}$$

The voltage V_{REF} is called the bandgap voltage reference voltage and its value at room temperature is 1.262V. The term bandgap reference is used for V_{REF} because the voltage in equation (3.13) is largely determined by the V_{G0} , which is the silicon bandgap voltage. We have now obtained both CTAT and PTAT voltages and the constant to obtain a voltage reference independent of temperature. Now we will present a typical circuit implementation of bandgap voltage reference.



Figure 3-28 A conventional bandgap reference circuit

Figure 3-28 shows the circuit diagram of a conventional bandgap voltage reference circuit. Transistors Q1 and Q2 are used, along with resistors R_1 and R_2 and amplifier Amp. The functioning of the circuit is explained as follows. The amplifier sets the same voltages at its input net *a* and *b*, because of its high gain (assuming zero offset voltage). Therefore, the same current flows into the transistor Q1 and Q2. Q2 is sized M times bigger than Q1. The voltage V_{BE2} is lower than V_{BE1} . The voltage V_{REF} can be written as,

$$\boldsymbol{V}_{\boldsymbol{R}\boldsymbol{E}\boldsymbol{F}} = \boldsymbol{V}_{\boldsymbol{B}\boldsymbol{E}\boldsymbol{1}} + \boldsymbol{I}\boldsymbol{R}_{\boldsymbol{1}} \tag{3.14}$$

I is also given by V_R/R_2 , because the same current flows in both Q1 and Q2. Since *b* is at the same voltage as *a*, which is equal to V_{BE1}. Therefore, *I* can be written as,

$$I = \frac{V_{BE1} - V_{BE2}}{R_2} = \frac{\Delta V_{BE}}{R_2}$$
(3.15)

so, V_{REF} can be written as,

$$V_{REF} = V_{BE1} + \frac{R_1}{R_2} \Delta V_{BE}$$
(3.16)

The equation (3.16) takes the form of bandgap voltage expression given by equation (3.11), where constant *K* is given by the ratio of R_1 and R_2 . The values of R_1 and R_2 can be chosen such that the temperature dependence of V_{REF} can be eliminated. Figure 3-29 shows the output of bandgap reference voltage at 1.24V. It shows that that output changes by 2.5mV, for a change of temperature from -20-100°C (17ppm/°C). Further, the dependence of V_{REF} on power supply can be investigated by examining the expression of equation (3.16). V_{BE} and ΔV_{BE} are clamp voltages obtained from the diodes of Q1 and Q2. They have very small dependence on power supply. Bandgap reference voltage has small process dependence, which is not significant for various power management applications. However, for high precision applications such as precision ADC, the process variation can be addressed by trimming the resistor R_2 in the design.



Figure 3-29 Variation of bandgap reference with temperature showing 17ppm/°C variation

The bandgap reference circuit shown in Figure 3-28 performs very well and provides almost an ideal voltage reference circuit. However, it has two primary limitations which become critical with CMOS scaling. The first limitation is that the output voltage of bandgap comes out at ~1.25V, which is much higher than the V_{DD} of modern processes. The second major limitation of this circuit is that it needs a supply voltage higher than 1.2V, because the reference voltage is produced in step-down fashion. These were critical limiting factors of bandgap architecture of Figure 3-28 and its usage for modern systems. New architectures for bandgap, which overcome these limitations, have been proposed in the literature [54][55][56]. In the next section, we will present the state-of-the-art low voltage bandgap references.

3.3.3 State-of-the-art Bandgap Circuits for Low Power Applications

CMOS process scaling has brought down the operating voltage of the system to sub-1V range. Further, the sub-threshold operation of systems to save power has brought the VDD voltages below 500mV [1]. A voltage reference circuit that can provide reference to support lower voltages was needed. It also needed to be operational at lower supply voltage, as well as consume lower power, particularly for mobile applications such as mobile phones, laptops, etc. The conventional architecture of bandgap could no longer support these voltages. New architectures were proposed to support low voltage operation. In this section, we will discuss a few recently published low voltage bandgap circuits.

Banba Bandgap Reference

The very first bandgap circuit that produced a reference voltage lower than 1V was published by H. Banba [54], and is popularly known as Banba bandgap. Figure 3-30 shows the architecture of

the Banba bandgap reference. It also works as follows. Net *a* and *b* are set at the same voltage by the amplifier. So,

$$V_a = V_b = V_{BE1} \tag{3.17}$$

Therefore, the current I is given by looking at node b,

$$I = \frac{V_b}{R_1} + \frac{V_b - V_{BE2}}{R_2}$$
(3.18)

Using equation (3.17), we get *I* as,

$$I = \frac{V_{BE1}}{R_1} + \frac{V_{BE1} - V_{BE2}}{R_2} = \frac{V_{BE1}}{R_1} + \frac{\Delta V_{BE}}{R_2}$$
(3.19)

Now, transistors M1, M2, and M3 form mirrors. Therefore, the same current comes out of all the transistors. Therefore, the Voltage V_{REF} is given by,

$$V_{REF} = IR_3 = \frac{R_3}{R_1} \left(V_{BE1} + \frac{R_1}{R_2} \Delta V_{BE} \right)$$
(3.20)

The equation (3.20) gives us an expression of bandgap reference similar to the expression of equation (3.16), with a scaling factor given by R_3/R_1 .



Figure 3-30 Architecture of Banba Bandgap circuit

There are few advantages of Banba bandgap reference, when compared with conventional architecture as shown in Figure 3-28. First, the bandgap reference voltage, which was always fixed at ~1.25V, can now be scaled down by carefully choosing R_1 and R_3 . Second, the minimum operating voltage for this circuit also scales down. The minimum operating voltage V_{min} for this circuit is given by,

$$\boldsymbol{V_{min}} = \boldsymbol{V_{BE1}} + \boldsymbol{V_{DS}} \tag{3.21}$$

where V_{DS} is the drain to source voltage of pMOS, M₁. The V_{BE} of BJT is between 700mV-800mV and V_{DS} of transistor M1 can be as low as 100mV. Therefore, the circuit can operate for voltages below 1V, which makes it suitable for modern scaled designs. The BSN SoC reported in [1] uses Banba architecture and starts operating at 0.9V.

The lower V_{min} for bandgap is even more important for ULP systems such as BSNs. The V_{min} of bandgap determines the voltage at which the BSN node can turn on because the reference is made available at that voltage and is used to turn on the power supplies for the system. The lower turnon voltage will increase the operational life time of the system. Therefore, it is necessary to bring the V_{min} of bandgap voltage down, as low as possible. Recent work [56] brings the V_{min} voltage down to 750mV. It uses switched capacitor circuits for the architecture where, instead of resistors, switching capacitors are used for generating the ratios of V_{BE} and ΔV_{BE} . The V_{min} of this circuit is also limited by equation (3.21) and a lower V_{min} is achieved by proper sizing and design. In the next section, we present a bandgap voltage reference circuit that starts operating at 800mV. Further, the power consumption of the proposed circuit is ~80nW, which is 2X lower than [56] and best in class for non-duty cycled bandgap reference. The proposed circuit is also lower in area. The next section presents the proposed bandgap circuit.

3.3.4 A 80nW, Low Power Bandgap Reference Operational from 0.8V



Figure 3-31 Circuit diagram of the proposed bandgap reference

Figure 3-31 shows the circuit diagram of the proposed bandgap reference. It uses switched capacitor techniques to generate the bandgap voltage. The circuit consists of a switched capacitor inverter, BJT transistors Q1 and Q2, current sources, and a sampling capacitor for holding bandgap voltage. The current sources are used to generate voltages V_{BE1} and V_{BE2} , as shown in Figure 3-31. The switched capacitor inverter circuit is used for generating ratios of V_{BE1} and V_{BE2} to obtain the bandgap voltage. The switched capacitor inverter circuits are typically used for generating voltages that are negative of V_{in} [58]. In Figure 3-31, the switched capacitor inverter is connected to Q1 and Q2, which provide the output voltages at V_{BE1} and V_{BE2} . The use of the switched capacitor inverter circuit removes the need for large resistance in Banba architecture and can provide lower area solution. Further, a clock is used for the inverter circuit that can be made to operate at lower frequency to reduce the power. The use of the switched capacitor inverter circuit, and lower clock frequency, provides lower area and lower power consumption compared to the state-of-the-art bandgaps.
The design description of the proposed circuit is divided into descriptions of each subblock. First, we explain the functioning of the switched capacitor inverter circuit, then the use of the principles of voltage inversion for generating bandgap reference, followed by the complete circuit implementation. Finally, we will present the results.

Switched Capacitor Voltage Inverter Circuit



Figure 3-32 Switched capacitor voltage inverter circuit and its operation

Figure 3-32 shows the switched capacitor voltage inverter circuit used in our design. Its functioning is explained as follows. The switched capacitor inverter uses non-overlapping clock phases, ϕ_1 and ϕ_2 . In phase ϕ_1 , node 1 is connected to V and node 2 is connected to ground, charging the top plate of capacitor C_f to V and the bottom plate to ground, as shown in Figure 3-32. In phase ϕ_2 , node 1 is connected to ground and node 2 to the output capacitor C_L. Since the top plate of the capacitor C_f was charged to V in phase ϕ_1 , charging the top plate to ground pushes the voltage on the bottom plate to -V, as voltage across capacitors cannot change since the bottom plate is not connected to a voltage source. The capacitor C_L will eventually charge to -V after a given number

of switching cycles. Therefore, the switched capacitor inverter circuit generates a voltage which is negative of the input voltage V.

Proposed Bandgap Circuit : Concept

The equation for bandgap reference voltage is given by equation (3.11),

$$V_{REF} = V_{BE1} + K \Delta V_{BE}$$

which can be rewritten as,

$$V_{REF} = (K+1)V_{BE1} - KV_{BE2}$$
$$V_{REF} = (K+1)\left(V_{BE1} - \frac{K}{K+1}V_{BE2}\right)$$

Therefore, we propose to obtain the reference voltage as,

$$V_{REF} = a(V_{BE1} - bV_{BE2})$$
(3.22)

The negative coefficient for V_{BE2} is obtained through a voltage inverter circuit. In the next section, we will show the working of the proposed bandgap circuit.

Proposed Bandgap Circuit

Figure 3-33 shows the circuit diagram of the proposed bandgap voltage reference circuit. It uses switched capacitor techniques to utilize a switched capacitor voltage inverter circuit for the generation of V_{REF} voltage. The two phases of the clock, ϕ_1 and ϕ_2 , are non-overlapping and facilitate the switching capacitor functionality. In phase ϕ_1 , the proposed circuit reduces into the circuit shown into Figure 3-33 (b). The capacitors, C₁ and C₂, are connected across the BJTs, Q1 and Q2, and are charged to voltages, V_{BE1} and V_{BE2} . Therefore, the charge stored on capacitor C₁ and C₂ is given by,

$$Q_1 = C_1 V_{BE1}$$
, $Q_2 = C_2 V_{BE2}$ (3.23)

The switch connections are changed in phase ϕ_2 and the circuit reduces into the circuit shown by Figure 3-33 (c). In phase ϕ_2 , the top plate of C₁ gets connected to the bottom plate of C₂ and the top plate of C₂ is connected to ground. Since top plate of C₂ is connected to ground, the output voltage on C_2 will be $-V_{BE2}$. Voltage on C_1 is V_{BE1} . Therefore the connection with C_1 with C_2 results in charge sharing given by,

$$Q = Q_1 - Q_2 = C_1 V_{BE1} - C_2 V_{BE2}$$
(3.24)

Now, the voltage on the capacitor C_L is given by, (The value of C_L doesn't play a role in charge sharing because it is the load capacitor used for final settling voltage.)

 $\boldsymbol{Q} = (\boldsymbol{C}_1 + \boldsymbol{C}_2) * \boldsymbol{V}_{REF}$



a) Circuit diagram of the proposed bandgap reference circuit



b) Circuit operation in phase φ_1





Therefore, the output voltage can be written as,

$$(C_{1} + C_{2}) * V_{REF} = C_{1}V_{BE1} - C_{2}V_{BE2}$$

$$V_{REF} = \frac{C_{1}}{C_{1} + C_{2}} \left(V_{BE1} - \frac{C_{2}}{C_{1}}V_{BE2} \right)$$
(3.25)

Equation (3.25) reduces in the form of bandgap reference equation given by equation (3.11), where the value of constant *b* can be obtained by selecting the value of capacitance C_1 and C_2 . The proposed circuit has noise injection through the charge coupling of switches. There will be an error in the output voltage given by (3.25) because of the charge sharing. Since the charge coupling will change with V_{IN}, therefore output voltage will also change with V_{IN}.

By applying the above circuit technique, we have successfully obtained the bandgap reference voltage. The proposed scheme has the following advantages over classical Banba bandgap reference. First, large resistances are no longer needed for generating bandgap voltages. Therefore, the area of the proposed circuit will be much smaller than the Banba architecture implementation for low power application. Second, the power consumption of the proposed scheme will be lower, because the elimination of resistors eliminates the static power consumption in the resistors, and the switched capacitor converter power can be lowered by lowering the clock frequency. The noise performance of the circuit is traded with capacitor C_L and settling time of the converter. Therefore, the proposed bandgap solution provides a low power, lower area solution for the implementation of voltage reference for our EHM system.

3.3.5 Prototype Implementation and Results

The proposed bandgap circuit was implemented in 130nm CMOS process. It has an area of 0.0264mm². The capacitors are implemented using MIM capacitors to reduce the effect of parasitic loading. The total area of the proposed circuit is much smaller than a conventional low power

bandgap circuit because it does not use big resistors. It consumes 80nW power at 0.8V Vin (V_{CAP}), which is 2X lower than the best reported power for non-duty-cycled bandgap reference.



Figure 3-34 Transient behavior of bandgap reference

The circuit is a switching capacitor circuit; therefore, it has settling time. Figure 3-34 shows the transient behavior of the bandgap circuit. It takes 15ms to achieve a settling time at 0.8V V_{in} . The output voltage of bandgap reference is 123mV. The settling time is directly dependent on the clock frequency and power supply. In some applications, the settling time of the bandgap can be large. Therefore, we also propose a fast start-up mode for the circuit. During start-up, the clock frequency can be made several times faster, which reduces the settling time. This is done during power-on reset mode, where the current source of the clock source is increased several times, increasing the frequency. We achieve a settling time of 20µs during start-up if used in fast start-up mode.

The circuit was verified for a temperature range of -20°C to 100°C. While this range is quite large for the intended ULP applications, the performance of the circuit in this range is crucial for it to compare with the state-of-the-art circuits. Figure 3-35 shows the variation of the bandgap output for a temperature change of -20°C to 100°C. The proposed bandgap circuit is designed to provide an output voltage of 122mV, and the output voltage varies by 1.5mV over a temperature variation of 120°C, achieving a performance of 100ppm/°C. The performance of the bandgap circuit with temperature is in line with the reported work. A better performance is usually achieved at higher output voltage.



Figure 3-35 Bandgap output variation with temperature

Figure 3-36 and Figure 3-37 show the output of a bandgap circuit with respect to process and mismatch variation, and with input voltage variation. Figure 3-36 shows the untrimmed output of the bandgap and it achieves a 3σ variation of the <2.5%. The variation of the output voltage can be reduced by trimming the bandgap output, using the capacitors used in switched capacitor

circuits to generate the constants for bandgap. Figure 3-37 shows the variation with V_{in} . The output varies by ~1% when an external constant clock is used. The variation with respect to power supply can be reduced by using a dedicated clock source.



Figure 3-36 Variation with process: Monte-Carlo Simulation



Figure 3-37 Variation with Vin Voltage

	[57]	[53]	[54]	[55]	[56]	This work
Power Consumption	300nW	1.85µW	13.6µW	20µW	170nW ¹	80nW
Technology	300nm	400nm	600nm	600nm	130nm	130nm
				BiCMOS		
Area (mm ²)	0.055	0.1	0.055	0.4	0.07	0.0264
% variation with	6	5.8	12	1.5	3	2.5
process						
Temp. variation	15	119	37	11	40	100
(ppm/°C)						
PSRR (%)	0.002%	0.138%	0.038%	0.0048%	0.005%	2%
Minimum Supply	1.4V	0.84V	1.4V	0.9V	0.75V	0.8V

Table 3-2 Comparison summary with previous low power bandgaps

¹uses duty-cycling to reduce power

Table 3-2 compares this work with previously reported state-of-the-art low power bandgap circuits. Our work reports operation from minimum input voltage of 0.8V improving, which is similar to the reported least operating voltage circuit in [56]. Our power consumption is 80nW, which is over 2X lower than [56], achieved without duty cycling the reference. The work in [56] achieves a power of 170nW by sampling the reference voltage on a capacitor by periodically turning it on and off. Even lower power can be achieved in our work if duty-cycling is employed. The power supply variation is higher in our circuit because our architecture doesn't use current sources which are used in all of the previous architectures. The lower area of the circuit (0.0264mm²) is achieved because big resistors are not used [55]. A low power, lower input voltage bandgap reference, which improves the power consumption by 2X and minimum operating voltage 0.8V, is presented.

3.4 Power Management Solution for ULP SoCs

In the previous sections, we presented the design of a boost converter energy harvester that can harvest energy from a thermo-electric generator, and a design of a low power bandgap voltage reference. The energy harvester circuit is used to harvest energy from low power ambient sources. It can harvest energy from a source with an ambient source voltage as low as 10mV, which is stored on an energy accumulator such as a battery or a capacitor. On the other hand, the bandgap circuit can be powered-up at a low voltage level of storage capacitor (V_{CAP}) which helps in reducing the start-up voltage of the system. The bandgap voltage itself is used for providing stable reference voltage. The output of these two blocks is used in our EHM system to provide multiple regulated output voltages for the voltage regulator circuits that are used to power different blocks of the system. This section talks about the design of the power management system that produces multiple output voltages to be used by the SoC.

3.4.1 Voltage Regulation for Integrated Circuits

One or more DC power supplies are needed for the functioning of different blocks inside a SoC. The power supply(s) is one of the most important design and system knobs for the SoC, which can determine several key performance metrics. Therefore, power supplies in a system are very well-regulated. There are several reasons to control the power supply. First, the operating voltage for a circuit can determine its operating frequency. Therefore, power supply is critical for the performance of the IC. Second, power supply also determines the power consumption of the circuit. Often, the performance and the power consumption of a circuit have contradictory attributes. A higher performing circuit would need a higher power supply and hence higher power. Therefore,

it is desirable to have a power supply that gives good performance, as well as restricts the power consumption. Figure 3-38 shows the energy-delay characteristics of a digital circuit obtained by changing the power supply, V_{DD} . It shows that energy consumption increases for a lower delay, which is obtained by operating the circuit at higher V_{DD} . Figure 3-38 shows that V_{DD} needs to be controlled to control power consumption or the performance of the circuit. The third main reason to control the power supply is to be able to operate the circuits within a safe operating voltage for better reliability. Fourth, a transient fluctuation in power supply can lead to a system shut down because of an internal set-up time or hold-time violation inside the chip. The other reasons to control power supply are meeting compliance for IO standards for chip-chip communications, maintaining good signal to noise ratio for analog circuits (sometimes for digital circuits as well), to maintain the functionality of memories, such as retention, etc. However, the power and performance needs of a system are the primary motivating reasons to set the level of power supply in most of the systems.



Figure 3-38 Energy-Delay characteristics of a digital circuit

Voltage Regulator Circuit: Linear Regulator



Figure 3-39 A linear regulator circuit to supply VDD

Figure 3-38 shows that V_{DD} needs to be regulated to control the power consumption or performance of a circuit. A circuit that controls V_{DD} is shown in Figure 3-39, which is a linear voltage regulator circuit. A linear regulator circuit, as shown in Figure 3-39, is often sold as 3-pin IC by various semiconductor companies [60][61][62]. The three pins would be V_{in} , Gnd and V_{DD} . The circuit operates in the following way. A reference voltage, usually a bandgap reference, as described in Section 3.3, is used with an amplifier in a negative feedback loop. If V_{DD} goes a little bit below the reference, the feedback from the amplifier increases the gate voltage of the nMOS and supplies more current to raise the V_{DD} . Similarly, if the V_{DD} goes high, the current supply from V_{in} is reduced. The negative feedback keeps the V_{DD} voltage the same as the reference. A linear regulator circuit is small in size and has a very low noise at its output. In modern SoCs, the linear regulator circuit is often integrated. Since modern systems need more than one power supply, multiple linear regulators are integrated on chips to generate these supplies [1]. Further, lower quiescent current and better transient performance linear regulators have been reported for low power systems [63][64]. However, there are two main limitations of a linear regulator circuit when used for low power applications. The first limitation is the low efficiency for voltage conversion. The efficiency of a linear regulator is given by,

$$\eta < \frac{V_{DD}}{V_{in}} \tag{3.26}$$

If the difference between V_{in} and V_{DD} is large, then efficiency can be very small. For a V_{DD} of 0.5V, with a V_{in} of 1.5V, the efficiency of the converter will be less than 33%. The second main limitation of the circuit is higher operating voltage. V_{in} always needs to be higher than V_{DD} . For an energy harvested system, it means that the voltage on the storage capacitor, V_{CAP} , needs to be higher than V_{DD} , which means that the energy stored on the capacitor cannot be used unless it reaches a threshold voltage that is higher than V_{DD} . For the BSN reported in [1], this voltage was 1.35V, because it needed to generate a 1.2V power supply.

To enable low voltage start-up for the system, we proposed a bandgap reference circuit in Section 3.3 that can provide stable reference at 0.8V of V_{in} . Further, the power supply(s) should become operational at low input voltages. For example, if the power supply is made available at 0.8V of V_{CAP} or V_{in} , then the system can start-up at 0.8V instead of 1.35V, which will increase the life time of the system. Therefore, there is a need to overcome the limitations of linear regulators to improve the life-time of ULP systems. Switching regulators provide better efficiency and can operate at lower V_{in} are proposed and explained in the next section.

Voltage Regulator Circuit: Switching Regulator

Switching regulators are broadly divided into two categories, inductor-based switching regulators and switched-capacitor regulators. The switched-capacitor regulators use the switching of capacitors to generate various ratios or multiples of the output voltages. The voltage inverter circuit, shown in Figure 3-32, is an example of a switched capacitor regulator where an output is inverse of the input voltage. Recent works show that it is possible to achieve good efficiency for switched capacitor converters for given sets of output voltages [65][66][67].

There are certain advantages and disadvantages of using switched-capacitor techniques rather than inductor-based switching regulators. An obvious key advantage is the elimination of the inductor and the related magnetic design issues. Application circuits are simple, and are usually integrated for low power systems. As there is no need for an inductor, the size of the PCB implementing the system will be small. Switched-capacitor inverters are low cost and compact, and are capable of achieving efficiencies equal to or greater than 90%. The current output is limited by the size of the capacitors and the current carrying capacity of the switches. However, switched-capacitor voltage converters do not maintain high efficiency for a wide range of ratios of input to output voltages. Since the current input to output ratio is scaled according to the basic voltage conversion (i.e., doubled for a doubler, inverted for an inverter), regardless of whether or not regulation is used to reduce the doubled or inverted voltage, any output voltage magnitude less than 2V_{in} for a doubler or less than |V_{in}| for an inverter will result in additional power dissipation within the converter, and efficiency will be degraded proportionally.

The inductor-based switching regulators give good efficiency for both varying input and output voltages. They can also start-up at lower input voltages V_{in} and can boost up from very low voltages, as was shown in Section 3.2. The rest of this section will talk about inductor-based switching regulator circuits. We will first present the basic principle behind an inductor-based switching regulator, then explain a multiple output switching regulator to provide multiple voltages

for an SoC. Finally, we will present an EHM (Energy harvesting and power management system), which harvests from solar cells and provides multiple regulated output voltages.



Figure 3-40 DC-DC converter architectures

Figure 3-40 shows the architectures of various DC-DC converters using inductors. The basic principle behind the operation of DC-DC converter involves charging and discharging of inductor current on a capacitor. In the first phase of a switching cycle, the inductor is charged and its current is raised, while in the second switching cycle, the inductor current discharges on the capacitor. The energy transfer takes place in electro-magnetic fashion where first, electrical energy is converted into magnetic energy in the inductor and then transferred back to electrical on the capacitor. The energy loss in this way is smaller and the efficiency of these converters is usually very high (in

~90% range). Figure 3-40 shows the architecture of three different kinds of DC-DC converters. Figure 3-40 (a) shows the architecture of a buck-converter, which is used to generate an output voltage V_{DD} lower than the input voltage V_{in} . In the first switching cycle, the inductor is connected between V_{in} and V_{DD} . Since V_{in} is higher than V_{DD} , inductor current starts increasing. After a set time, the high side switch (HS) is disabled and the low side (LS) switch is enabled, which configures the inductor between ground and V_{DD} . The inductor current discharges, transferring the stored energy onto the output capacitor. Figure 3-40 (b) is a boost converter and is used when V_{DD} > V_{in} . We covered the design of the boost converter in Section 3.2. Figure 3-40 (c) shows the architecture of a buck-boost converter when V_{in} can be higher or lower than the output voltages. Additional switches are used in buck-boost converters; therefore, their efficiency is slightly lower than the buck or boost converters.

Better efficiency and the ability to support (almost) any voltage conversion ratio are the main reasons for the wide scale use of the DC-DC converter. The low loss in a DC-DC converter is typically achieved by using low resistance switches (often big in size), lower parasitic loss in inductor and capacitor, and choice of switching control scheme for a particular application. However, implementation of a DC-DC converter requires external off-chip inductors and capacitors, which can increase the cost and area of the system. Also, multiple V_{DDS} are needed in a system, but multiple instances of the DC-DC converter cannot be implemented because of the cost and area. Therefore, linear regulators with lower efficiencies are often used to implement multiple V_{DDS} in a ULP system such as BSN [1]. There is a need to provide a low cost and efficient solution to implement multiple regulated output voltages for ULP systems. In the next section, we present low cost multiple output DC-DC converter design suitable for ULP systems.

3.4.2 Single Inductor Multiple Output DC-DC Converter for ULP Systems

Figure 3-2 shows the power management block of a BSN chip [1]. The BSN chip needs multiple rails for optimal power performance. The existing power management solution for the chip includes multiple linear regulators and a switched capacitor regulator to supply different rails. A linear regulator, such as an LDO, suffers from poor efficiency. The theoretical maximum efficiency of an LDO cannot be more than V_{DD}/V_{in} . Therefore, more than 63%, 26%, and 12% of the energy on the V_{DD} rails, 0.5V, 1.0V, and 1.2V gets wasted in obtaining the regulated V_{DD} voltage. There is a need to provide energy efficient solutions for the power management which can increase the lifespan of the BSN chip. This work proposes a highly efficient, single inductor, multiple output power management solution for BSNs.



Figure 3-41 Single inductor multiple output buck-boost converter for ULP systems

The power management solution for ULP systems needs to meet the following three main requirements. First, the solution should be highly efficient; second, it should be low cost; and, third,

it should start operating at lower input voltage. Figure 3-41 shows the architecture of the proposed single inductor, multiple output DC-DC buck-boost converter for ULP systems. The use of a DC-DC converter ensures good efficiency for the converter. The use of buck-boost architecture enables operation at lower V_{in} voltage. Finally, a single inductor, multiple output design provides a low cost solution for the system. Single inductor, multiple output (SIMO) DC-DC converters have been proposed as a low cost solution for mobile applications [68][69]. The inductor is used by different V_{DD} rails in time-division, multiplexed (TDM) fashion for regulation. In this work, we propose the use of SIMO for ULP systems. Figure 3-41 shows the architecture of the proposed SIMO converter. When LS control is enabled, switches S_H and S_B are enabled and other switches are open. The inductor is connected between ground and Vin, which starts raising current in the inductor. After a set peak, current in the inductor LS control is disabled and switch S_L is closed, along with either S_1, S_2 , or S_3 . The stored energy in the inductor gets transferred in each of the three rails shown. The input voltage V_{in} can be lower or higher than all the V_{DD}s. This particular architecture enables low voltage, low cost, and efficient implementation of power management for a ULP system. The proposed architecture was implemented in 130nm CMOS process. We achieve a low voltage operation, from 700mV compared to 1.35V in [1].

In this section, we talked briefly about the SIMO buck-boost architecture to enable low voltage, efficient and low cost solution for ULP systems. The complete EHM solution will include an energy harvester to harvest from ambient sources, and a power management solution as shown in Figure 3-1. The EHM system is implemented using a boost converter and we explained the architecture in Section 3.2. Further, to enable low voltage operation for a ULP system, bandgap voltage reference and buck-boost architecture were covered in Section 3.3 and in the previous

section. The implementation of a complete EHM system is shown in Figure 3-42. A boost converter harvests from ambient sources, while the SIMO converter provides a solution for power management. This implementation required two inductors. A further improvement over this architecture is proposed, where a boost converter and power management are integrated. We propose a single inductor EHM system, where the boost converter and power management were merged into one solution. The next section shows the implementation of the proposed improvement using a photo-voltaic or solar cell (PV).



Figure 3-42 Architecture of the proposed EHM system

3.4.3 Single Inductor EHM for ULP Systems²

Energy harvesting from ambient sources, such as solar or thermal, is used to enable battery free operation of wireless sensors, body sensor nodes (BSN) [1] or Internet of Things (IoT). The energy harvested from a solar cell depends on illumination level and can vary greatly. The energy harvesting system needs to harvest energy efficiently, in varying conditions, to prolong the operational lifespan of the system. Further, BSNs, like most SoCs, need multiple supply rails to power core circuits, memories, and RF and IO cells. The existing solutions for BSNs use low drop out (LDO) regulators which have poor efficiency [1]. BSNs need an energy harvesting and power

² A lot of content of this section comes from [AS3] and [AS10]

management (EHM) system that can efficiently harvest energy from ambient sources, as well as provide the multiple regulated output voltages. The system also must be low cost for its widespread deployment. Single inductor multiple output (SIMO) EHM systems have been recently proposed to provide low cost and efficient solutions for BSNs [70]. In this section, we present a SIMO EHM system that can harvest from a solar cell (0.4V to 3.3V i/p voltage) to a storage capacitor or a battery interface. It also provides multiple output voltages (1.2V, 1.5V and 3.3V) at the same time, using the same inductor. It supports 0-5V output voltage for capacitive storage and 1.7-5V for a battery interface. It has a maximum power point tracking (MPPT) circuit, and an over-voltage and under-voltage protection scheme for battery management. A cold start-up circuit operating at minimum input voltage of 0.38V is also presented.

Architecture



Figure 3-43 Proposed energy and power management solution using single inductor

Since the output power from a solar cell can be much smaller than the peak power needed by the system for applications such as RF communication, it is not possible to supply the V_{DD} rails directly from it. Therefore, the energy from the solar cell is first stored on an energy accumulator (a big capacitor or battery) and used to supply other V_{DD} rails [1]. Figure 3-43 shows the concept

of the switching scheme used in our EHM system. The energy from the solar cell (PV) is harvested and stored on the accumulator, and from there it is used to supply different voltages needed for a system. The arrangement of the switches in this way eliminates the need for additional switches to reconfigure the accumulator to supply in cases of high output load. This reduces the number of switches needed for EHM compared to [70]. The storage or battery is kept on the +ve side of the inductor and the PV module and regulated output voltage are kept on the negative side. The inductor current flows from the –ve of the inductor to the +ve terminal for storage. For supply regulation, the current flows from battery or storage into the loads (i.e from + to -) as shown in Figure 3-44.



Figure 3-44 Inductor current waveform for the EHM switching scheme showing the change in direction of Inductor Current for boost operation

Figure 3-45 shows the complete architecture of the proposed EHM system. A SIMO control scheme is used with all the converters in the design, and operated in discontinuous conduction mode (DCM), using a pulse frequency modulation scheme (PFM), to achieve higher efficiency at light loads. Each converter generates two output signals, Busy and Ready. The Busy signal indicates that a particular converter is getting the service of the inductor, while the Ready signal

indicates that a particular converter needs to be serviced by the inductor. The switching control is performed by the Digital Controller. Using the Busy and the Ready signal from each converter, it generates an EN signal based on priority to enable a particular converter which, in turn, controls common switches MP and MN through signals HS and LS, as shown in Figure 3-45. The highest priority is assigned to the heavily loaded rail. The least priority is assigned to the energy harvester block.



Figure 3-45 Complete architecture of the EHM System

Architecture of the buck-converter

In order to achieve maximum overall system efficiency across a wide range of output conditions, especially for light loads, each converter inside the EHM must also provide maximum efficiency. Figure 3-46 shows the control architecture of the buck converter used in EHM. It implements a PFM control scheme with peak inductor current (I_{PEAK}) control for high side switching and inductor zero current detection (ZD) for low side switching. A 70nA comparator is used for

regulation. If output rail is below Vref, the comparator output goes low which generates the Ready signal as shown in Figure 3-46. The digital controller sends an EN signal based on the priority of the converter to start the switching. The switching continues until the rail is charged above the reference voltage and the Busy signal goes to high. Once the rail is charged, the Busy and Ready signals go low and the inductor is made available for other rails. 70nA quiescent current, I_{PEAK} control, and ZD scheme enable high efficiency for the buck converter from 1µA to 30mA load current. While the ZD and comparator designs are conventional, the I_{PEAK} control scheme is novel and has not been previously implemented for micro-power converters. The conventional scheme to implement I_{PEAK} control uses a method where R_{DS} of a MOS transistor is matched with the switch R_{DS} , which is several orders of magnitude smaller than the representative MOS device. As a result, the method is sensitive to mismatch (+/-20%) and has large power overhead [46]. In this Section, we present an I_{PEAK} control scheme which is low power and less sensitive to mismatch, and similar to the I_{PEAK} scheme proposed for boost converter energy harvester in Section 3.2.



Figure 3-46 Architecture of the buck converter used in EHM

I_{PEAK} control for buck converter



Figure 3-47 Peak inductor current control scheme in buck converter to achieve high efficiency

The need for controlling the peak inductor current for maximizing efficiency was shown in Figure 3-11 in Section 3.3.2 and Appendix C. If I_{PEAK} is high, the conduction loss in the converter dominates, while switching loss dominates when I_{PEAK} is low. Therefore, controlling the peak inductor current is necessary to optimize the efficiency of the converter. Figure 3-47 shows the circuit of the I_{PEAK} control scheme, which is explained as follows. A large resistor R sets the node A close to its threshold voltage of M1. Therefore, voltage at A is given by Vo-V_T which goes to the gate of M2. Using long channel saturation current approximation of MOS transistor, the current in M2 is given by,

$$I = K\mu C (Vo - V_T - V_I + V_T)^2 = K\mu C (Vo - V_I)^2$$
(3.27)

The capacitor is pre-charged to $V_0/2$. Figure 3-47 shows the HS turn on time T_H is given when capacitor C_X charges from Vo/2 to Vi/2. So T_H is given by,

$$T_H = C_X(Vi - Vo)/2I = C_X/2K\mu C(Vo - Vi)$$
(3.28)

Now, the peak inductor current is approximately given by,

$$I_P = (V_I - V_O)^* T_H / L \tag{3.29}$$

Using equation (3.27),

$$I_P = C_X / 2K \mu CL \tag{3.30}$$

The expression in equation (3.30) is independent of input voltage Vi and output voltage Vo, which enables wide input and output voltage range support for the system. It is also independent of the threshold voltage of the transistor. As a result, the peak inductor current is made to be constant. It is fixed by controlling the size of on-chip capacitor C_X at design time.



Figure 3-48 Variation of IPEAK Vi showing <1% variation with Input Voltage Vi

The proposed scheme performs well when compared with conventional MOS R_{DS} methods. In this scheme, the timing is generated by the integration of charge on a capacitor from Vi/2 to Vo/2, which makes it less sensitive to mismatch. For example, if Vi is 5V and Vo is 1.5V, then the integration happens over 1.75V. Even a 100mV offset because of all the circuits involved in the

design results only in a mismatch of 6%. On the other hand, in the MOS R_{DS} method, the voltage drop across the switch is compared to a MOS transistor in linear region. Usually, the drop across the switches is very small, in the range of 50mV, which makes it very sensitive to mismatch [46]. Figure 3-48 shows the performance of the I_{PEAK} control scheme with Vi. It achieves less than 1% variation in I_{PEAK} with 2.3V variation in Vi. Figure 3-49 shows the monte-carlo simulation result for I_{PEAK} control and it shows +/-6% variation because of the process and mismatch which can be further controlled by trimming C_X in Figure 3-47.



Figure 3-49 Variation of peak inductor current with process and mismatch variation

Architecture of the boost-converter

The circuit is designed using a 130nm 3.3V process with drain extended MOS transistor to support up to 5V operation to handle a wider range of different battery types, such as Lithium based battery chemistries, as well as store higher amounts of energy on the storage device. While the peak inductor current of the boost converter can be configured using a technique similar to the one used in Figure 3-47, the conventional ZD cannot be used because of the reliability issues involved in designing with 3.3V transistors. Figure 3-50 shows the control circuits to generate LS and HS timing signals for the boost converter. The LS signal is generated when C_Y charges from Vi/2 to Vo/2 and HS signal is generated when C_Y charges from ground to Vi/2, with their values given in Figure 3-50. The ratio of LS and HS times gives the boost converter ratio assuming the voltage drop across the switches much smaller than Vi or Vo. Therefore, the zero crossing of the inductor current in the boost converter can be accurately predicted by the given circuits. Figure 3-50 also shows the inductor current waveform for different input voltages, with different peak current showing accurate zero crossing, which helps in achieving wide input and output voltage range for the boost converter.



Figure 3-50 High side control scheme for the boost converter to accurately predict the zero crossing of the inductor current enabling reliable operation for 5V battery/storage using 3.3V devices

The boost converter also uses a PFM architecture with a constant voltage maximum power point tracking, which is well suited for a slow changing condition such as low illumination. The maximum power point tracking scheme is implemented to operate the PV cell at its maximum power point. The maximum power point of a PV cell occurs at ~76% of the open circuit voltage

[73]. The implementation of the MPPT control is similar to the MMPT control explained in Section 3.2 for TEG. Figure 3-51 shows the architecture of the boost converter circuit. The node V_{MPP} is set at 76% of open circuit voltage of the PV cell. The boost converter also has a buck-boost mode to harvest when output voltage is lower than the input voltage. A programmable over-voltage protection comparator is used to control the output voltage.



Figure 3-51 Boost converter architecture



Figure 3-52 Cold start circuit using an internal charge-pump to start operation from 380mV

In this implementation, we also present a cold-start up circuit. Figure 3-52 shows the cold-start circuit. A local rail is produced using a charge pump circuit, which supports a ring oscillator to

control the switching during start-up. The process used doesn't support 5V gate to source operation. Therefore, the accumulator voltage cannot be used for internal circuits inside the converter, and instead the 3.3V rail is used. Since the circuit is powered by 3.3V rail in normal operation, it needs to be charged up during start-up as well. 3.3V rail and Storage node, V_{BAT} , are connected together during the cold start mode, as shown in Figure 3-52. Once V_{BAT} crosses 1.7V, the 3.3V rail can reliably handle switching. Therefore, the connection between 3.3V rail and V_{BAT} is removed and normal switching operation commences. The 3.3V buck converter and boost converter have buckboost mode to take care of varying PV and V_{BAT} levels.





Figure 3-53 Voltage regulation output of 1.2V rail showing <15mV ripple

The proposed EHM system is implemented in 130nm CMOS process with drain extended devices to support 5V operation. Figure 3-53 shows the simulation result of the regulated output voltage at 1.2V, showing less than 15mV ripple at 10mA of load current. Figure 11 shows the

measurement result of the multiple regulated EHM output voltage rails coming out from the EHM system. It supports a peak load of 30mA on one VDD, with other rails lightly loaded, or a peak of 10mA on each VDD (max 30mA total).



Figure 3-54 Measurement result of multiple output V_{DD}s coming at 1.2V, 3.3V and 1.5V



Figure 3-55 Efficiency of buck Converters

Figure 3-55 shows the efficiency measurements for the buck converters used in the SIMO converter. The peak efficiency is 92% on 3.3V rail at high load and 83% at 1μ A load. The 1.2V

and 1.5V achieve peak efficiency of 82% and 81%, respectively, at high load condition, while 68% and 70% at 1 μ A load. The architecture supports a maximum of 30mA of load current which can support ULP systems such as a BSN [1]. Figure 3-56 shows the efficiency result of the boost converter. The boost converter achieves a peak efficiency of 90% at higher input voltages and an efficiency of 81% at 0.4V, Vin supplying 10 μ A of current. The circuit starts up at 380mV. Figure 3-57 shows the die-photo.



Figure 3-56 Efficiency of the boost converter



Figure 3-57 Die photo of EHM system

	[70]	[71]	[40]	This Work
Process	0.18µm	0.25µm	0.35µm	0.13µm
# of o/p	3	1	1	4
o/p Voltages	1V, 1.8V &3V	5V	0-3.3V	5V, 1.2V,1.5V & 3.3V
Efficiency	83% max @100μW, 67% @1μW, 70% @10mW	87%	80% @0.5V 95% @ 3V for boost	81% @ 0.4V, 92% @ 3.3V for boost* 92% @ high load, 83% @1µA for 3.3V buck * *(in SIMO)
Idle Power	0.4µW	2.4µW	1µW	1.2µW
Start-up	-	1V	0.33V	0.38V
Max. o/p V	3V	3V	3.3V	5V
Area (mm ²)	4.625	11.56	-	2.25
Max. Load	10mW	10mW	25mW	100mW
SIMO Reg.	~	×	×	✓
Bat. Mgmt.	×	×	\checkmark	\checkmark
Cold start	×	×	✓	✓

Table 3-3 Comparison with existing state-of-the-art harvesters

Table 3-3 shows the comparison of the proposed EHM system with state-of-the-art energy harvesters. Among the reported work, this work is the first complete EHM system with energy harvester, multiple output regulated voltages for the system, a cold start circuit, a wide input voltage range 0.4-3.3V support, and a wide-output voltage to enable 5V operation. The solution in [70] has energy harvester and regulated output voltage, but it doesn't support wide input voltage range, battery management or cold-start. The solution in [40] provides battery management, but doesn't support voltage regulators. The static power consumption of the system is 1.2μ W and is higher than [70] because of the implementation of all the control schemes needed for complete system design. The proposed I_{PEAK} control scheme enables efficient operation over a wide input-

output voltage range for the system. These features provide efficient energy harvesting and power management system for BSNs, WSNs or IoT.

3.5 Conclusions

In this chapter, we presented architectures to implement energy harvesting and power management solution for ULP systems. A thermoelectric boost converter combines an IPEAK control scheme with offset compensation and duty-cycled comparators to enable energy harvesting from TEG inputs as low as 5mV to 10mV, 50% to 75% lower than prior work. Maintaining constant IPEAK allows the converter to sustain high efficiency across a broad V_{IN} range, achieving 52% and 84% efficiency at 20mV and 400mV, respectively, which improves on prior designs. These features allow the converter to extend the operating window for thermal harvesting with low thermal gradients, which is ideal for body-worn sensors. We also presented a bandgap reference voltage circuit to enable low operating voltage for ULP systems. Our work reports operating from a minimum input voltage of 0.8V and achieving a lower power consumption of 80nW, which is over 2X lower than previous work achieved without duty cycling the reference. The proposed circuit also has the lowest area for bandgaps because big resistors are not used for lowering the power. Further, a power management solution that combines the boost converters and voltage regulators is presented to provide a completely integrated EHM solution. Among the reported work, this work is the first complete EHM system with energy harvester, multiple output regulated voltages for the system, a cold start circuit, a wide input voltage range 0.38-3.3V support, and a wide-output voltage to enable 5V operation. These features provide efficient energy harvesting and power management system for ULP systems such as BSNs, WSNs or IoT.

Chapter 4

Modeling and Implementation of Power Management techniques for ULP Systems

We presented a low power clocking scheme and efficient, low voltage, low power EHM circuits in Chapter 2 and Chapter 3. These circuits improve the overall lifespan of a ULP system by decreasing the power consumption, increasing the harvested energy, and reducing the operating voltage of the system. Further improvement of ULP system lifespans can be achieved by employing power management techniques, such as dynamic voltage and frequency scaling (DVS). The improvements from DVS are orthogonal to the design improvements presented in Chapter 2 and Chapter 3. The application of DVS is achieved by changing the supply voltages. Design of DC-DC converters and their properties plays a significant role in the actual implementation of DVS. This Chapter presents a model of a DC-DC converter that can be used to study different power management techniques. The model helps us in studying the overhead and impact of various power management techniques reported in literature. Based on the model, we can determine the merit of one power management technique over the other. Further, a design of single inductor multiple output (SIMO) DC-DC converter is presented to implement a block level DVS, and panoptic dynamic voltage scaling (PDVS). The PDVS technique is studied using the proposed model and is identified as a more practical and efficient way to implement DVS.

4.1 Modelling for Power Management³

This section presents a model for inductor-based DC-DC converters and their application to quantify the benefits of power management techniques for ultra-low power (ULP) SoCs. Various power management techniques, such as dynamic voltage and frequency scaling (DVFS), clock gating, and power gating are now commonly employed in many SoCs. However, the power benefits of these techniques cannot be quantified accurately without assessing their impact on the DC-DC converter that delivers power. For example, DVFS uses a high voltage to support higher performance and lower voltage to save power. However, changing the output voltage of a DC-DC converter incorporates significant power overhead, and the efficiency can vary widely across voltage and current loads. These overheads may offset the benefits realized from DVFS. There is a need to characterize the benefits of power management techniques like DVFS in conjunction with their impact on the DC-DC converter. This is particularly important for ultra-low energy nearor sub-threshold systems that operate in a very dynamic power environment, and whose power constraints are stringent. This section presents a model that enables the study of power management techniques by taking into account their impact on DC-DC converters of different topologies. The model is based on an analytical treatment of inductor-based DC-DC converters, and it captures their efficiency trends with varying current load and output voltage. Since parameter selection will influence the specific behavior of the modeled converter, the model provides a rapid and effective tool for early design phase exploration of the impact of a converter, by using parameters based on different prior designs or by sweeping parameters to investigate the optimal design requirements for the overall system.

³ A lot of content of this section comes from [AS9] and [AS4]



Figure 4-1 Structure of the proposed model

Figure 4-1 shows the structure of the proposed model, which is broken into two parts. First, a model of inductor based DC-DC converters shown in Figure 4-1 (a) takes the operating condition of a workload as an input, which includes peak efficiency, load at which the peak occurs, minimum efficiency, maximum and minimum output voltages, settling time, and the decoupling capacitance of the DC-DC converter, and generates an efficiency calculation. In order to capture the dynamic load condition of the converter in the context of the operation of a specific power management scheme, the second part of the model shown in Figure 4-1 (b) uses one or more of the DC-DC converter models in a larger system model that includes the input voltage (Vi), output voltage (Vo), load current (I_L), time of operation, parasitic capacitance on the block, switching frequency, and activity factor. These parameters can change dynamically in power management techniques like DVFS. Using these parameters as input, the model calculates the overhead cost and change in the efficiency of each DC-DC converter in the system and provides the total system level energy consumed while executing a power management technique for the given workload profile.

The energy savings for a power management technique are typically reported at the load circuit operating voltage and load level in literature [75]. For example, the authors in [75] report the
energy savings obtained by scaling the voltage and frequency to a lower value. The paper does not calculate the total energy drawn from the original source of the supply voltage, for example a battery, which may not change linearly with load at the final circuit. Changing the operating condition of a voltage regulator causes deviation from its optimal behavior. If the output voltage of a buck converter is reduced, the efficiency of the converter decreases. Also, reducing the output voltage means that the capacitor C_L is discharged to a lower voltage by dissipating its stored energy. The actual benefits can be obtained by taking these losses and overhead into account. Figure 4-2 shows a block diagram of a typical inductor-based DC-DC converter. It includes a bias generator and comparators that cause the static loss. The control scheme that implements the switching pattern of the power switches MP and MN can vary across topologies. The switching loss is a function of the control scheme and the load. The power switches MP and MN, parasitic resistance of inductor (LPAR), and capacitor (CESR) cause the conduction loss, which is determined by the load current and output voltage. These losses are all a function of the operating condition. The proposed model accurately predicts the trends in behavior of DC-DC converters across topologies, implementing both pulse width modulation (PWM) and pulse frequency modulation (PFM) control schemes.



Figure 4-2 Loss mechanisms inside a typical switching DC-DC converter

4.1.1 DC-DC converter model

In order to model DC-DC converter trends and to capture the impact of DC-DC converters on power management strategies, it is essential to account for the converter efficiency, which is the power delivered to the load divided by the total power drawn by the converter. This efficiency of the DC-DC converter is a function of its output load, output voltage (Vo), the switching frequency, the switch resistance, and the parasitic resistance in the inductor and capacitor, as shown in Figure 4-2. In this section, we derive models for the efficiency for both PWM and PFM control schemes. Additionally, it is important to model how the converter will respond to changes in its usage. In a dynamic power environment like DVFS, Vo and load current vary dynamically, which changes the efficiency of the converter and results in energy overhead. Also, the converter can take significant time to settle from one voltage to another, resulting in timing overhead. Additional energy overhead comes in the form of charging and discharging of the decoupling capacitor. To quantify the benefit of a given power management technique like DVFS, we need to account for these overheads, in addition to modeling the efficiency at a fixed load. In this section, we derive and describe the proposed model for an inductor based DC-DC converter that accounts for these overheads and that can be used in a larger system model to study specific power management techniques.

DC-DC Efficiency with Load Current: PWM Control Scheme

Voltage and frequency are varied in DVFS to trade off power consumption with speed. This changes the load current and output voltage of the DC-DC converter, which changes its efficiency. In this section, we define a model that captures the change in efficiency that results from changing

load conditions. One prior work [76] models the power loss in a DC-DC Buck converter using a PWM switching scheme with the following equation:

$$P_{Buck} = a \sqrt{(I_L^2 + \Delta i^2/3)f_s} + b \left\{ \frac{I_L^2}{\Delta i f_s} + \frac{\Delta i}{3f_s} + \frac{C_L V_{DD1}^2}{R_{L0}\Delta i} \right\} + df_s \Delta i$$
(4.1)

where I_L is the load current, Δi is the current ripple in the converter, fs is the switching frequency, C_L is the decoupling capacitor, R_{L0} is the inductor series resistance, and *a*, *b*, and *d* are constants. Equation (4.1) represents the power loss in terms of various constants that cannot be obtained and that are non-intuitive to approximate prior to the design of the converter, so it is difficult to apply this equation for design space exploration or for general modeling of DC-DC converter trends. Instead, we propose a model that accurately captures the trends in the efficiency of the DC-DC converter in terms of the peak and minimum efficiency values of the converter, which can be either predicted, specified as targets, or pulled from prior work. To derive this simplified model, we begin by following previous work [76][77] in the observation that (4.1) leads to an efficiency in the form.

$$\eta_{I_L} = \eta_2 - (\eta_2 - \eta_1) * (\log(I_L/I_0))^2/4$$
(4.2)

where η_2 is the peak efficiency occurring at load Io, and η_1 is the least efficiency at a given load. For the verification of the model proposed in equation (4.2), let us consider the following cases. For a light load condition in Equation (4.1),

 $I_L \sim \Delta i$

so the power loss given by equation (4.1) in the buck converter takes the form of

$$P_{Buck} = \alpha I_L + \beta / I_L$$

The efficiency of the converter will be given by power delivered (VIL) over power drawn:

$$\eta_{I_L} = V I_L / (V I_L + \alpha I_L + \frac{\beta}{I_L})$$
(4.3)

Expanding equation (3) using Taylor's series we get,

$$\eta_{I_L} = V/(V+\alpha) \{ 1 - \frac{\beta}{(V+\alpha)I_L^2} + \frac{\beta^2}{(V+\alpha)^2I_L^4} + \cdots \}$$
(4.4)

It is clear from Equation (4.3) that the efficiency decreases as load current decreases for the cases of light load conditions on the converter. The proposed equation in the model is given by,

$$\eta_{I_L} = \eta_2 - (\eta_2 - \eta_1) * (\log(I_L/I_0))^2/4$$

This expression also decreases for the light load condition, capturing the correct behavior of DC-DC converter efficiency trend. We know from Taylor's series that,

$$\ln(I_L/I_0) = -\left\{ \left(\frac{I_0}{I_L} - 1\right) - \frac{1}{2} \left(\frac{I_0}{I_L} - 1\right)^2 + \cdots \right\}$$
(4.5)

Since, $I_0 \gg I_L$ and converting natural logarithm into logarithm of base 10 we get,

$$\log(I_L/I_0) = -2.303\{\left(\frac{I_0}{I_L}\right) - 1/2\left(\frac{I_0}{I_L}\right)^2 + \dots\}$$

so the efficiency equation in (4.2) can be rewritten as,

$$\eta_{I_L} = \eta_2 - 2.303(\eta_2 - \eta_1)/4 * \left\{ \left(\frac{I_0}{I_L} \right)^2 - \left(\frac{I_0}{I_L} \right)^3 + \cdots \right\}$$

$$\ln(I_L/I_0) = -\left\{ \left(\frac{I_0}{I_L} - 1 \right) - 1/2 \left(\frac{I_0}{I_L} - 1 \right)^2 + \cdots \right\}$$

$$Using, x > \log x$$
(4.6)

The proposed model in (4.2) for the DC-DC converter reduces to (4.6) under light load, which follows the behavior of the DC-DC converter as reported in literature [76] and compares well with (4.4) with maximum efficiency η_2 in (4.2) can be obtained by equating the constants in (4.4) and (4.2),

$$\eta_2 = V/(V+\alpha) \tag{4.7}$$

Equations (4.2) and (4.6) are not bounded for the cases when I_L becomes very small, but they can predict the behavior for light load condition in a PWM control switching scheme based DC-DC converter with less than 5% error. Figure 4-4 shows this result.

Now consider the case when load is very high compared to the point of peak efficiency. The constant *a* in equation (4.1) represents the resistance of the MOS transistor used for switching [76]. The model of [76] assumes it to be a constant. However, the resistance of the MOS transistor increases with the increase in load current and it is given by,

$$R_D = K * (1 + \frac{I_L}{I_{DSAT}})$$
(4.8)

where I_{DSAT} is the saturation current of the transistor. Clearly, as load current increases, resistance increases. For light load condition, it is correct to assume that resistance doesn't change, as I_{DSAT} is much larger compared to I_L . However, with an increase in load, the MOS resistance increases, causing elevated conduction loss. Also, at higher load the increased current in the inductor causes elevated conduction loss in the inductor's parasitic resistance. Overall, the I²R loss increases, because of the increase in current and because of the increase in resistance caused by that increase in current. For a high load we know that,

$$I_L \gg \Delta i$$

so the power loss in the buck converter using equation (4.1) of [76] takes the form of,

$$P_{Buck} = \alpha_1 I_L + \beta_1 I_L^2$$

The equation for efficiency can be written as,

$$\eta_{I_L} = V I_L / (V I_L + \alpha_1 I_L + \beta_1 I_L^2)$$
(4.9)

This expression shows that efficiency decreases as the load current increases. Using Taylor series expansion, this expression can be written as,

$$\eta_{I_L} = V/(V + \alpha_1) \{ 1 - \frac{\beta_1 I_L}{(V + \alpha_1)} + \left(\frac{\beta_1 I_L}{(V + \alpha_1)}\right)^2 + \dots \}$$
(4.10)

In the proposed equation in this paper, equation (4.2), the efficiency also decreases for the higher load. Expanding (4.2) using Taylor's series,

$$\eta_{I_L} = \eta_2 - (\eta_2 - \eta_1)/4 * \{k1 + k2\frac{I_L}{I_0} + k3\left(\frac{I_L}{I_0}\right)^2 + \dots\}$$
(4.11)

Equation (4.11) and (4.10) follow closely, with constants η_2 , η_1 , kI, etc. in (4.11) can be obtained by equating with respect to the powers of I_L in (4.10). The proposed equation matches the trend of equation reported in literature [76].

While equation (4.1), (4.4) or (4.10) can be more analytical versions of the DC-DC converter efficiency formulation, they are not very useful for early design space exploration or for studying system level power management techniques because of the unknown constants. In contrast, the compact model in (4.2) can be expressed in terms of peak efficiency and minimum efficiency, making it easy to use.

DC-DC Efficiency with Load Current: PFM Control Scheme

In the PFM control scheme with a constant peak inductor current, the switching and conduction loss scale with frequency, and the efficiency remains flat for higher loads. The power loss is given by,

$$P_{Buck} = K * P_{Out} + C \tag{4.12}$$

where K and C are constants. The first term indicates switching and conduction loss that scales with frequency. The second term indicates the static loss that does not scale with frequency. Therefore,

$$\eta_{I_L} = \frac{P_{Out}}{(P_{Out} + K * P_{Out} + C)}$$
(4.13)

Using P_{Out}=VI_L and expanding using Taylor series,

$$\eta_{I_L} = \eta_2 - \frac{a}{I_L} \tag{4.14}$$

Equation (4.14) shows that the efficiency increases and becomes constant as load increases. This happens because power loss in PFM schemes scales with the load. At light load condition, the static loss dominates and reduces the efficiency.

Verification of the model:



Figure 4-3 Efficiency Variation with load current in a) PWM scheme with $\eta_2=0.9$, $\eta_1=0.68$ and $I_L=1mA$ b) PFM scheme with $\eta_2=0.88$ and a=5x10-5

The equations we have provided accurately model the trends for PWM, PFM, and combinations of those control schemes. Figure 4-3 shows the variation of efficiency with respect to load when using dedicated PWM and PFM control schemes. In the PWM scheme, the efficiency degrades at high load and at light load conditions, while in the PFM scheme the efficiency remains flat as load increases and degrades at light load conditions. Figure 4-4 shows the comparison of the model



a) Comparison of model with the measured PWM DC-DC converter reported in literature



b) Comparison of model with [8], implementing both PFM and PWM



c) Comparison of model with [6], schemes implementing both PFM and PWM

Figure 4-4 Comparison of the load equation with measured work in literature

equations for PWM and PFM control schemes, with the measured results reported in literature [78]-[81]. Figure 4-4 (a) shows the results for the PWM scheme. For equation (4.2), we have selected η_2 as the peak efficiency reported in the corresponding paper and I_O as the load for that

peak efficiency. The value of η_1 is obtained experimentally, based on [78]-[81], and we set η_1 =10 for all the papers ([78]-[81]) employing the PWM control scheme. The converters [78]-[80] and [81] in part implement PWM. We find that the model predicts the efficiency behavior of the converters very accurately (<5% error for these papers). For [78], [80], and [81], the error is less than 3%.

We also compared the model with reported works that employ the PFM switching scheme. Figure 4-4 (b) and Figure 4-4 (c) show the results. For the PFM scheme, we used η_2 as the peak efficiency reported in the corresponding paper. The constant *a* represents the static loss of the converter and will vary from one converter to another. It causes the degradation in efficiency at light load condition in a PFM control scheme. For this comparison, we set the value of constant *a* in (4.14) to match the least efficiency reported in each paper. The model predicts the behavior of [79] correctly for the PFM scheme, while it deviates at higher load for [81]. This is because we assumed in our model that the PFM scheme is used only for a light load condition, whereas [81] shows results for loads up to 400mA.

To further illustrate the usefulness of the model, we also compared it with the schemes where both PFM and PWM schemes are employed. This is often done to increase the range of load a converter can support. Figure 4-4 (b) and Figure 4-4 (c) show the results of comparison of the model, with efficiencies reported in [79] and [81]. The model accurately predicts the behavior of such converters. There can be other combinations in which the model can be used. For example, it can be used for a segmented switch scheme if two or more curves for (4.2) are used in conjunction, with each curve having a peak efficiency (η_2) corresponding to a different value of Io.

The proposed model is accurate in predicting the efficiency trend with respect to the load if the control scheme is PWM or PFM. Also, an approximate peak efficiency of a DC-DC converter can be obtained very early in the design cycle, as it is dependent on the technology, size of the inductor, and ripple on rails. Therefore, the model can be employed for analyzing the DC-DC converter overheads early in design, while implementing power management techniques.

Efficiency with Output Voltage

The peak efficiency of an inductor-based DC-DC converter decreases with a decreasing output voltage. For a given load current, the switching loss and conduction loss of the converter remain the same. The decreased output power level at lower voltages results in decreased efficiency. The efficiency as a function of voltage can be modeled as,

$$\eta_V = \eta_1 + m(V - V_{min}) \tag{4.15}$$



Figure 4-5 Dynamic efficiency variation with current and voltage

where *m* is the slope of the line given by $m = (\eta_2 - \eta_1)/(V_{max} - V_{min})$. This approximate linear behavior is reported in both [76] and [78]. Plugging (4.2) for PWM or (4.14) for PFM and using (4.15), lets us write the combined voltage and load efficiency as:

$$\eta = \eta_V * \eta_{I_L} \tag{4.16}$$

Figure 4-5 shows the output of the proposed model with V_0 and with load current assuming a PWM control scheme. A DC-DC converter designed for a specific voltage and load will follow this trend when its load current or output voltage changes.

Settling Time

The settling time of a converter is the time it takes to reach the desired supply voltage. A typical converter has a large inductor and a large filter capacitor that makes the settling time very large (few μ s to ms [82]). In a dynamic environment such as DVFS or Ultra DVS (UDVS), the output voltage V₀ is expected to change. The settling time of a converter to reach the desired voltage becomes an important overhead for these scenarios. The settling time Δ T in our proposed model is approximated as,

$$\Delta T = \frac{T}{V} * \Delta V \tag{4.17}$$

when output voltage is charged to V from ground, and where T is the settling time of the converter. We assume that the inductor carries the same amount of current for each cycle of charging. Consequently, the rate at which the output reaches a given voltage will be linear with time. T/V is the slope of this curve and is set by assuming that at each switching cycle the inductor carries the same amount of current even for the cases when the output voltage is rising from zero.

Supply Rail Switching Energy

The change in the output voltage of a converter results in a change of the stored energy on the capacitance of the supply voltage rail of the load blocks. Some of this stored energy is dissipated if the new voltage is lower than the previous voltage, whereas energy is consumed from the source supply, Vin, if the new voltage is higher than the previous voltage. The additional energy overhead Ec is given by equation (4.18), where V_1 and V_2 are the new and previous voltages of the converter. If V1 is greater than V2, work is to be done by the supply Vin. When V1 is less than V2, no work is done by Vin; hence, energy overhead will be zero.

$$E_C = V_{in}C_L \max(V_1 - V_2, 0) \tag{4.18}$$

In some cases, the voltage on the capacitor is not immediately discharged to lower voltage. V_{DD} slowly discharges from V2 to V1 while running workloads. In such cases, E_C will be lower than given by equation (4.18). Equation (4.16) helps us to predict the losses at a given load or voltage condition, while (4.17) and (4.18) give the conversion energy and timing overhead. These equations enable a framework where overheads that originate from dynamic changes to the DC-DC converter output can be calculated for techniques like DVFS to accurately measure their energy benefits.

4.1.2 Evaluation of DVS techniques using the proposed model

The DC-DC converter model can be used for assessing a variety of block level power management techniques. Since the model captures both the efficiency of the converter for fixed loads and the cost of making dynamic changes to the converter output voltage and load, we can use it to model system level implementations of varying complexity. In the most basic case, the model can provide additional insight into the system level cost of reducing the voltage delivered to a block or to a

chip. For example, Figure 4-6 shows the measured energy consumption of a microcontroller in a sub-threshold system on chip [86] across voltage. The minimum energy voltage occurs at below



Figure 4-6 Energy consumption for a microcontroller across voltage with and without consideration of the DC-DC converter efficiency.

0.3 V. However, when we consider the amount of energy drawn from the battery or energy storage node, including the overhead of loss in a DC-DC converter, the situation changes significantly. The top line in Figure 4-6 shows the energy consumed from the source (at the input to the DC-DC converter) using our model, which was fitted to low load DC-DC converter measurements from [87] to illustrate an example converter for this system. Since the output voltage and load current both vary for the block as V_{DD} decreases, it is not accurate to assume a constant efficiency loss in the converter, and our model captures the changing efficiency across the space. This result shows that reporting block level consumption only can result in an inaccurate view of the total impact on the battery, and that the actual optimal voltage [92] for minimizing energy consumed at the battery may be higher than anticipated from block measurements alone.



Figure 4-7 Efficiency profiles for two different converters (a) and the impact of a converter change on the overall energy drawn from the battery (b)

Further, the DC-DC model can help designers to choose the optimal specification for a converter to use for a given block or chip. This is especially important for embedded converters serving extremely low power systems like the one in [86], which operates from harvested energy. For example, we revisit the design in Figure 4-6 and consider a different converter design with lower overall peak efficiency, but whose peak efficiency occurs at a lower voltage. Figure 4-7 (a) shows the efficiency versus V_{DD} for two different converters: one is the converter used in Figure 4-6 and modeled on [87], and the other is a hypothetical converter. The new converter has a lower overall peak efficiency than the original design, which might lead to the misconception that it will hurt the overall energy consumption of any chip. However, its peak efficiency comes at a lower voltage, so its efficiency scales better across the lower range of voltage values. The impact of this is that, at lower voltage (and lower current) values, the second converter provides more efficient operation. Figure 4-7 (b) shows the impact of using this converter alongside the curves from Figure 4-6. Not only is the total energy drawn from Vin much lower, the optimal voltage for minimizing energy

occurs at a voltage much nearer to the optimal voltage of the block. The flexibility of our model allows a designer to experiment with different converter specifications while co-designing an embedded converter with its loading system, providing for rapid design space exploration early in the system design process.



Figure 4-8 Dedicated DC-DC converter per block.

In addition to supporting the co-design of embedded DC-DC converters, our model can enable higher level comparisons of power management techniques that apply to multiple blocks. The next example illustrates the application of our model to two different power management strategies. DVFS is commonly used to save power in a SoC by changing the supply voltage at the full chip level. Even larger energy savings can be realized by implementing block level DVFS, so that each block can use a voltage that is best matched to its own workload. Figure 4-8 shows the idealized implementation in which each block has a dedicated DC-DC converter. However, it may not be practically possible to implement such a system because of the area and cost of replicating DC-DC converters for each block. We study this scheme using our model to analyze its benefits by taking into account the overheads discussed earlier in the chapter. Later on, we compare this with a more practical implementation of block level power management.

VDD	Load	Time of operation
V1	i1	Τ1+ΔΤ1
V2	i2	Τ2+ΔΤ1

VDD	Load	Time of operation
0.9V	100µA	8µs
1.2V	900µA	8µs

Optimal Condition for a Block

Example Table

Figure 4-9 Operating condition for dedicated DC-DC.

Framework for Energy Calculation in DVFS

In this section, we establish the framework for computing the system level energy consumption of a multiple block DVFS system, where individual DC-DC converters are modeled using the equations from Section 4.1.1. Figure 4-9 shows the operating condition of an example block that has a dedicated DC-DC converter. V_{DD} and load will change with time. We have assumed a uniform random distribution for the power supply voltage setting in the range of 0.4V to 1.2V. ΔT is the settling time of the converter, and we use T=20µs and V=1V in equation (4.18) [76]. The load capacitor on each block is assumed to be 200pF.

$$E_{op} = \frac{V_1 i_1 (T_1 + \Delta T_1)}{\eta_1} + \frac{V_2 i_2 (T_2 + \Delta T_2)}{\eta_2} + \cdots$$
(4.19)

Eop is the operating energy and η is calculated using equation (4.2).

$$E_{C} = V_{in}C_{L}(\max(V_{1} - V_{2}, 0) + \max(V_{2} - V_{3}, 0) + \cdots)$$
(4.20)

$$E_{TOTAL} = E_{OP} + E_{O}$$

Each block is modeled as a chain of inverters with different depths. The delay of the block is calculated as its time of operation. The power supply level changes for 100 iterations. The rate at which the voltage changes to a new value is varied from 10ns to 1ms. The energy dissipation in

each case is compared with a single V_{DD} (always 1.2V) block. Figure 4-10 shows the result of our experiment. We find that at fast rates of voltage scaling (a voltage transition every~10ns), the overheads of a DC-DC converter dominate, and there is an energy loss. Energy benefits can be realized for T_{OP} greater than 1µs with a maximum benefit of more than 150% achievable at slower rates of V_{DD} transitions. This implies that, for these assumptions, the 5 V_{DD} system would save energy relative to the single V_{DD} system, when transitioning V_{DD} to adapt to changes in the workload that are slower than ~1µs.



Figure 4-10 Energy Savings with rate of Voltage Scaling

Panoptic Dynamic Voltage Scaling

This section applies the DVFS modeling approach to a different DVS implementation. Figure 4-11 shows the block diagram of a block level voltage scaling technique called panoptic dynamic voltage scaling (PDVS) [88]. In the PDVS technique, a block can switch from one voltage to

another by the use of headers, as shown in Figure 4-11. The advantage of this technique is that it enables a much faster switching. An equivalent DVFS voltage can be realized by dithering between the supplies. This scheme is more practical and has lower cost.



Figure 4-11 PDVS: Block Level Voltage Scaling Technique

Framework for Energy Calculation in PDVS

We reproduce the operating condition of a block from Figure 4-9. This block operates at different voltages for different times to accomplish optimal energy operation. We break down the operating condition of Figure 4-9 into Figure 4-12. If a block has to operate at V1 (0.4V<V1<0.8V) for T1 time, PDVS accomplishes it by connecting the block to 0.4V for T11 and 0.8V for T12, such that T11+T12=T1 of Figure 4-12. This approach is called voltage dithering. T11 and T12 are such that the performance of the block does not change. A final operating condition is given by Figure 4-12 (b). The load on each supply will change, depending on the blocks that are connected to it, and results in a continuous time varying load on each supply. We include time to obtain the energy. Each supply has larger load variation, which will have an impact on the overall efficiency. The total energy is given by,

$$E_{OP} = 0.4 \int i_1(t)/\eta_1(i_1)dt + 0.8 \int i_2(t)/\eta_2(i_2)dt + 1.2 \int i_3(t)/\eta_3(i_3)dt$$
$$E_C = V_{in}C_L(\max(V_1 - V_2, 0) + \max(V_2 - V_3, 0) + \cdots)$$
(4.21)

$$E_{TOTAL} = E_{OP} + E_C$$

We keep the same system set-up as was used for the dedicated DC-DC converter case. It should be noted that there will be an insignificant overhead of settling time in this case. We assume a capacitive load of 20pF on each block, since the local block virtual V_{DD} rail switches instead of the total chip-wide V_{DD} rail (with all of its decoupling capacitance).

VDD	Load			T _{OP}
	0.4V	0.8V	1.2V	
v1 V	i11 µA	0	0	T11
v1 V	0	i12 µA	0	T12
v2 V	0	i21 µA	0	T21
v2 V	0	0	i22 µA	T22

a) Operating Condition

Load			dT _{OP}
0.4V	0.8V	1.2V	
$i_{l}(t)$	$i_2(t)$	<i>i</i> ₃ (<i>t</i>)	1e-9

b) Final load table

Figure 4-12 PDVS Operating Condition Evaluation

4.1.3 Comparison of DVFS and PDVS using the Model

Figure 4-13 compares the PDVS scheme with a dedicated DC-DC converter case using the system level model that incorporates the DC-DC converter equations. The PDVS scheme has a break-

even time of 30ns compared to 1µs in the dedicated supply case. This is because of the very small settling time in PDVS, as the block is charged to the rail almost instantly. It also has lower conversion energy. The total energy benefits from the PDVS scheme are, however, lower than the dedicated DC-DC converter case, because it sees much wider load variation. The PDVS scheme, however, is better- suited for implementing block level DVFS because of its much smaller break-even time, allowing it to adjust to short changes in the workload. PDVS scales much better to larger numbers of blocks, since it requires only one DC-DC converter output per voltage rail rather than per block, as in the dedicated DVFS scheme. These results are, of course, influenced by the values of the parameters in the model, and the model makes it very easy to investigate how the results will vary when the assumptions change.



Figure 4-13 Energy Benefits of PDVS and Dedicated DVFS

Figure 4-14 shows how the PDVS savings for the same scenario as before will change as a function of the capacitance of the virtual V_{DD} rail of each PDVS block. As the capacitance of the block

increases, the breakeven time moves to larger times, meaning that the workload needs to remain at a new value for a longer time to make it worthwhile to switch the voltage of that block.



Figure 4-14 Energy Benefits of PDVS for different values of the virtual VDD capacitance of the block

4.2 Implementation of Power Management for PDVS

Minimizing power consumption by employing power management techniques is a common practice today in integrated circuit (IC) design. Several techniques have been researched and applied to this end. Initially, Dynamic Voltage Scaling (DVS) [93], where the power supply of an Integrated Circuit (IC) is modulated according to its performance needs, has been successfully utilized to reduce the power consumption. Usually, voltage is scaled up for the IC to perform at a higher rate or scaled down to trade off performance for lower power consumption. Also, the IC can be divided into multiple voltage domains as multi-V_{DD} systems, where multiple power rails are used to operate non-critical paths at lower voltages to save power. Moreover, it has been

claimed that the lowest energy operating point is achieved when the processing rate matches the rate of the work-load. Such a system needs to operate an individual block at different voltages at different times.



Figure 4-15 Dynamic Voltage Scaling for multi-core system

Practical implementation of power management techniques, such as DVS, involves several overheads. Figure 4-15 shows the implementation of DVS for a single V_{DD} , multi-core system. The V_{DD} is scaled according to the performance or power needs of the IC. This implementation involves several overheads. The output capacitor (C_L) of a DC-DC converter is usually large; therefore, it has a large settling time. The energy stored on the capacitor is also high, so changing the output voltage involves energy overheads. Usually, these overheads limit the rate at which V_{DD} can be scaled [95] and, hence, the amount of energy that can be saved. The flexibility of DVS is also limited, as individual cores do not operate at their optimal voltages. Operating each core with its own V_{DD} can save more power. However, the number of DC-DC converters needed for this purpose increases linearly with core. Low Drop Out (LDO) and switched capacitor converter present on-chip options for implementing DVS for the multi-core system, albeit at lower efficiency, which again limits power savings.



Multi-core system

Figure 4-16 PDVS implementation for multi-core system

Panoptic dynamic voltage scaling (PDVS), as explained in the previous Section, has been proposed [97] to overcome these limitations of DVS. Figure 4-16 shows the implementation of a PDVS scheme. Each core inside a multi-core system is connected to three different rails through header-switches. The blocks can be connected to a given V_{DD} rail, depending on power or performance requirements. It allows the cores to switch from one voltage to another. Utilizing three different voltage levels, a block can be made to operate at its optimal voltage by using the voltage dithering technique [98]. Using PDVS techniques, several limitations of classical DVS can be overcome. In PDVS, each block can theoretically be made to operate at its optimal voltage which results in higher overall power savings [97]. The voltage rails in PDVS are fixed and the blocks are connected to these rails depending on their throughput requirement. The fixed voltage level in PDVS eliminates the settling time and energy overhead costs usually present in the classical DVS can implement a faster rate voltage scaling technique and can save more power [94][95]. However, the implementation of PDVS requires at least three DC-DC converters. The other cost involved is

higher area owing to the routing, switches and level-converter (LVL) needed to implement it. The routing, the switches, and the LVLs area overhead are less than 15% [94] for each block, which doesn't amount to significant cost given the energy benefit. However, the cost of three or more DC-DC converters needed to implement PDVS can amount to significant cost when compared to a system implementing a single V_{DD} DVS. Therefore, a practical PDVS system has not yet been implemented. This section presents a power management solution to implement PDVS.

Several design challenges have to be overcome to implement a low cost power management solution for PDVS. We need more than three DC-DC converters to implement PDVS. A simple implementation will be to have as many dedicated DC-DC converters as needed by the PDVS scheme. However, this will still be costly and not practical. A single inductor multiple output (SIMO) [96] DC-DC converter, where one inductor is used to generate multiple output supplies, is proposed for PDVS. The three output rails for PDVS are generated through SIMO architecture which is a lower cost, higher efficiency solution. To further reduce the cost and system volume, the capacitors are integrated. The integration of capacitors is only possible if smaller capacitances are used. However, use of small sized capacitances will increase the ripple on the power supply. We use small sized on-chip capacitances. The ripple on the power supply is reduced through a proposed hysteretic control scheme. Further, the use of SIMO will also result in higher ripple and cross regulation issues in the design, because of the changes in loads on different V_{DD} rails. This issue is addressed by designing the SIMO converter to be able to configure itself based on the load information which is available in a PDVS system. This section proposes the design of a SIMO DC-DC converter with on-chip capacitors, utilizing the features of PDVS technique. It provides a cost efficient, energy efficient way to implement block level DVS. The proposed circuit is the first reported practical implementation for PDVS and implements low cost and efficient PDVS. The use of SIMO reduces the cost of multiple DC-DC converter requirements for the implementation. The proposed solution provides three output rails at 0.9V, 0.7V, and 0.4V, at peak efficiency of 86% with integrated capacitance.

The design description of the proposed circuit is divided into the following sections. First, we talk about the overall system architecture and the SIMO control scheme. Second, we will describe the hysteretic control scheme which helps in reducing the size of the capacitors. Third, we will describe the scheme for co-designing SIMO with PDVS load environment which helps in dealing with cross-regulation and higher ripple arising from SIMO architecture. Finally, we will present the measurement results and compare the proposed design with the state-of-the-art.

4.2.1 PDVS SIMO Architecture

Figure 4-17 shows the block diagram implementation of the proposed SIMO DC-DC buck converter to drive an example PDVS system. It consists of hysteretic comparators, a SIMO controller, and buck DC-DC converter, and provides three output rails at $V_{OH}=0.9V$, $V_{OM}=0.7V$, and $V_{OL}=0.4V$. The comparator, SIMO controller and the DC-DC converter implement the control loop to provide the output voltages. The SIMO controller steers the inductor current (I_L) to different rails through switches S1-S3. The hysteretic comparator compares each rail to its reference voltage and provides a digital output. It also controls the switching turn-on time of the comparator, which we will explain later in this section. The hysteretic comparator controls the regulation and switching of each rail. These comparators can be disabled when a particular rail is



Figure 4-17 SIMO DC-DC buck converter for low voltage DVS-enabled. On-chip decoupling capacitance saves system level volume and cost

not needed in the system. The digital output of the comparator goes into the controller, along with the priority signal from the PDVS controller. These signals are used to generate control for the DC-DC converter, as well as to assign switching sequence for the switches S1-S3. The priority signal from the PDVS controller provides the current loading scenario on each rail. The SIMO controller uses this signal to set the switching priority, and assigns highest and lowest priority to switches S1-S3. For example, if 0.9V rail is heavily loaded, the priority is set on this rail and, in the event that 0.9V comparator output goes low, the SIMO controller starts steering inductor current into 0.9V rail. In a case where all three rails are heavily loaded, this switching configuration can result in higher ripple or cross regulation. However, it is not possible for all of the rails to be loaded simultaneously in a PDVS system. As a result, higher ripple is typically not seen on other rails.

Control Architecture

The design goals of the control architecture are following. First, the architecture should be able to support lower sized capacitance with small ripple. Second, it should provide high efficiency and, finally, the static power consumption of the circuit should be small. We will first explain the control of an individual controller making the SIMO to operate as a simple buck converter providing a single V_{DD}. This can be achieved in the system by disabling other two converters by disabling their comparator. The design decisions at this level are used to reduce the overall system power and cost. The switching time of the converter is controlled by the comparator, which helps reduce the additional control circuit, which, in turn, reduces static power consumption of the system. We used two approaches to reduce the dimensions of the passives in the converter. First, we used 65nm advanced process nodes compared to [96][68][69][99][100][101]. The smaller CMOS technology will enable faster switching frequency for the converter, which will enable lower sizes for inductor and capacitors. Secondly, we propose a new hysteretic control scheme, which further brings down the size of the capacitor to nF range.

Figure 4-18 shows the conventional control scheme to generate the High Side (HS) switching control for a DC-DC converter. Usually, two approaches are common. In the first approach, a fixed delay is generated and the turn-on time of the HS switch, MP, is controlled by the fixed delay. This delay generates the current in the inductor and HS switching is enabled, followed by Low Side (LS) switching. The HS control determines the amount of energy being transferred in each cycle.

The second approach for HS control uses current sensing of the inductor and is more popular for higher output power switching converters. However, this scheme suffers from inaccuracies and higher energy overhead [46], which makes it unsuitable for low-energy lower voltage systems. Also, fixing the current of the inductor requires a high amount of decoupling capacitor to reduce the ripple. The inductor current gets stored on the capacitor for the cases of light load. This can cause higher ripple if the capacitor is small.



Figure 4-18 Conventional HS control



Figure 4-19 Proposed control scheme

Figure 4-19 shows the circuit of a proposed HS control scheme. It uses a hysteretic comparator to control the HS switch. The hysteresis of the comparator and the delay of the comparator set the ripple on the rail. The high side transistor, MP, is turned ON when V_0 goes below Th_{LO} of the

hysteretic comparator, where Th_{HI} and Th_{LO} are the higher and lower thresholds of the comparator, respectively, with hysteresis given by, Th_{HI} - Th_{LO} . The inductor starts charging the output rail and its current starts ramping up. Once V_O crosses Th_{HI} , MP is disabled and inductor current starts discharging through to the output capacitor.



Figure 4-20 Control Architecture to reduce the ripple

Figure 4-20 shows the behavior of the HS control circuit. At the light load condition, the voltage at the output rises quickly, since the current drawn from the output capacitor is low and most of the current is used to charge the capacitor. Therefore, the hysteresis of the comparator sets a lower inductor current and insuring lower ripple. If the load current increases, the rise time of the output voltage increases, and, as a result, MP is on for a longer duration of time (since higher current is drawn out of the output). This increases the peak current in the inductor. The proposed circuit adapts itself with the output load. This scheme makes the ripple on the rail less dependent on the output capacitor. Figure 4-21 shows simulation results of the variation of the output ripple of the

capacitor, with load current at different values of the decoupling capacitor. The ripple voltage varies from 30-60mV for 0.8V V_{DD} and 1.2V Vin. We selected 4.3nF of output capacitor, which gives us ~5% ripple on the rail. This value of capacitance is significantly smaller than the typical decoupling capacitors (µF range) used on output rails in a DC-DC converter. At these values, the capacitance can easily be integrated on-chip. A significant percentage of the capacitance can come from the parasitic capacitance of the cores connected to these rails [105].



Figure 4-21 ripple Voltage for different values of decoupling capacitor at different loads

The proposed HS control scheme supports a load up to 50mA and can operate the converter in both continuous conduction mode (CCM) and discontinuous conduction mode (DCM). At light load condition, the converter goes into DCM. The HS turn on time is used to charge the inductor. After LS control cycle, the inductor current goes to zero. However, V_0 goes below Th_{LO} later than LS cycle, due to the light load condition. The HS control scheme starts again, once V_0 goes below Th_{LO}. Operating in DCM at light load helps in achieving good efficiency as well as controlled ripple. The converter operates in CCM at high load condition. At high load condition, V_0 goes below Th_{L0} before the inductor current goes to zero and, hence, conducts continuously. Operating in CCM helps in targeting heavy load condition for the converter. Figure 4-22 shows the simulation results of the output voltage and inductor current at light load and heavy load conditions, respectively. It shows that the converter operates in CCM at high load and in DCM at light load condition.



Figure 4-22 Converter Operation in CCM and DCM mode

The static power consumption of the HS circuit is dictated by the power consumption of the comparator. More power can be saved by lowering the comparator quiescent current. However, the performance of the comparator also controls the amount of ripple seen on the rail. For example, if the comparator has higher delay (lower power), then response of the comparator to the changes at the output voltage will be slower, which will result in elevated ripple. Figure 4-23 (a) shows the variation of ripple with the comparator quiescent current. The ripple decreases with increased

quiescent current. The ripple becomes constant after 25μ A of comparator current. After this point, the hysteresis of the comparator and output capacitor controls the amount of ripple. We select 25μ A quiescent current for the comparator of 0.9V and 0.7V rails and use two comparators for 0.4V rail. One comparator has 3μ A, while the other has 100nA quiescent current. This is done to lower the static power consumption of the converter for the low power mode, where all the cores in the PDVS system can be connected to 0.4V. The other two converters can be disabled, while 0.4V rail will supply the V_{DD} with higher ripple.



Figure 4-23 Transient behavior of the converter

Hysteresis of the comparator is also important in determining the peak inductor current, which is important in determining the overall efficiency of the converter, particularly at light load condition. At higher value of inductor current, conduction loss increases, thus decreasing the efficiency, while, at lower value, the switching loss reduces the efficiency. We control the inductor current through the hysteresis in the comparator. At high load condition, the losses are governed by the load current, as the converter operates in continuous conduction mode. The comparator and the output capacitor are also important for the transient behavior of the converter. Figure 4-23 (b) shows the condition when output load changes from 100μ A to 10mA in 10ns. It shows that the converter can continue to regulate even for a fast change in load condition. However, owing to the small size of the output capacitor, a very sharp change in output load to high load, such as 40-50mA, can result in overshoot or undershoot at the output rail. The converter needs a few µs to recover under such conditions. This happens because the converter goes from DCM to CCM in a very short time, which can result in higher inductor current build- up and can cause higher ripple. However, if the load changes slowly, then higher ripple is not seen on the rail.



Figure 4-24 Low Side control Scheme

Figure 4-24 shows the low side (LS) control of an individual controller. The low side control is implemented to keep the LS switch on for a fixed time. For the rest of the time, LS conducts as a diode. The LS switch is implemented with an L_{VT} device. As a result, the diode doesn't contribute significant loss. A conventional scheme for LS control implements a zero detection comparator. For light load condition and lower inductor current, as in the current implementation, the performance required for zero detect comparator is very high. For higher performance, static current in the comparator becomes higher, which adds significant overhead on the DC power

consumption of the converter. The scheme to implement fixed delay turn on time for LS eliminates this overhead with a smaller penalty in efficiency.

SIMO Digital Controller



Figure 4-25 Digital control logic of SIMO controller

Figure 4-25 shows the simplified circuit diagram of the SIMO controller. The HS control is enabled when any one of the three comparator outputs goes high. At this time LS is disabled. Once HS is disabled, LS turns on for a given pulse width, as shown in Figure 10. Switches S1-S3 are the SIMO switches for 0.9V, 0.7V, and 0.4V rails redrawn here from Figure 4-17. The priority selection selects between c1 and c2 and correspondingly S1 and S2. For example, if 0.7V rail has higher priority, then c2 gets connected to p1 and c1 to p2. Similarly, b1 gets connected to S2 and b2 gets connected to S1. Only one switch out of S1, S2, and S3 is turned on at a given time. Priority

selection plays a role in selecting a particular switch. If more than one rail is below its Th_{LO} , then the switch corresponding to the rail with higher priority is turned on. We assign higher priority to the rail with a higher load. This design is well-suited to the PDVS system. If one rail is heavily loaded, then other rails are going to be lightly loaded in PDVS. Also, since load information is well known in PDVS (through the header switch connection), priority can be assigned correctly.

In this scheme, the rail with higher load gets serviced first, which prevents it from having higher ripple. The rail with lower load discharges slowly and can be serviced in the meantime. The proposed circuit can see higher cross regulation in the case where difference in load becomes too large among rails. This happens in the CCM mode of operation. If one rail is heavily loaded with (40-50mA), while other rail is lightly loaded at (~10-100 μ A), then the lightly loaded rail can charge up because of the higher current present in the inductor. Several methods have been proposed to overcome this problem. One of the methods is to short both terminals of the inductor, which results in energy loss. We dump the extra current on the 0.4V rail, which has the least priority. In the event that voltage goes higher, a clamp can be used to control the voltage at 0.4V rail.

Control of Ripple through PDVS

Figure 4-26 shows the distribution of load current for different scenarios. It shows the cases when most of the cores are connected to 0.9V and 0.7V, respectively. It provides an insight that, if one V_{DD} rail is heavily loaded (case when most of the cores are connected to that V_{DD}), the other V_{DD} rails are lightly loaded. This is a unique characteristic of PDVS which can be used to the advantage

of SIMO converter design. This feature can be used to address the issue of cross-regulation in SIMO converters. Cross-regulation arises in SIMO converters usually operating in



b) current consumption when most of the cores are connected to 0.7V, total load current=8mA

Figure 4-26 PDVS load conditions when most of the cores are connected to one supply

CCM, when the changes in load current of one of the output voltage rails result in the change of output voltage of another rail. This happens largely because of the load transients on voltage rails in a system. Often, SIMO converters are over-designed to address the worst case load transients to
address cross-regulation. The well-defined loading configurations in a system implementing PDVS, the load transients, are well known in advance. It is also well known which of the three rails is going to be heavily loaded and which is lightly loaded at any point in time. This information can be easily obtained by scrutinizing the number of cores (or type of cores) getting connected to a given rail at a given time. Since the information is already known or needed to configure the header switches in PDVS (Figure 4-16). We use this information in our SIMO converter design to set the priority of the switching rails. For example, if a SIMO is designed to provide 0.9V, 0.7V and 0.4V rails, and 0.9V rail is heavily loaded, then the 0.9V rail is set on the priority. This way, the 0.9V rail is regulated first and then the 0.7V and 0.4V. If loading information is not well-known, priority cannot be set correctly and higher cross-regulation can be seen on the rails. In order to maintain the rails within the specified error, often higher values of capacitors are used, with value in μ F range. This design proposes new circuit methods, in conjunction with the PDVS, to significantly lower the value of the decoupling capacitor. The capacitor used in the proposed SIMO are 2.1nF for 0.4V, and 4.3nF for 0.7V and 0.9V V_{DD} rails.

4.2.1 Prototype Implementation and Results

In PDVS, the abrupt change of load due to DVS control is deterministic. If most of the cores switch to 0.7V from 0.9V, the PDVS controller will indicate this to the SIMO converter, which will prioritize the 0.7V rail. The feed-forward information can be used to dynamically reassign the priority that controls cross-regulation due to abrupt load transition. This eliminates complex feedback schemes that will also reduce efficiency at light load conditions. Figure 4-27 shows the measured ripple on the 0.7V and 0.9V rails with and without our DVS priority scheme. The ripple on the 0.9V supply, due to our priority scheme, reduces by 30mV in measurements. Figure 4-27

shows the steady state outputs of the SIMO at high and moderate loads. When the SIMO supplies 10mA on 0.9V, 1mA on 0.7V, and 1mA on 0.4V, the efficiency is 86%, with measured ripple of 40mV. The measured gate leakage of the 0.9V and 0.7V decap was 5μ A and 1μ A at room temperature.



SIMO output at light load/moderate load: 10mA on 0.9 V, 100µA on 0.7V and 0.4V

Figure 4-27 Measured outputs of the SIMO. During different load switches at constant total load of 10 mA, compensating PDVS changes with rail priority adjustments reduces ripple by 30 mV. High load efficiency is 86%, and low load efficiency is 62%.

Figure 4-28 shows the measured efficiency of the converter in various configurations. The peak

efficiency for the standalone 0.9V converter is 88% at 10 mA, and with the low voltage comparator,



the efficiency of the standalone 0.4V converter was 60% at 100μ A and 68% at 1mA (vs. 33% for ideal LDO) over 23% better than the switched capacitor converter reported in [104]

Figure 4-28 Measured efficiency plots for different configurations

Table 4-1 shows the comparison of the proposed work with prior low load SIMO and on-chip converters. The proposed design is a first integrated SIMO design with 1μ H external inductor. The measurement result shows that proposed design achieves a peak efficiency of 86% with a maximum ripple of 40mV.

Table 4-1 Comparison with prior art for low load converters. This design supports 3 outputs at lower voltage and load than any prior SIMO with comparable efficiency and ripple but no off-chip capacitors.

	This work	[103]	[101]	[104]
Technology	65nm	65nm	250nm	45nm
Topology	SIMO buck	SIMO buck	SIMO buck/boost	Switched Capacitor
# Outputs	3	5	4	1
V _{IN} (V)	1.0-1.1	3.4-4.3	1.8-2.2	1.8
V _{OUT} (V)	0.0.0.7.0.4	2.8, 2.5, 1.8,	2.25, 2.0,	10.0.9
	0.9,0.7,0.4	1.5, 1.2	1.35, 1.25	1.0-0.8
Target I _{LOAD}	0.9V: 10mA-10µA	<230mA per	100mA-	8mA-100µA
	0.7V: 10mA-10µA	output	~20mA	
Efficiency @	0.4 V: ImA-1μA 86% @ [10mA.	83.1% @	90% @ high	69% @ 1V. 5mA
load	1mA, 1mA]	75mA/out	80% @ low	66% @0.9V, 5mA
Max Ripple	40mV	40mV	22mV	50mA
Caps	On Chip	Off-chip	Off-chip	On Chip
Area (mm ²)	2 (0.03 w/o caps)	1.86	1.8 x 2.1	0.16
Inductor	1µH	2.2µH	10µH	NA

Figure 4-29 shows the implementation of the proposed with on-chip decaps. The capacitors for the design are implemented using nMos capacitors. The value of the capacitance is 4.3nF for 0.9V and 0.7V rails and 2.3nF for 0.4V rail. The capacitance contributes to ~1 μ A of standby current, because of the gate oxide leakage. The total area of the converter is 2mm². The area of the control circuits is 0.03mm².



Figure 4-29 Die microphotograph of the SIMO chip and performance summary

We proposed a SIMO DC-DC converter design for the PDVS system that can support a peak load of 50mA on the 0.9V and 0.7V rails, and 10mA on the 0.4V supply. To reduce system level volume and cost, we integrate two 4.3nF (0.9, 0.7V) and one 2.1nF (0.4V) poly-Nwell decaps on-chip. The PDVS load circuit uses multiple header switches to select each local core V_{DD} from a discrete set of global V_{DD} rails, depending on the local workload, which allows V_{DD} switching times near 1ns [2]. The PDVS controller shares connection information with the SIMO controller, allowing it to

set the priority of switching for the given rails. In this implementation, the highest priority is given either to 0.7V or 0.9V, since they will see a higher load than the 0.4V sub-threshold rail, unless their loads are power gated. The 0.4V supply can be used for operations during idle mode, in which case it will be lightly loaded, with the other rails potentially off. We have a lower static current mode for 0.4V rail. The proposed topology gives 86% peak efficiency over a wide range while using on-chip decap. It provides a low cost, highly efficient implementation for PDVS.

4.3 Conclusions

A model has been presented that can accurately capture the behavior of inductor based DC-DC converters in a dynamic environment. The converter model has been validated and compared with measured results from a variety of DC-DC converter topologies in existing literature. We used this model to study block level power management techniques for a SoC by incorporating it into a higher level model of the multiple block system. The system model predicts that there is a break-even time before the benefit of voltage scaling becomes positive, and our proposed modeling framework provides a quantitative means for comparing multiple power management techniques in a given use case scenario. Further, we proposed a SIMO DC-DC converter design for the PDVS system that can support a peak load of 50mA on the 0.9V and 0.7V rails, and 10mA on the 0.4V supply. To reduce system level volume and cost, we integrated two 4.3nF (0.9, 0.7V) and one 2.1nF (0.4V) poly-Nwell decaps on-chip. The proposed topology gives 86% peak efficiency over a wide range, while using on-chip decap. It provides a low cost, highly efficient implementation for PDVS.

Chapter 5

Conclusions

ULP SoCs, such as BSNs, WSNs, or IoTs, are used for sensing activities, our health, and the environment. These miniaturized nodes are responsible for sensing data periodically, processing it, and communicating through RF. The information about our environment, health, home, etc. can significantly improve the way we live. The sensor nodes can be used for many sensing applications – measuring and reporting everything from the flow of crude oil in a remote pipeline, to the degree of corrosion in a steel bridge, to EKG, EEG and EMG signals of a home health patient. These devices require small size and must consume extremely low power to be able to operate from harvested energy for their longer lifespan. Significant progress has been made over the last few years. However, there is a need to improve the operating lifetime of these systems, for their extensive deployment in the environment. This dissertation successfully improves the lifetime of the system by increasing the amount of harvested energy, reducing the losses in voltage conversion, reducing the operating voltage, and reducing the standby power consumption. Further, a modeling framework to enable accurate power management, and integrated on-chip implementation of the PDVS power management technique are also presented.

5.1 Impact of contributions on system life-time

This dissertation improves the lifetime of the system by increasing the amount of harvested energy, reducing the losses in voltage conversion, reducing the operating voltage, and reducing the standby

power consumption. Figure 5-1 shows the performance of the harvesting solution and low voltage operation compared with the state-of-the-art system [1]. The proposed solution starts operation at 0.8V compared to 1.35V in [1]. Therefore, the energy needed to start the system for BSN SoC [1] is 9.11 μ J and 3.2 μ J for the proposed EHM solutions, using a storage capacitor of 10 μ F. The proposed solution lowers the amount of energy needed to start the system by ~3X.



b) Energy Harvesting and operation with proposed Solution

Figure 5-1 Energy System level performance impact of the proposed EHM Solution

Further, both systems will have to start from zero energy level in a harvesting environment; therefore, system start-up time is another key metric. This is the time the system needs to achieve the operating voltage after ambient sources have available energy. The system start-up happens after cold start-up of 600mV for both systems. However, the BSN SoC [1] needs to achieve a start-

up voltage of 1.35V to start operating, compared to 0.8V in the proposed system. Further, the harvesting efficiency is 72% in the proposed system, compared to 38% in BSN SoC [1]. The system start-up time for BSN SoC comes out to be 384ms, compared to 20ms for the proposed system, thus lowering the system start-up time by ~20X. Therefore, the proposed solution reduces the start-up energy for the system by 3X and start-up time by 20X.



Figure 5-2 Current Consumption break-down of a BSN SoC [86] while performing ECG

Figure 5-2 shows the current consumption of various blocks in a BSN SoC [86]. The proposed solution reduces the power consumption of various blocks in the system. It reduces the power consumption of the clock source from 2.7μ W to 1nW. The power consumption of supply regulation is brought down from 4μ W to 0.5μ W. Further, not included in this dissertation, the author has also proposed an ADC operating at 100nW. These contributions can bring the power consumption of the system to ~8 μ W. Now supposing the system is powered to 1.5V in both the cases and the harvesting source is cut-off. The time for the system to shut down is 112ms for the BSN SoC in [1], while using the proposed solution the system can be kept on for 1s, increasing the power down time for the system by 9X. The contributions help in reducing the start-up by 20X

and operational time by 9X for a system which is operating all the time. Figure 5-3 summarizes the results.



Figure 5-3 Lifetime improvement by proposed solution. The system start-up energy is reduced by 3X, start-up time by 20X and operational time is increased by 9X



b) Power consumption with proposed solutions



We showed the benefits from the proposed solution in a particular application. Now, a more general study of the improvement is presented. Figure 5-4 shows a system with power management and clock sources separated from the system. Both the clock source and power management circuit consume standby power, and they are always on. The power is delivered from the V_{CAP} through power management circuits with efficiency given by η . For the BSN chip [1], the power management circuit consumes 3µA current, while the clock source consumes 2µA current. The proposed solution reduces this power consumption to 0.3µA and 2nA respectively. A more general characteristic of the system can be obtained by varying the current for processing and communication blocks. These blocks are usually duty-cycled to save power. Therefore, a wider picture of the benefits from the proposed solution can be obtained by sweeping the average load current I_L, of processing and communication blocks.



Figure 5-5 Operational lifetime improvement in ULP systems by using proposed contributions

Figure 5-5 shows the improvement in the operational lifespan of a BSN chip using the proposed solution. A 4X improvement in lifespan is achieved, using the proposed solution for cases when the system consumes higher power (mA). The 4X benefit is achieved through more efficient DC-DC converters and by enabling lower voltage operation for the system. The lifespan is increased by over 70X when the system consumes low power. At low power conditions, the lifespan is determined by power consumption of blocks that are always on. These are clock sources, and bandgap reference. The improved designs for clock sources, bandgap references, etc. presented in this dissertation, reduce the power consumption significantly. Further, lower voltage application also provides more energy for sustaining the operation for a longer time. These benefits are shown assuming the same amount of energy is harvested in both systems. However, our energy harvesting solution is more efficient and harvests from very low voltages, as well, which will increase the amount of harvested energy to further improve the lifetime.

5.2 Summary of Contributions

The solutions proposed in the dissertation improves the lifetime of the system by increasing the amount of harvested energy, reducing the losses in voltage conversion, reducing the operating voltage, and reducing the standby power consumption. The following contributions are made to achieve higher life-time for ULP systems.

• A stable on-chip clock source which can lock to a given clock frequency, a locking scheme to lock the clock source and ultra-low power on-chip clock sources to reduce the idle mode or standby.

- A ULP crystal oscillator circuit as an alternative clocking solution, with off-chip component with lower power and higher stability.
- A low power, high efficiency energy harvester that can harvest from 10mV ambient source to increase the amount of harvested energy.
- A low power, low voltage bandgap reference for energy harvesting and power management, to reduce the idle mode power consumption and enable low voltage operation.
- A single inductor, multiple output buck-boost converter, which enables high efficiency circuits as well as low voltage operation.
- A single inductor, multiple output energy harvesting and power management solution to harvest from low power solar cells to enable high efficiency, low cost, integrated harvesting and power management system.
- A model which accurately establishes the benefits of power management techniques for ULP SoCs in the presence of voltage regulators, to enable design time choices for regulators and power management techniques.
- Single Inductor Multiple output Voltage regulators with on-chip decoupling capacitors to implement panoptic dynamic voltage scaling technique.
- While not included in this dissertation, the author also made the following contributions for overall improvement in the ultra-low power system design.
 - A clock and data recovery (CDR) circuit for the RF interface of BSN;
 - A low power interconnect circuit operational from 0.3V to reduce the interconnect power consumption;
 - An ultra-low power analog to digital converter for ECG, EKG applications;

- Design and system level improvements and options for inter-chip communication between ULP ICs;
- Modeling techniques to assess circuit and system reliability;
- Circuit technique to improve the start-up time of a crystal oscillator by an order of magnitude.

5.2 Conclusions and Open Problems

The work that this dissertation describes improves the operational lifespan of ULP systems. The energy harvesting and power management circuits, voltage reference circuits, clock sources, PDVS SIMO DC-DC, overcome several challenges for increasing the lifespan of ULP systems. This section offers the key lessons from each of these areas and discusses opportunities for future work.

The ULP on-chip clock source provides a method for implementing a low power, stable, on-chip clock. However, the solution presents opportunity for improvements. For example, power consumption can be reduced significantly by reducing the power consumption of an uncompensated clock. A design effort is already underway to reduce this power. The proposed locking scheme locks only within the jitter at the input clock. Therefore, there is an opportunity to improve the locking scheme by implementing the lock through a PLL kind of architecture, to eliminate interference from noise. Further, the crystal oscillator power can potentially be reduced further if the swing of the oscillation is reduced to 100mV or lower. Even lower power consumption can further improve the system lifespan.

The energy harvesting solution provides an efficient means to harvest from low power, low voltage sources. However, a key improvement needed in this area is the ability to cold start circuit operational from very low voltages. These systems currently need an external source for wake-up, which comes in the form of RF kick-start or mechanical switch or transformers. There is a need for cold-start circuits that can operate from even lower voltages, as well as be small in shape and sizes. Further, the maximum power point tracking scheme needs external components, and it is based on predicted system behavior. Often, the MPP point can vary from its predicted point and a more robust maximum power tracking solution can further improve the system performance. Cost of the MPP circuit can also be reduced by reducing the external components.

The single inductor, multiple output solutions presented in this dissertation provide a low cost means for achieving higher efficiency. The solution can support current loads of up to 30mA, which caters to low power applications. However, even wider applications can be targeted by increasing the load current consumption to a few 100mA. A nice improvement over proposed designs will be SIMO for ULP systems that can support a few 100mA. The PDVS SIMO can be improved further by improving the transient behavior at fast load transients.

Appendix A: List of Acronyms

Analog to Digital Converter		
Analog Front End		
Bill of Material		
Body Sensor Node		
Continuous Conduction Mode		
Clock Data Recovery		
Complementary Metal Oxide Semiconductor		
Central Processing Unit		
Complementary to Absolute Temperature		
Direct Current		
Duty-Cycled Clock		
Discontinuous Conduction Mode		
Digitally Controlled Oscillator		
Delay Locked Loop		
Dynamic Voltage and Frequency Scaling		
Dynamic Voltage Scaling		
Electrocardiogram		
Electroencephalogram		
Energy Harvesting and Power Management		
Electrocardiogram		
electromotive force		

EMG	Electromyogram
ESR	Effective Series Resistance
HS	High Side
IC	Integrated Circuits
ΙΟ	Input Output
ІоТ	Internet of Things
$\mathbf{L}\mathbf{V}_{\mathbf{T}}$	Low Threshold Voltage
LS	Low Side
LDO	Low Drop Out
MIM	Metal Insulator Metal
MPP	Maximum Power Point
MPPT	Maximum Power Point Tracking
nMOS	n-type Metal Oxide Semiconductor
Osccmp	Compensated Oscillator
Oscref	Reference Oscillator
Oscucmp	Uncompensated Oscillator
PDVS	Panoptic Dynamic Voltage Scaling
PFM	Pulse Frequency Modulation
pMOS	p-type Metal Oxide Semiconductor
PTAT	Proportional to Absolute Temperature
PV	Photo Voltaic (Solar Cell)
PWM	Pulse Width Modulation
RF	Radio Frequency

- **RLC** Resistor Inductor Capacitor
- **RMS** Root Mean Square
- **RTC** Real Time Clock
- SAR Successive Approximation Register
- **SIMO** Single Inductor Multiple Output
- SoC System on Chip
- **TDM** Time Division Multiplexing
- **TEG** Thermo Electric Generator
- **UDVS** Ultra Dynamic Voltage Scaling
- ULP Ultra Low Power
- WSN Wireless Sensor Node
- **XTAL** Crystal

Appendix B: Efficiency Dependence of

DC-DC on Peak Inductor Current



Figure A. 1 Losses in DC-DC converter

We presented methods to control the peak inductor current (I_P) in Chapter 3 for the buck and the boost converter designs. In this section we present the analytical reasons for controlling and maintaining a peak inductor current in DC-DC converter designs. The control of the peak current of the inductor is central to the maximization of the efficiency in a DC-DC converter. To understand the impact of peak inductor current on the efficiency of the DC-DC converter, we have to understand the losses in the DC-DC converter. In a DC-DC converter design, a control circuit is used for voltage regulation and for other control mechanisms. The control circuit usually contains analog circuits which consume static or bias currents for their operation. The energy overhead of the control circuits constitutes the Static losses in a DC-DC converter. This loss will be a constant for a given Vi and Vo combination. We denote the static energy loss as E_{ST}. The control circuit continuously switches transistors MP and MN of Figure A.1 to charge the inductor. There is an energy overhead in switching the circuits and the transistors. This overhead is called switching loss. This loss is also a constant for a given Vi and Vo combination. We denote the switching loss as E_{SW}. Finally, the third component of losses inside the DC-DC converter is called the conduction loss (E_{CN}) and it occurs in the switches MP and MN and the parasitic resistance of the inductor and capacitor. The switch MP is turned on for a set time which charges the inductor to a peak current (I_P). The conduction loss is a function of the peak current. The total energy transferred in one switching cycle is also a function of I_P. Therefore, value of I_P plays a great role in the overall efficiency. The analytical analysis that follows explains the dependence of efficiency of a DC-DC converter on I_P. The total loss in one switching cycle of a DC-DC converter can be written as the sum of static, switching and conduction loss. Therefore,

$$E_L = E_{ST} + E_{SW} + E_{CN} \tag{A.1}$$

$$E_L = a + E_{CN} \tag{A.2}$$

Where *a* is the constant for the sum of switching and static loss. If the energy transferred in each cycle is given by E_T , then the percentage loss is given by,

$$\eta_L = \frac{E_L}{E_T} = \frac{a + E_{CN}}{E_T} \tag{A.3}$$

A switching cycle in DC-DC converter involves charging the inductor to I_P and then transferring the stored energy in the inductor to the load. Assuming that the losses involved in charging the inductor is very small compared to the energy stored in it, the total energy transferred from source voltage Vi to output load can be approximately given by,

$$E_T = 0.5LI_P^2 \tag{A.4}$$

Now we will try to find out the conduction loss in terms of I_P.



Figure A. 2 High side conduction

The switching cycle involves two conduction cycles, high side (HS) and low side (LS) switching as explained in Chapter 3. Both HS and LS switching involves conduction loss. Let us say that the total resistance shown in the HS conduction is given by $R_{\rm H}$. Assuming that the inductor charges from 0 to $I_{\rm P}$ current in high side turn on time $T_{\rm HS}$. Therefore, total conduction loss in high side switching can be written as,

$$E_{CN,H} = \int_0^{T_{HS}} i^2 R_H \, dt \tag{A.5}$$

Also, the inductor charging is given by,

$$L\frac{di}{dt} = (Vi - Vo) \Longrightarrow dt = \frac{Ldi}{Vi - Vo}$$
(A.6)

Putting it in equation (A.5) we get,

$$E_{CN,H} = \int_0^{I_P} \frac{i^2 R_H L di}{V i - V o} \tag{A.7}$$

Therefore,

$$E_{CN,H} = \frac{LR_H I_P^{\ 3}}{3(Vi - Vo)}$$
(A.8)

Similarly, the low side conduction loss can also be obtained. Figure A. 3 shows the case when inductor current discharges.



Figure A. 3 Low side loss

Total conduction loss in high side switching can be written as,

$$E_{CN,L} = \int_0^{T_{LS}} i^2 R_L \, dt \tag{A.9}$$

The inductor discharge rate will be given by,

$$L\frac{di}{dt} = -Vo \Longrightarrow dt = -\frac{Ldi}{Vo}$$
(A.10)

Therefore,

$$E_{CN,L} = -\int_{I_P}^{0} \frac{i^2 R_L \, L di}{Vo} \tag{A.11}$$

Therefore,

$$E_{CN,L} = \frac{LR_H I_P^3}{3Vo} \tag{A.12}$$

Total conduction loss is given by,

$$E_{CN} = E_{CN,H} + E_{CN,L} = \frac{L}{3} \left(\frac{R_H}{Vi - Vo} + \frac{R_L}{Vo} \right) I_P^3 = b I_P^3$$
(A.13)

Where *b* is given by $\frac{L}{3} \left(\frac{R_H}{Vi - Vo} + \frac{R_L}{Vo} \right)$. Therefore, total loss using equation (A.2) is given by,

$$E_L = a + bI_P{}^3 \tag{A.14}$$

Therefore total percentage loss is given by,

$$\eta_L = \frac{E_L}{E_T} = \frac{a + bI_P{}^3}{0.5LI_P{}^2} \tag{A.15}$$

Therefore,

$$\eta_L = \frac{a}{0.5LI_P^2} + \frac{bI_P}{0.5L} \tag{A.16}$$

The expression in equation (A.16) increases at both high and low values of I_P . The minimum loss is given by,

$$\frac{d\eta_L}{dI_P} = -\frac{4a}{LI_P^3} + \frac{2b}{L} \tag{A.17}$$

The I_P value for minimum loss is given by,

$$I_P = \sqrt[3]{\frac{2a}{b}} \tag{A.18}$$

Therefore a minimum loss or the maximum efficiency of the DC-DC converter is given by the peak inductor current I_P given by equation (A.18). The given analytical expression proves the need for maintaining a constant peak inductor current.

Appendix C: List of Publications

- [AS1] A. Shrivastava, and B. H. Calhoun, "A 50nW, 100kbps Clock/Data Recovery Circuit in an FSK RF Receiver on a Body Sensor Node" VLSI Design Conference. Jan. 2013.
- [AS2] Y. Zhang, F. Zhang, Y. Shakhsheer, J. Silver, M. Nagaraju, A. Klinefelter, J. Pandey, J. Boley, E. Carlson, A. Shrivastava, et al., "A Batteryless 19 uW MICS/ISM-Band Energy Harvesting Body Sensor Node SoC for ExG Applications ", Journal of Solid State Circuit, Jan. 2013.
- [AS3] A. Shrivastava, Y. Ramadass, S. Bartling and B. H. Calhoun, "Single Inductor Energy Harvesting and Power Management Circuit for Body Sensor Nodes", ISSCC SRP 2013.
- [AS4] A. Shrivastava, and B. H. Calhoun, "Modeling DC-DC Converter Efficiency and Power Management in Ultra Low Power Systems"" Sub-threshold Conference Oct-2012.
- [AS5] A. Shrivastava, and B. H. Calhoun, "A 150nW, 5ppm/o C, 100kHz On-Chip Clock Source for Ultra Low Power SoCs" Custom Integrated Circuits Conference. Sept. 2012.
- [AS6] A. Shrivastava, J. Lach, and B. H. Calhoun, "A Charge Pump Based Receiver Circuit for a Voltage Scaled Interconnect" International Symposium on Low Power Electronics and Design, Jul. 2012
- [AS7] F. Zhang, Y. Zhang, J. Silver, Y. Shakhsheer, M. Nagaraju, A. Klinefelter, J. Pandey, J. Boley, E. Carlson, A. Shrivastava, et al., "A Batteryless 19uW MICS/ISM-Band Energy Harvesting Body Area Sensor Node SoC", International Solid State Circuits Conference, Feb. 2012.
- [AS8] A. Shrivastava, and B. H. Calhoun, "A sub-threshold clock and data recovery circuit for a wireless sensor node", Sub-threshold Conference. Sept. 2011.

- [AS9] A. Shrivastava, and B. H. Calhoun, "A DC-DC Converter Efficiency Model for System Level Analysis in Ultra Low Power Applications", Journal of Low Power Electronics Application 3-2013
- [AS10] A. Shrivastava, Y. K. Ramadass, S. Khanna, S. Bartling, and B. H. Calhoun, "A 1.2µW SIMO Energy Harvesting and Power Management Unit with Constant Peak Inductor Current Control Achieving 83-92% Efficiency Across Wide Input and Output Voltages", Symposia on VLSI Technology and Circuits, June 2014

Upcoming Publications : Own

- [AS11] SIMO for PDVS Implementation, Submitted
- [AS12] A 10mV TEG boost converter, Submitted
- [AS13] Ultra-low power crystal oscillator, Submitted
- [AS14] A low power ADC Circuit, Submitted

Upcoming Publications : Group

[AS15] BSN Chip-2

[AS16] Low Power Clocking solution

Patents:

[AS17] A. Shrivastava, and Y. Ramadass, "Apparatus and Method for Controlling Peak Inductor Current in a Switched Mode Power Supply" US Patent application 13768448

- [AS18] A. Shrivastava, Y. Ramadass, and S. Bartling, "Single Inductor Energy Harvesting and Management Interface System and Method" US Patent application
- [AS19] A. Shrivastava, and B. H Calhoun, "Single Inductor Multiple Output (SIMO) Step-down DC-DC Converter for Ultra Low Power SOCs" US Patent application 61/700979
- [AS20] A. Shrivastava, and B. H. Calhoun. US Patent Application 61/698,534. "On-Chip Clock Source for Ultra Low Power SoCs"
- [AS21] A. Shrivastava, et al, US Patent 8120439 "A Fast Start-up Crystal Oscillator"
- [AS22] A. Shrivastava, and B. H. Calhoun. US Patent Application. "Ultra-Low Power crystal oscillator circuit"

Bibliography

- [1] F. Zhang, et. al, "A Battery-less 19µW MICS/ISM-Band Energy Harvesting Body Area Sensor Node SoC ", IEEE International Solid State Circuits Conference, 2012
- [2] M. A. Hanson, H. C. Powell Jr, A. T. Barth, K. Ringgenberg, B. H. Calhoun, J. H. Aylor, and J. Lach, "Body Area Sensor Networks: Challenges and Opportunities", Computer, vol. 42, no.
 1: IEEE Computer Society Press Los Alamitos, CA, USA, pp. 58–65, 1/2009
- [3] B. H. Calhoun, S. Khanna, J. Ryan, and B. Otis, "System design principles combining subthreshold circuit and architectures with energy scavenging mechanisms," IEEE International Symposium on Circuits and Systems, 2010
- [4] B. H. Calhoun, A. Wang, and A. P. Chandrakasan, "Modeling and sizing for minimum energy operation in subthreshold circuits," IEEE Journal of Solid-State Circuits, Sept. 2005
- [5] X. Huang, A. Ba, G. Dolmans, H. de Groot, J. Long, "A 915MHz 120µW-RX/900µW-TX envelope-detection transceiver with 20dB in-band interference tolerance," IEEE International Solid-State Circuits Conference, 2012
- [6] Jagdish Pandey, Jianlei Shi, Brian Otis, "A 120uW MICS/ISM-Band FSK Receiver with a 44uW Low-Power Mode Based on Injection-Locking and 9x Frequency Multiplication," IEEE International Solid State Circuits Conference, 2011
- [7] A. Wang, and A. Chandrakasan, "A 180-mV Subthreshold FFT Processor Using a Minimum Energy Design Methodology" IEEE Journal on Solid State Circuit, Jan. 2005
- [8] Y.-S. Lin, D. Sylvester, and D. Blaauw, "A Sub-pW Timer Using Gate Leakage for Ultra Low-Power Sub-Hz Monitoring Systems," IEEE Custom Integrated Circuits Conference, 2007

- [9] Y.-S. Lin, D. Sylvester, and D. Blaauw, "A 150pW program-and-hold timer for ultra-lowpower sensor platforms" IEEE International Solid State Circuits Conference, 2009
- [10] Y. Lee, B. Giridhar, Z. Foo, D. Sylvester, D. Blaauw, "A 660pW Multi-stage Temperature Compensated Timer for Ultra-low Power Wireless Sensor Node Synchronization", IEEE International Solid State Circuits Conference, 2011.
- [11] Y. Tokuyanga, S. Sakiyama, A. Matsumoto, S. Dosho, "An On-chip CMOS Relaxation Oscillator with Voltage Averaging Feedback," IEEE Journal on Solid State Circuit, June 2010.
- [12] K. Wong, and D. Evans, "A 150mA Low Noise, High PSRR Low-Dropout Linear regulator in 0.13µm Technology for RF SoC Applications," IEEE European Solid State Circuits Conference, 2006.
- [13] D. Yoon, D. Sylvester, and D. Blaauw, "A 5.58nW 32.768kHz DLL-Assisted XO for Real-
- Time Clocks in Wireless Sensing Applications" IEEE International Solid State Circuits Conference, 2012
- [14] E. Vittoz, and J. Fellarath, "CMOS Analog Integrated Circuits Based on Weak Inversion Operations," IEEE Journal of Solid State Circuits, vol.12 no. 3 pp 224-231 Jun. 1977
- [15] W. Thommen, "An improved Low Power Crystal Oscillator," IEEE European Solid State Circuits Conference, 1999
- [16] Hector Ivan Oporta, "An Ultra-low Power Frequency Reference for Timekeeping Applications," Master's Thesis, Oregon State University, December, 2008
- [17] D. Lanfranchi, E. Dijkstra and D. Aebischer, "A Microprocessor-Based Analog Wristwatch Chip with 3 seconds/year Accuracy," IEEE International Solid State Circuits Conference, 1994

- [18] D. Aebischer, H. J. Oguey, and V.R. Von Kaenel, "A 2.1Mhz Crystal Oscillator Time Base with a Current Consumption under 500nA," IEEE European Solid State Circuits Conference, 1996
- [19] A. Shrivastava, R. Yadav, and P. K. Rana, "Fast Start-up Crystal Oscillator" US Patent 8120439.
- [20] S. M. Chen, T. J. Liang, and K. R. Hu, "Design, Analysis, and Implementation of Solar Power Optimizer for DC Distribution System," IEEE Transactions on Power Electronics, April 2013
- [21] R. C. N. Pilawa-Podgurski, and D. J. Perrault, "Submodule Integrated Distributed Maximum Power Point Tracking for Solar Photovoltaic Applications," IEEE Transactions on Power Electroncis, June 2013
- [22] A. K. Abdelsalam, A. M. Massoud, S. Ahmed, and P. Enjeti, "High-Performance Adaptive Perturb and Observe MPPT Technique for Photovoltaic-Based Microgrids," IEEE Transactions on Power Electronics, April 2011
- [23] Y. Lee, S. Bang, I. Lee, Y. Kim, G. Kim, M. H. Ghaed, P. Dutta, and D. Blaauw, "A Modular 1 mm³ Die-Stacked Sensing Platform With Low Power I²C Inter-Die Communication and Multi-Modal Energy Harvesting," IEEE Journal of Solid State Circuits, Jan. 2013
- [24] M. Fojtik, D. Kim, G. Chen, Y-S. Lin, D. Fick, J. Park, M. Seok, M-T. Chen, Z. Foo, D. Blaauw, and D. Sylvester, "A Millimeter-Scale Energy-Autonomous Sensor System With Stacked Battery and Solar Cells," IEEE Journal of Solid-State Circuits, March 2013
- [25] Y. Nakase, S. Hirose, H. Onoda, Y. Ido, Y. Shimizu, T. Oishi, T. Kumamoto, and T. Shimizu, "0.5 V Start-Up 87% Efficiency 0.75 mm² On-Chip Feed-Forward Single-Inductor Dual-Output (SIDO) Boost DC-DC Converter for Battery and Solar Cell Operation Sensor Network Micro-Computer Integration," IEEE Journal of Solid State Circuits, Aug. 2013

- [26] S. Roundy, P. Wright, and J. Rabaey, "Energy Scavenging for Wireless Sensor Networks with Special Focus on Vibrations," Kluwer Academic Press, 2003
- [27] N. J. Guilar, R. Amritharajah, and P. J Hurst, "A Full-Wave Rectifier With Integrated Peak Selection for Multiple Electrode Piezoelectric Energy Harvesters," IEEE Journal of Solid State Circuits, Jan. 2009
- [28] T. Hehn, F. Hagedom, D. Maurath, D. Marinkovic L. Kuehne, A. Frey, and Y. Manoli, "A Fully Autonomous Integrated Interface Circuit for Piezoelectric Harvesters," IEEE Journal of Solid State Circuit, Sep. 2012
- [29] Y. K. Ramadass, and A. P. Chandrakasan, "An Efficient Piezoelectric Energy Harvesting Interface Circuit Using a Bias-Flip Rectifier and Shared Inductor," IEEE Journal of Solid State Circuits, Jan. 2010
- [30] J. Masuch, M. Delgado-Restituto, D. Milosevic, and P. Baltus, "Co-Integration of an RF Energy Harvester Into a 2.4Ghz Transceiver," IEEE Journal of Solid State Circuits, July 2013
- [31] G. Papotto, F. Carrara, and G. Palmisano, "A 90nm CMOS Threshold-Compensated RF Energy Harvester," IEEE Journal of Solid State Circuits, Sept. 2011
- [32] G. Papotto, F. Carrara, A. Finocchiaro, and G. Palmisano, "A 90nm CMOS 5-Mbps Crystal-Less RF-Powered Transceiver for Wireless Sensor Networks," IEEE Journal of Solid State Circuits, Feb. 2014
- [33] T. Becker, M. Kluge, J. Schalk, T. Otterpohl, and U. Hillerigmann, "Power Management for Thermal Energy Harvesting in aircrafts," IEEE Sensors, Oct. 2008
- [34] J. Haider and J. Ghojel, "Waste Heat Recovery from Low Power Diesel Engine using Thermoelectric Generators," International Conference on Thermoelectrics, 2001

- [35] A. K. Hyder, R. L. Wiley, G. Halpret, D. J. Flood, and S. Sabripour, "Spacecraft Power Technologies," Imperial College Press, 2000
- [36] V. Leonov, T. Torfs, P. Fiorini, and C. Van Hoof, "Thermoelectric Converters of Human Warmth for Self-Powered Wireless Sensor Nodes," IEEE Sensors Journal, vol. 7, no. 5, pp. 650–657, May 2007.
- [37] M. Kishi, H. Nemoto, T. Hamao, Y. Yamamotto, S. Sudou, M. Mandai, and S. Yammamoto,
 "Micro thermoelectric modules and their application to wristwatches as an energy source,"
 International Conference on Thermoelectrics, pp. 301–307, 1999.
- [38] E.J. Carlson, K. Strunz, and B.P. Otis, "A 20 mV Input Boost Converter With Efficient Digital Control for Thermoelectric Energy Harvesting," IEEE Journal of Solid-State Circuits, vol.45, no.4, pp.741-750, April 2010.
- [39] Y. K. Ramadass and A. P. Chandrakasan, "A Battery-Less Thermoelectric Energy Harvesting Interface Circuit with 35mV Startup Voltage," IEEE Journal of Solid-State Circuits, vol.46, no.1, pp.333-341, January 2011.
- [40] K. Kadirvel, Y. Ramadass, U. Lyles, J. Carpenter, V. Ivanov, V. McNeil, and A. Chandrakasan, "A 330nA Energy-Harvesting Charger with Battery Management for Solar and Thermoelectric Energy Harvesting," IEEE International Solid State Circuits Conference, 2012.
- [41] J.P. Im, S.W. Wang, S.T. Ryu, and G.H. Cho, "A 40mV Transformer-Reuse Self-Startup Boost Converter with MPPT Control for Thermoelectric Energy Harvesting," IEEE Journal of Solid-State Circuits, vol.47, no.12, pp.3055-3067, December 2013.
- [42] L. E. Bell, "Cooling, Heating, Generating Power, and Recovering Waste Heat with Thermoelectric Systems," Science 321, 1457 (2008).

- [43] J. Damaschke, "Design of a low-input-voltage converter for thermoelectric generator," IEEE Transactions on Industry Applications, vol. 33, no. 5, pp. 1203–1207, September 1997
- [44] S. Lineykin and S. Ben-Yaakov, "Modeling and Analysis of Thermoelectric Modules," IEEE Transactions on Industry Applications, vol. 43, no. 2, pp. 505–512, March-april 2007
- [45] A. Pressman, K. Billings, and T. Morey, "Switching Power Supply Design, 3rd Edition" The McGraw-Hill Publication, 2009
- [46] H.F. Pooya and G.A. Rincon-Mora, "An Accurate, Continuous, and Lossless Self-Learning CMOS Current-Sensing Scheme for Inductor-Based DC-DC Converters," IEEE J. Solid-State Circuits, vol.42, no.3, pp.665-679, March 2007.
- [47] P. E. Allen, and D. R. Holberg, "CMOS Analog Circuit Design," 2nd Edition, Oxford Press
- [48] R. J. Wildar, "New developments in IC voltage regulators," IEEE Journal of Solid State Circuits, Feb. 1971
- [49] K. E. Kujik, "A precision reference voltage," IEEE Journal of Solid State circuits, June 1973
- [50] A. P. Brokaw, "A simple three terminal IC Bandgap Reference," IEEE Journal of Solid State Circuits, Dec. 1974
- [51] P. Kinget, C. Vezyrtzis, E. Chiang, B. Hung, and T.L. Li, "Voltage References for Ultra-Low Supply Voltages," IEEE Custom Integrated Circuits Conference, 2008
- [52] M. Seok, G. Kim, D. Sylvester, and D. Blaauw, "A 0.5V 2.2pW 2-Transistor Voltage Reference," IEEE Custom Integrated Circuits Conference, 2009
- [53] K. N. Leung, and P. K. T. Mok, "A CMOS Voltage Reference Based on Weighted ΔV_{GS} for CMOS Low-Dropout Linear Regulators," IEEE Journal of Solid State Circuits, Jan. 2003

- [54] H. Banba, A. Shiga, A. Umezawa, T. Miyaba, T. Tanzawa, S. Atsumi, and K. Sakui, "A CMOS bandgap reference circuit with sub-1V operation," IEEE Journal of Solid State Circuits, May, 1999
- [55] K. Sanborn, D. Ma, and V. Ivanov, "A sub-1V low noise Bandgap Voltage Reference," IEEE Journal of Solid State Circuits, Nov, 2007
- [56] V. Ivanov, R. Brederlow, and J. Greber," An Ultra Low Power Bandgap Operational from 0.75V," IEEE Journal of Solid State Circuits, July 2012
- [57] K. Ueono, T. Hirose, T. Asai, and Y. Amemiya, "A 300 nW 15 ppm/ C 20 ppm/V CMOS voltage reference circuit consisting of subthreshold MOSFETs," IEEE J. Solid-State Circuits, vol. 44, pp. 2047–2054, July, 2009.
- [58] P. Favrat, P. Deval, and M. J. Declercq, "A high-efficiency CMOS voltage doubler," IEEE Journal of Solid State Circuits, Mar. 1998
- [59] A. Richelli, L. Mensi, L. Colalongo, P. L. Rolandi, Z. M. Kovacs-Vanja, "A 1.2-to-8V Charge-pump with improved Power Efficiency for Non-Volatile Memories," Solid-State Circuits Conference, 2007
- [60] National Semiconductor LM78XX, Datasheet
- [61] Motorola MC7812, Datasheet
- [62] Texas Instruments UA7812, Datasheet
- [63] G. Rincon-Mora, P. E. Allen, "A low-voltage, low quiescent current, low drop-out regulator," IEEE Journal of Solid-State Circuits, Jan. 1998
- [64] M. Al-Shyoukh, H. Lee, and R. Perez, "A Trasient-Enhanced Low Quiescent Current Low-Dropout Regulator with Buffer Impedance Attenuation," IEEE Journal of Solid-State Circuits, Aug. 2007

- [65] D. El-Damak, S. Bandyopadhyay, A. P. Chandrakasan, "A 93% efficiency reconfigurable switched-capacitor DC-DC converter using on-chip ferroelectric capacitors," ISSCC 2013
- [66] H-P Le, S. R. Sanders, and E. Alon, "Design Techniques for Fully Integrated Switched Capacitor DC-DC converters," IEEE Journal of Solid-State Circuits, Sept. 2011
- [67] Y. K. Ramadass, A. A. Fayed, and A. P. Chandrakasan, "A Fully-Integrated Switched-Capacitor Step-Down DC-DC Converter With Digital Capacitance Modulation in 45 nm CMOS," IEEE Journal of Solid-State Circuits, Dec. 2010
- [68] D. Ma, W-H Ki, C-Y Tsui, and P.K.T Mok, "Single-inductor multiple output switching converters with time-multiplexing control in discontinuous conduction mode," IEEE Journal of Solid State Circuits, Jan. 2003
- [69] D. Ma, W-H Ki, and C-Y Tsui, "A pseudo-CCM/DCM SIMO switching converter with freewheel switching," IEEE Journal of Solid State Circuits, June 2003
- [70] K. W. R Chew, S. Zhuochao, H. Tang, and L. Siek, "A 400nW single-inductor dual-input-trioutput DC-DC buck-boost converter with maximum power point tracking for indoor photovoltaic energy harvesting," ISSCC 2013
- [71] Y. Qiu, C. V. Liemped, B. Op het Veld, P.G Blanken, and C. V. Hoof, "5µW-10mW Input Power Range Inductive Boost Converter for Indoor Photo Voltaic Energy Harvesting with Integrated Maximum Power Point Tracking Algorithm," International Solid State Circuits Conference 2011
- [72] I. Doms, P. Merken, R. Mertens, and C. V. Hoof, "Integrated Inductive Capacitive Power Management Circuit for Thermal harvester with Output Power 10 to 1000 μW," International Solid State Circuits Conference, 2009

- [73] T. Esram, and P. L. Chapman, "Comparison of Photovoltaic Array Maximum Power Point Tracking Techniques," IEEE Transactions on Energy Conservation, June 2007
- [74] A. Shrivastava, and B. H. Calhoun, "Modeling DC-DC Converter Efficiency and Power Management in Ultra Low Power Systems" Sub-threshold Microelectronics Conference, 2012.
- [75] S. M. Martin, K. Flautner, T. Mudge, and D. Blaauw, "Combined dynamic voltage scaling and adaptive body biasing for lower power microprocessors under dynamic workloads," International Conference on Computer Aided Design, 2002.
- [76] V. Kursun, S. G. Narendra, K. V. De, and E. G. Friedman, "Efficiency Analysis of a High Frequency Buck Converter for On-Chip Integration with a Dual-VDD Microprocessor," IEEE European Solid State Circuits Conference, 2002
- [77] Y. Choi, N. Chang, and T. Kim, "DC–DC Converter-Aware Power Management for Low-Power Embedded Systems," IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems, 2007
- [78] K. Yao, M. Ye, M. Xu, and F. C. Lee, "Tapped-Inductor Buck Converter for High-Step-Down DC-DC Conversion," IEE Transactions on Power Electronics. 2005, 20, 775-780
- [79] S. Bandyopadhyay, Y. K. Ramadass, and A. P. Chandrakasan, "20 μA to 100 mA DC–DC Converter With 2.8-4.2 V Battery Supply for Portable Applications in 45 nm CMOS," IEEE Journal of Solid-State Circuits. 2011, 12, 2807-2820.
- [80] W. Li, J. Xiao, Y. Zhao, and X. He, "PWM Plus Phase Angle Shift (PPAS) Control Scheme for Combined Multiport DC/DC Converters," IEEE Transactions on Power Electronics, 2012, 27, 1479-1489
- [81] J. Xiao, A. V. Peterchev, J. Zhang, and S. R. Sanders, "A 4µA Quiescent-Current Dual-Mode Digitally Controlled Buck Converter IC for Cellular Phone Applications," IEEE Journal of Solid-State Circuits. 2004, 12, 2342-2348.
- [82] T. Kuroda, K. Suzuki, S. Mita, T. Fujita, F. Yamane, F. Sano, A. Chiba, Y.Watanabe, K. Matsuda, T. Maeda, T. Sakurai, T. Furuyama, "Variable supply-voltage scheme for low-power high-speed CMOS digital design," IEEE Journal of Solid-State Circuits. 1998, 33, 454–462.
- [83] Y. K. Ramadass, A. Fayed, and A. P. Chandrakasan, "A Fully-Integrated Switched-Capacitor Step-down DC-DC converter with digital capacitance modulation in 45nm CMOS," IEEE Journal of Solid-State Circuits. 2010, 45, 2557- 2565.
- [84] G. Patounakis, Y. W. Li, and K. L. Shepard, "L. A fully integrated on-chip DC-DC conversion and power management system," IEEE Journal of Solid-State Circuits, 2004, 39, 443 - 451.
- [85] C. F. Lee, and P. K. T. Mok, "A monolithic current-mode CMOS DC-DC converter with onchip current-sensing technique," IEEE Journal Solid-State Circuits. 2004, 39, 3-14.
- [86] Y. Zhang, F. Zhang, Y. Shakhsheer, J. Silver, A. Klinefelter, M. Nagaraju, J. Boley, J. Pandey, A. Shrivastava, E. Carlson, A. Wood, B. H. Calhoun, and B. Otis, "A Battery-less 19μW MICS/ISM-Band Energy Harvesting Body Sensor Node SoC for ExG Applications," IEEE Journal of Solid-State Circuits. 2013, 48, 199 - 213.
- [87] Y. Ramadass, and A. P. Chandrakasan, "Minimum Energy Tracking Loop With Embedded DC–DC Converter Enabling Ultra-Low-Voltage Operation Down to 250 mV in 65 nm CMOS," IEEE Journal Solid-State Circuits. 2008, 43, 256 - 265.
- [88] Y. Shaksheer, S. Khanna, K. Craig, S. Arrabi, J. Lach, and B. H. Calhoun, "A 90nm data flow processor demonstrating fine grained DVS for energy efficient operation from 0.25V to 1.2V," IEEE Custom Integrated Circuits Conference 2011.

- [89] A. Richelli, S. Comensoli, Z. M. Kovacs-Vajna, "A DC/DC Boosting Technique and Power Management for Ultralow-Voltage Energy Harvesting Applications," IEEE Transactions on Industrial Electronics. 2012, 59, 2701-2708
- [90] G. Bassi, L Colalongo, A. Richelli, Z. Kovacs-Vajna, "A 150mV-1.2V fully-integrated DC-DC converter for Thermal Energy Harvesting," International symposium on Power Electronics, Electrical Drivers, Automation and Motion. 2012, 331-334
- [91] A. Richelli, L. Colalongo, S. Tonoli, Z. M. Kovacs-Vajna, "A 0.2 V DC/DC Boost Converter for Power Harvesting Applications," IEEE Transactions on Power Electronics 2009, 24, 1541-1546
- [92] B. H. Calhoun, A. Wang, and A. P. Chandraksan, "Modeling and Sizing for Minimum Energy Operation in Subthreshold Circuits," IEEE Journal of Solid-State Circuits. 2005, 40, 1778-1786
- [93] T. Kuroda, K. Suzuki, S. Mita, T. Fujita, F. Yamane, F. Sano, A. Chiba, Y. Watanabe, K. Matsuda, T. Maeda, T. Sakurai, and T. Furuyama, "Variable supply-voltage scheme for low-ower high-speed CMOS digital design," IEEE J. Solid-State Circuits, vol. 33, no. 3, pp. 454–462, Mar. 1998.
- [94] Y. Shakhsheer, S. Khanna, K. Craig, S. Arrabi, J. Lach, B.H. Calhoun, " A 90nm data flow processor demonstrating fine grained DVS for energy efficient operation from 0.25V to 1.2V," Custom Integrated Circuits Conference, 2011
- [95] A. Shrivastava, B. H. Calhoun "A DC-DC Converter Efficiency Model for System Level Analysis in Ultra Low Power Applications" Journal on Low Power Electronics and Applications. JLPEA 215-232

- [96] D. Goder, H. Santo, "Multiple output regulator with time sequencing," U.S. Patent 5 617 015, Apr. 1, 1997
- [97] M. Putic, L. Di, B. H. Calhoun, J. Lach, "Panoptic DVS: A Fine-Grained Dynamic Voltage Scaling Framework for Energy Scalable CMOS Design," International Conference on Computer Design, pp.491-497, 2009.
- [98] V. Gutnik and A. Chandrakasan, "Embedded power supply for lowpower DSP" IEEE Transaction on VLSI, 5(4):425-435, 1997.
- [99] H.-P Le, C.-S. Chae, K.-C Lee, S.-W. Wang, G.-H Cho and G.-H. Cho,"A Single-Inductor Switching DC-DC Converter With Five Outputs and Ordered Power-Distributive Control", IEEE Journal on Solid-State Circuits, vol. 42, pp. 2706-2714, Dec. 2007
- [100] Y. J. Woo, H.-P. Le, G.-H Cho, G.-H Cho and S.-I Kim, "Load-Independent Control of Switching DC-DC converters With Freewheeling Current Feedback," IEEE Journal on Solid-State Circuits, vol. 43, pp. 2798-2808, Dec. 2008
- [101] M.-H Huang and K.-H Chen, "Single-Inductor Multi-Output (SIMO) DC-DC Converters With High Light-Load Efficiency and Minimized Cross-Regulation for Portable Devices," IEEE Journal of Solid-State Circuits, vol. 44, pp. 1099-1107, April 2009
- [102] C.-S. Chae, H.-P. Le, K.-C. Le, G.-H. Cho and G.-H. Cho, "A Single-Inductor Step-up DC-DC Switching Converter With Bipolar Outputs for Active Matrix OLED Mobile Display Panels," IEEE Journal of Solid-State Circuits vol. 44, pp. 509-524, Feb. 2009
- [103] C-W Kuan, and H-C Lin, "Near-independently regulated 5-output single-inductor DC-DC buck converter delivering 1.2W/mm2 in 65nm CMOS," International Solid-State Circuits Conference, 2012

- [104] Y. Ramadass, A. Fayed, and A. Chandrakasan, "A Fully-Integrated Switched-Capacitor Step-Down DC-DC Converter with Digital Capacitance Modulation in 45 nm CMOS," JSSC, Vol. 45, No. 12, Dec. 2010.
- [105] M. Popovich, et al, "Decoupling capacitors for multi-voltage Power Distribution Systems ...," TVLSI, Vol. 14 No. 3 March 2006.